TABLE OF CONTENTS

ABSTRACT	ii
ACKNOWLEDGMENTS	iv
FUNDING AND SUPPORT	x
CONTRIBUTION OF STUDY INVESTIGATORS	xi
PUBLICATIONS AND CONFERENCE PRESENTATIONS	xiii
OTHER PRESENTATIONS AND AWARDS	xvi
LIST OF FIGURES	xxii
LIST OF TABLES	xxiii
ABBREVIATIONS	xxiv
CHAPTER 1. THESIS INTRODUCTION	1
CHAPTER 2. THE BURDEN OF NUTRITITION RELATED-DISEASE AND TRENDS IN DIET	ΓARΥ
INTAKE	5
2.1. Overview	5
2.2. Epidemiology of nutrition-related disease	6
2.2.1 Prevalence of nutrition-related disease in New Zealand	7
2.3. Trends in dietary intake	9
2.3.1 Trends in body size	11
2.4. Cost of nutrition-related disease	12
2.5. Summary	13
CHAPTER 3. TRADITIONAL METHODS OF DIETARY ASSESSMENT	15
3.1. Overview	15
3.2. Dietary assessment	16
3.2.1 24-hour dietary recall	16
3.3. Food record	20
3.3.2 Food frequency questionnaires	22
3.3.3 Diet history	23
3.4. Sources of error in dietary assessment	24
3.4.1 Converting food to nutrients using a food composition database	26
3.5. Validity in dietary assessment	27
3.5.1 Biomarkers	27
3.5.2 Use of doubly labelled water to assess total energy expenditure	28
3.5.3 Validity of dietary energy intake data	31
3.5.4 Implications of measurement error	32
3.6. Summary	33
CHAPTER 4. UNDER-REPORTING REMAINS A KEY LIMITATION OF SELF-REPORT	RTED
DIETARY INTAKE: AN ANALYSIS OF THE 2008/09 NEW ZEALAND ADULT NUTRI	TION
SURVEY	34
4.1. Introduction to publication	34
4.2. Author contribution	34
4.3. Abstract	36

4.4. Introduction	37
4.5. Methods	38
4.6. Results	40
4.7. Discussion	48
4.8. Conclusions	50
CHAPTER 5. IMAGE-ASSISTED DIETARY ASSESSMENT: A SYSTEMATIC REVIEW	OF THE
EVIDENCE	51
5.1. Introduction to publication	51
5.2. Author contribution	51
5.3. Abstract	53
5.4. Introduction	54
5.5. Methods	55
5.5.1 Eligibility criteria	55
5.5.2 Exclusion criteria	55
5.5.3 Information sources and search strategy	56
5.5.4 Data extraction	58
5.5.5 Quality assessment	58
5.6. Results	58
5.6.1 Search results	58
5.6.2 Characteristics of included studies	60
5.6.3 Image-based food records	60
5.6.4 Image-assisted food records	64
5.6.5 Image-assisted 24-hour recalls	65
5.7. Discussion	74
5.8. Conclusions	
CHAPTER 6. FEASIBILITY OF A SENSECAM-ASSISTED 24h RECALL TO REDUCE U	JNDER-
REPORTING OF ENERGY INTAKE	
6.1. Introduction to publication	
6.2. Author contribution	
6.3. Abstract	
6.4. Introduction	80
6.5. Methods	81
6.5.1 Analysis	82
6.6. Results	83
6.6.1 Participants	83
6.6.2 Dietary recall	83
6.6.3 Feedback survey	85
6.7. Discussion	88
6.8. Conclusion	90
CHAPTER 7. WEARABLE CAMERAS CAN REDUCE DIETARY UNDER-REPORTING: D	OUBLY
LABELLED WATER VALIDATION OF A CAMERA-ASSISTED 24-HR RECALL	93
7.1. Introduction to publication	93

7.2. Author contribution	93
7.3. Abstract	95
7.4. Introduction	
7.5. Subjects and methods	
7.5.1 Study design	
7.5.2 Use of wearable camera	
7.5.3 Energy intake assessment	
7.5.4 Measurement of total energy expenditure	101
7.5.5 Resting energy expenditure	101
7.5.6 Statistical analysis	102
7.6. Results	102
7.7. Discussion	111
7.8. Conclusions	113
CHAPTER 8. THE USE OF A WEARABLE CAMERA TO CAPTURE AND CATEGOR	ISE THE
ENVIROMENTAL AND SOCIAL CONTEXT OF SELF-IDENTIFIED EATING EPISODES	114
8.1. Introduction to publication	114
8.2. Author contribution	114
8.3. Abstract	116
8.4. Introduction	117
8.5. Methods	118
8.5.1 Data collection	118
8.5.2 Data processing	119
8.5.3 Annotation protocol	120
8.5.4 Data analysis	120
8.6. Results	122
8.8. Discussion	129
8.9. Conclusion	132
CHAPTER 9. THESIS DISCUSSION	133
9.1.1 Contributions of this thesis	141
9.2. Future implications	141
9.3. Future research recommendations	144
9.4. Conclusion	149
APPENDICES	151
APPENDIX 1	153
Systematic review data extraction form	153
APPENDIX 2	157
Ethics approval letter for the feasibility study from the University of Auckland	d Human
Participants Ethics Committee	157
APPENDIX 3	161
Ethics approval letter for the feasibility study from the Inter-divisional researce	ch ethics
APPENDIX 4	165

Participant information sheet and consent form for the feasibility study	165
APPENDIX 5	173
Registration form for the feasibility study	173
APPENDIX 6	177
Brief information sheet for participants in feasibility study on photography in publ	ic places 177
APPENDIX 7	
Participant contact details for feasibility study	181
APPENDIX 8	185
Baseline demographics for feasibility study	185
APPENDIX 9	191
Feedback survey form for the feasibility study	191
APPENDIX 10	195
Ethical approval letter for the PICTURE study from the University of Au	ickland Human
Participants Ethics Committee	195
APPENDIX 11	201
Invitation letter for the PICTURE study	201
APPENDIX 12	205
Participant information sheet and consent form for PICTURE study	205
APPENDIX 13	213
Brief information sheet on photography for the PICTURE study	213
APPENDIX 14	217
SenseCam user instructions for PICTURE study	217
APPENDIX 15	221
Participant registration form for PICTURE study	221
APPENDIX 16	225
Participant contact details form for PICTURE study	225
APPENDIX 17	229
Participant characteristics form for PICTURE study	229
APPENDIX 18	233
Urine log form for PICTURE study	233
APPENDIX 19	239
24-hour dietary recall data sheet for PICTURE study	239
References	245

LIST OF FIGURES

Figure 1 Attributable burden (percentage of DALYs) for selected risk factors in New Zealand, 2006
Figure 2 Body size of New Zealand adults aged ≥15 years in 1997 compared to 2008/9 11
Figure 3 Outline of United States Department of Agriculture five step multiple pass 24-hour dietary recall method. ⁸⁶
Figure 4 Screenshots food record software applications for handheld devices, MyNetDiary, Easy Diet Dairy, and My Fitness Pal
Figure 5 Example of an estimated food record booklet
Figure 6 Elimination of doubly labelled water to assess total energy expenditure
Figure 7 Percentage of low energy reporters by body size using the WHO & Obesity International Taskforce BMI cut-offs ¹⁹¹
Figure 8 Percentage of low energy reporters by age group45
Figure 9 Percentage of low energy reporters by ethnicity 46
Figure 10 Percentage change of low energy reporters in the NZ Adult Nutrition Survey 2008-9 by body size, age, and ethnicity compared to the 1997 NZ National Nutrition Survey
Figure 11 MEDLINE search strategy for the systematic review examining the evidence for image- assisted methods of dietary assessment
Figure 12 Flowchart and inclusion process for the systematic review examining the evidence for image-assisted methods of dietary assessment
Figure 13 Procedure to estimate energy intake with the wearable camera
Figure 14 Under-reported energy intake in the 24h dietary recall alone (MP24) and with the assistance of the wearable camera (MP24+SC) compared to total energy expenditure (TEE) for men and women
Figure 15 Change to self-reported energy intake in the 24h dietary recalls (MP24) after viewing the wearable camera images (MP24+SenseCam), compared to total energy expenditure (TEE) for men and women
Figure 16 Agreement between mean self-reported energy intake (MP24+SenseCam) and TEE for men
Figure 17 Agreement between mean self-reported energy intake (MP24+SenseCam) and TEE for women (n=20)
Figure 18 Flowchart of study procedures to capture and categorise the environmental and social contexts of eating episodes using wearable camera images

LIST OF TABLES

Table 1 Intakes of major nutrients and fruit and vegetables associated with nutrition-relateddisease in New Zealand10
Table 2 Physical characteristics of New Zealand Adults in 1997 compared to 2008/9 11
Table 3 Summary of estimates for direct and indirect costs of nutrition-related diseases in New Zealand 12
Table 4 Baseline characteristics of participants 41
Table 5 Ratio of reported energy intake to estimate resting metabolic rate (EI:RMR _{est}) 42
Table 6 Studies that have evaluated or validated an image-assisted method of dietaryassessment compared to a reference method assessing dietary energy intake.68
Table 7 Characteristics for the ten participants included in the analysis 83
Table 8 Energy and macronutrient intakes from one interviewer-administered multiple pass 24- hour dietary recall (MP24), and the MP24 with the assistance of SenseCam images (MP24+SC).
Table 9 Energy provided from unreported foods, change to portion size or misreported foodsrevealed from one SenseCam-assisted interviewer administered MP24+SC.86
Table 10 Participant evaluation of the SenseCam-assisted 24-hour recall (MP24+SC). 87
Table 11 Demographic characteristics of the study sample. 103
Table 12 Distribution by day of the week for the dietary recalls collected for the total sample 104
Table 13 Daily energy intake (EI) measured in three Multiple Pass 24h dietary recalls (MP24 andMP24+SenseCam) and total energy expenditure (TEE) measured with the doubly labelled watertechnique.105
Table 14 Alterations in energy intake for all participants viewing wearable camera images aftercompletion of the multiple pass dietary recalls (MP24+SenseCam)
Table 15 Eating episodes identified by 40 participants during three wearable camera image-assisted 24h dietary recalls.122
Table 16 A comparison of eating episodes that could be annotated for context using SenseCamimages (n=528) compared to those that could not (n=214).123
Table 17 Meal and snack eating episodes (N=528) from 40 participants by location, environmentalcontext, positional context and social context.125
Table 18 Duration, energy intake and density of meals and snacks at home (N=312) viewingmedia screens from 40 participants.128

ABBREVIATIONS

AMPM	Automated Multiple Pass Method
ANS 08/9	New Zealand Adult Nutrition Survey 2008/9
BMI	Body Mass Index
BMR	Basal Metabolic Rate
CVD	Cardiovascular Disease
cm	Centimetre
DALY	Disability-Adjusted Life Year
DH	Diet History
DLW	Doubly Labelled Water
EI	Energy intake
g	Gram
FFQ	Food Frequency Questionnaire
FR	Food Record
HbA1c	Glycosylated Haemoglobin
IOTF	International Obesity Taskforce
HRC	Health Research Council
HRQL	Health Related Quality of Life
kcal	Kilocalorie
kg	Kilogram
kJ	Kilojoule
LERs	Low Energy Reporters
MJ	Megajoule
MP24	Multiple Pass 24-hour Recall
MP24+SC	Multiple Pass 24-hour Recall plus SenseCam
mpFR	mobile phone Food Record
MUFA	Monounsaturated Fat
NCDs	Non-communicable diseases
NIHI	National Institute for Health Innovation
NDNNS	National Diet and Nutrition Survey

NHANES	National Health and Nutrition Examination Survey
N.Z.	New Zealand
NZEO	New Zealand European
NZFCD	New Zealand Food Composition Database
NNS97	New Zealand National Nutrition Survey 1997
OECD	Organisation for Economic Co-operation and Development
PDA	Personal Digital Assistant
PICTURE	Passive Image Capture to Record Everyday Events
PUFA	Polyunsaturated Fat
REE	Resting Energy Expenditure
RFPM	Remote Food Photography Method
SAFA	Saturated Fat
SC	SenseCam
SD	Standard Deviation
SEM	Standard Error
TEE	Total Energy Expenditure
U.K.	United Kingdom
U.S.	United States of America
USDA	United States Department of Agriculture
WHO	World Health Organization
Yrs	Years

CHAPTER 1. THESIS INTRODUCTION

This thesis investigates the use of wearable cameras to (1) reduce the reporting bias for dietary energy intake (EI) associated with traditional self-reported dietary assessment, and (2) passively record and assess contexts of dietary behaviours. Dietary assessment methods are commonly used for monitoring population health and nutritional status, researching diet-disease relationships, informing public health interventions, guiding and evaluating health and nutrition policies, and informing clinical decisions and advice. ¹⁻⁷ However, due to the complexity of eating behaviours, reporting errors, and difficulties in objectively measuring dietary intake, traditional dietary assessment methodologies are inherently flawed. Currently no method exists that accurately assesses the dietary intake of an individual or a population in free-living settings. ⁸⁻¹⁰ This is of concern, as rates of non-communicable diseases (NCDs) around the world are predicted to rise over the next decade, ¹¹⁻¹³ and the major NCDs (including, cardiovascular disease, diabetes, obesity, stroke, and certain cancers), are all associated with poor diet. ^{3-5, 14}

The need for improved methods of dietary assessment is well recognised. ^{8, 9, 15, 16} Walter Willet, Professor of Nutrition Epidemiology at Harvard University, School of Population Health, describes the need for better research methods to study diet-disease relationships: *"our knowledge of many of these relationships will depend largely on epidemiologic data, and for many relationships indefinitely'. For this reason it is crucial to refine maximally our methods of data collection, analytic procedures, and interpretation of findings"*. ^{17(p19)} The implications sub-optimal methods and the associated measurement error is that it often attenuates the estimated disease relative risk (causes bias towards 1) and reduces the statistical power to detect an effect. Therefore an important relationship between diet and disease may therefore be obscured. ¹⁸

One of the most common methods of dietary assessment is the 24-hour dietary recall, a retrospective method usually conducted by a skilled interviewer (in person or over the telephone), but can be completed solely by a participant using an automated website or software package.^{2, 19-22} The 24-hour dietary recall is the method chosen for national nutrition surveillance programmes in New Zealand and a number of other high income countries including the United States of America (U.S.) and Australia.²³⁻²⁵ Additionally, the 24-hour dietary recall method is a popular method used in a wide-range of small and large-scale nutrition research, and in clinical

and community settings. ¹ The important role the 24-hour dietary recall plays in nutrition monitoring and assessment therefore means efforts to improve the method should be undertaken. Computer-assisted systems help to ensure correct methodical procedures are followed, and sophisticated portion size tools have improved portion size estimations.^{19, 20, 26, 27} However, these developments have not addressed key limitations of 24-hour dietary recalls, the reliance on self-report and memory. ^{1, 8, 28}

Five reviews over the past nine years have summarised, described, and evaluated the strengths and weaknesses of various technologies and computer-assisted systems for dietary assessment. ^{16, 19, 29-31} The reviews found technology-assisted methods are preferred by participants and can reduce the burden of research, but none has thoroughly examined image-assisted methods of dietary assessment and the use of wearable cameras. ^{16, 19, 29-31} Image-assisted dietary assessment is an emerging field of nutrition research, and encompasses any method that uses images/video of eating episodes to assist self-report used in traditional methods, or uses images/video as the primary record of dietary intake.

The central tenet of this thesis is to examine the feasibility and validity of image-assisted methods of dietary assessment, and to identify and minimise limitations of the 24-hour dietary recall method, using wearable cameras (image-assisted 24-hour dietary recall). Wearable cameras may improve 24-hour dietary recalls since images have been shown to enhance memory recall and provide an objective record of daily events. ^{32, 33} As such, the use of wearable cameras and self-report are potentially complementary; the wearable camera images provide visual prompts to augment memory recall ^{32, 33} and provide a visual representation to assist portion size estimation, ^{32, 33} while participant self-report provides specific details on dietary intake that are decipherable from the images (hidden ingredients). More accurate and complete recall of dietary intake during the measurement period should reduce the magnitude of measurement error. ^{8, 34} Additionally, wearable camera images may provide a novel method to passively and objectively assess the context of dietary behaviours in free-living settings. ³⁵⁻³⁹ Developing methods to better assess and understand the determinants of dietary behaviours. ⁴⁰⁻⁴⁴

The thesis objectives are as follows:

- 1. To evaluate self-reported dietary energy intake in the 2008/9 Adult Nutrition Survey and determine the prevalence of low energy reporters
- To undertake a systematic review of existing evidence for image-assisted methods of dietary assessment
- To explore the feasibility of using wearable cameras to assist an interviewer-administered 24-hour dietary recall
- 4. To validate a wearable camera image-assisted 24-hour dietary recall using the criterion measure doubly labelled water
- 5. To assess the utility of wearable cameras to objectively record and reliably assess environmental and social contexts of eating episodes

It was hypothesised that images captured by wearable cameras would reduce self-report bias for dietary EI compared to self-report alone by providing additional information to augment memory recall, and enhance self-report.

A brief overview of this thesis is as follows: Chapter Two provides general background on nutrition-related disease relevant to this thesis. The prevalence of nutrition-related disease, and risk factors, the associated cost and impact of nutrition-related disease, and recent national and global trends in dietary intake and body size will be discussed. Chapter Three provides an overview of dietary assessment. The traditional methods of dietary assessment, and associated sources of measurement error and validity, are discussed.

Five modules of research were conducted as part of this PhD and these form Chapters Four to Nine. Chapter Four is a retrospective analysis of EI data reported in the 2008/9 New Zealand Adult Nutrition Survey (ANS 08/9). The aim was to estimate the prevalence of low energy reporters (LERs) by gender, age, body size, and ethnicity in the most recent national nutrition survey. Chapter Five is the first systematic review of existing evidence for image-assisted methods of dietary assessment. The review included all relevant studies and technologies, and provides the first comprehensive examination of image-assisted methods of dietary assessment. Chapter Six describes a feasibility study that explores wearable cameras to enhance dietary self-report in the 24-hour dietary recall method (image-assisted 24-hour dietary recall).

Results from the feasibility study informed the design of a subsequent larger study called PICTURE "Passive Image Capture to Record Everyday Events", that validated wearable camera image-assisted 24-hour dietary recalls, using the criterion measure doubly labelled water (Chapter Seven). The study assessed the degree to which wearable cameras altered self-reported intake and how this affected measurement bias. Chapter Eight describes a body of work that uses the dietary intake data and images captured during the validation study (Chapter 7) to explore the potential use of visual images for research into human eating and nutrition behaviours in free-living settings. Chapter Nine discusses the findings of all the research undertaken investigating image-assisted methods of dietary assessment, summarises the potential of wearable cameras to reduce dietary reporting bias for dietary EI and assess context of dietary behaviours, and makes recommendations for future studies and development of image-assisted methods of dietary assessment.

CHAPTER 2. THE BURDEN OF NUTRITITION-RELATED DISEASE AND TRENDS IN DIETARY INTAKE

2.1. Overview

The following section provides an overview on the global burden of nutrition-related disease, and trends in dietary intake and body size in New Zealand (N.Z.). The studies and reports included in this section were obtained from searches of MEDLINE (1946 to September 2014), Web of Science, Google Scholar, and PubMed (to September 2014). The most recent national data available were used to describe the prevalence of nutrition-related disease and associated risk factors at the time of writing. Data regarding nutritional epidemiology and trends in population dietary intake (sections **2.2** and **2.3**) provide the rationale for the objectives of the thesis (Chapter One).

The chapter is structured as follows:

- Section 2.2 Epidemiology of nutrition-related disease
- Section 2.2.1 Prevalence of nutrition-related disease, and associated risk factors
- Section 2.3 Trends of dietary intake and body size
- Section 2.4 Cost of nutrition-related disease
- Section 2.5 Summaries of sections 2.2 to 2.4

2.2. Epidemiology of nutrition-related disease

Epidemiology provides evidence that bears directly on the health of the population ⁴⁵ and is defined by the World Health Organisation (WHO) as the study of the distribution and determinants of health-related states or events (including disease) and the application of this study to the control of diseases and other health problems. ⁴⁶ Nutritional epidemiology specifically focuses on diet-disease relationships, and the monitoring of the nutritional status of populations. ⁴⁷

Globally approximately 34.5 million out of 52.8 million deaths (almost two thirds) annually are due to non-communicable diseases (NCDs). ⁴⁸ When health loss is considered, 1.1 billion annual disability-adjusted life years can be attributed to NCDs. ⁵ The disability-adjusted life year (DALY) is a measure of overall disease burden, and can be thought of as the number of years lost due to ill-health, disability or early death. Predominant global nutrition-related NCDs, which include cancer (not all related to nutrition), cardiovascular disease (CVD), obesity, and diabetes mellitus, have reached epidemic proportions. ⁶ Globally the top five nutrition-related risk factors are high blood pressure, high blood glucose levels, overweight and obesity (high body mass index, BMI), high cholesterol, and low fruit and vegetable intake (middle and high income countries only). ⁴ Combined, these risk factors account for approximately 24.8 million (71%) of deaths from NCDs worldwide annually. ⁴ Regarding health loss, these major risk factors account for approximately 540 million or 50% of DALYs attributable to NCDs. ^{4, 5} Additionally, diets high in salt account for 61 million DALYs, and diets low in seafood omega-3 fatty acids, poly unsaturated fatty acids, and diets high in trans fatty acids account for 51.5 million or 2.9% of DALYS attributable to NCDs. ^{4, 5}

New Zealand

New Zealand is ethnically and culturally diverse, with a population of approximately four million residents (N=4,541,420 at 3rd July 2014)⁴⁹ and a median age of 38 years.⁵⁰ Information from the most recent census in 2013 indicates the population is ageing slowly, with a substantial proportion (14.3%) aged 65 years or older.⁵¹ Europeans are the largest ethnic group (74.0%), followed by Maori (14.9%), Asian (11.8%), Pacific peoples (7.4%), and other ethnic groups (3.7%, note people could select more than one ethnicity therefore values do not add up to 100). The majority of the population have a formal qualification (79.1%), and one in five (20.0%) have a bachelor's degree or higher. The average household disposable income is \$25,495 NZD (\$21,773 USD) which falls

Chapter 2. The burden of nutrition-related disease and trends in dietary intake

slightly below the average household disposable income (\$23,938 USD) for countries that belong to the Organisation for Economic Co-operation and Development (OECD).⁵²

In N.Z., the burden of nutrition-related disease is comparable to that described globally. ⁵³ Cancer (17.5%), coronary heart disease (9.3%) overweight and obesity (7.9%) and diabetes (3.0%) are major attributable causes of total estimated DALYs. ⁵⁴ **Figure 1** shows the dietary risk factors high blood pressure, high cholesterol, overweight and obesity, inadequate fruit and vegetable intake, and high salt intake are attributable for approximately 24.8% of total DALYs. ⁵⁴



Figure 1 Attributable burden (percentage of DALYs) for selected risk factors in New Zealand, 2006.

Figure taken from Health Loss in New Zealand. A report from the New Zealand Burden of Diseases, Injuries and Risk Factors Study, 2006–2016 ⁵⁴

2.2.1 Prevalence of nutrition-related disease in New Zealand

This section briefly discusses the prevalence of the four major nutrition-related NCDs in N.Z.; cancer, CVD, obesity, and diabetes. ⁵⁴ Cancer is the largest cause of death but it is common to report the annual incidence of cancer using an age-standardised rate for the total population rather than the prevalence. In 2009, the incidence of cancer in N.Z. was 344/100,000 people. ⁵⁵ With respect to CVD the most recent data for N.Z. are reported for all persons rather than adults,

therefore the prevalence 7% appears relatively low. ⁵⁶ However, 2005 projections from the 2003/4 Auckland Diabetes Heart and Health study suggested 20% of adults aged ≥35years are likely to have CVD. ⁵⁷ Regarding obesity, the prevalence amongst N.Z. adults aged ≥15 years is 28%, which is one of highest rates the world. ⁵⁸⁻⁶⁰ Compared with all other nations in the OECD, N.Z. has the fourth highest mean BMI. ⁶⁰ Regarding diabetes (type 1 and type 2), the most recent data available are from the Adult Health Survey 2011/12, which indicated that the prevalence of diabetes is 5%. ⁵⁸ However, blood analytes measured in the Adult Nutrition Survey 2008/9 (ANS 08/9) indicated that 6.9% of adults aged ≥15years had a proportion of glycosylated haemoglobin (HbA1c) of ≥6.5%, which is indicative of diabetes. ⁶¹

2.2.1.1. Prevalence of nutrition-related risk factors

As discussed above, high blood pressure, high cholesterol, inadequate fruit and vegetable intake, and high salt intake are the leading nutrition-related risk factors in N.Z. attributable to total annual DALYs. ⁵⁴ The only national data for prevalence of high blood pressure (adults \geq 15 years) is based on self-reported use of medication for hypertension (16%), reported in the 2012 N.Z. Adult Health Survey.⁵⁸ However, data from the 2003/4 Auckland Diabetes, Heart and Health Study (≥35 years) that measured blood pressure indicated 22% of N.Z. European (NZEO) (n=2021), 37% of Maori (n=1006), and 38% of Pacific people (n=996) had high pressure (population estimates not reported). ⁶² Similarly, the most recent data on the prevalence of high cholesterol (10%) is based on self-reported use of medications to reduce cholesterol reported in the 2012 N.Z. Adult Health Survey, ⁵⁸ which is substantially lower than the 68% of NZEO adults (n=2021) with high cholesterol measured (total cholesterol ≥5mmol/L) in the Auckland Diabetes, Heart and Health Study (2003/4). ^{63, 64} With respect to fruit and vegetable intake, the most recent data available is based on self-reported intake in the 2012 N.Z. Adult Health Survey with 41% not meeting recommendations for fruit (2 servings) and 32% not meeting recommendations for vegetables (3 servings). ⁵⁸ For sodium (principally from salt) intake, the most recent available data are from the ANS 08/9 which used measured urinary sodium. 65 At 3500mg/day N.Z. adults consume more than twice the recommended suggested target for reducing chronic disease (1600mg/day) set by the N.Z. Ministry of Health and the Australian Government. ⁶⁶

2.3. Trends in dietary intake

For N.Z. adults, dietary intakes of key nutrients associated with nutrition-related disease are suboptimal compared with recommendations (presented in **Table 1**). It is important to note that data regarding dietary intake are obtained using self-reported methods of dietary assessment, and thus prone to bias (discussed in section **3.5**). ^{2, 8, 9}

Energy intake is an important factor in nutrition-related disease, as it is the surrogate measure of the total quantity of food intake with many nutrients (macronutrients, mineral and B vitamins) highly correlated with its intake. ^{8, 67} It is difficult to compare EI with recommended intakes as age, body composition, and levels of physical activity all affect energy demands. ⁶⁸ Furthermore, substantial under-reporting of EIs observed in national nutrition survey data make it difficult to accurately monitor and interpret trends in EI. ⁶⁹⁻⁷³ However, with this in mind reported EIs for N.Z. adults decreased substantially for men and slightly for women (statistically non-significant) between 1997 and 2008/9. ^{65, 74} Regarding saturated fat intake, N.Z. adults have substantially reduced their intake over the past decade but with 13% of total energy provided by saturated fat it is still greater than recommendations. ⁶⁵ Trends for sodium intake are more difficult to assess as urinary sodium was not assessed in the 1997 national nutrition survey. However, data from a urinary excretion study (n=700) in 1998 suggests sodium intake has remained stable at ~3500mg/day over the past decade. ^{66, 75} The inadequate intake of fruit and vegetables in N.Z. was discussed above in section **2.2.1.1**. Data from 2003-2012 suggest that the inadequate intake of fruit and vegetables have remained relatively stable over the past nine years. ^{58, 61, 76}

Characteri stic	Method	Recommendati	Me	In daily intake		
ollo	Assessm ent	classifications	1997* 200		08/9*	
Energy						
≥15yrs	1x 24- hour dietary recall	7.7MJ to 18.6MJ/day (Men) 6.1MJ to 15.3MJ/day (Women)	12.0MJ 8.0MJ	10.7MJ 7.6MJ		▼ NC
Saturated Fa	at	()				
≥15yrs	1x 24- hour dietary recall	≤12% of total energy ⁷⁷	15% of total 13% energy			•
Sodium						
≥15yrs	Urinary sodium	Recommended Upper Level = 2300mg/day Suggest dietary target for reducing chronic disease = 1600mg/day ⁶⁶	3464mg/day 3500mg/day (1998, ≥18yrs)^			NA
Fruit & Vegetable			2003/4 [#]	2006/7 [#]	2011/12 [#]	
≥15	self- reported	Recommended servings/day Fruit ≥ 2 vegetables ≥ 3	45% not meeting recommend ations for fruit 31% not meeting recommend ations for recommend ations for vegetables		41% not meeting recommendati ons for fruit 32% not meeting recommendati ons for vegetables	NA
** NC = no	statistical of	difference, 🔻 = sigr	nificant decrea	ase, 🛦 = signif	icant increase,	NA = not

Table 1 Intakes of major nutrients and fruit and vegetables associated with nutrition-related disease in New Zealand

applicable.

*Data taken from the 1997 National Nutrition Survey 74 and the 2008/9 Adult Nutrition Survey of N.Z. 65

^Sodium intake was not assessed in the 1997 National Nutrition Survey therefore data was taken from a N.Z. urinary excretion study conducted in 1998 (n=700). 75

[#]Data taken from the N.Z. Health Surveys. ^{58, 61, 76}

2.3.1 Trends in body size

Trends in body mass and obesity in N.Z. are presented in **Table 2** and **Figure 2**. Mean body weight and prevalence of overweight and obesity in N.Z. adults have increased substantially between national nutrition surveys. ⁶⁵ This indicates N.Z. adults are in positive energy balance, despite decreases in reported EIs. ⁶⁵ Body size data from the 2011-12 N.Z. Health survey suggests the trend is continuing (Male obese = 28.1%, female = 28.8%). ⁵¹ Similarly, data for the U.S. and U.K show an increasing BMI trend, and are predicted to rise further. ⁷⁸

Table 2 Physical characteristics of New Zealand Adults in 1997 compared to 2008/9

Body Weight	1997	2008/9	Change
Male	80.4kg	85.1kg	A
Female	68.7kg	72.6kg	A





Figure 2 Body size of New Zealand adults aged \geq 15 years in 1997 compared to 2008/9. Data taken from the 1997 National Nutrition Survey ⁷⁴ and the 2008/9 Adult Nutrition Survey of N.Z. ⁶⁵



2.4. Cost of nutrition-related disease

The costs associated with nutrition-related disease in N.Z. are considerable. ⁷⁹ Economic costs can be direct or indirect. Direct costs include hospital costs (inpatient and outpatient), allied health professional costs, general practitioner visits, residential/aged care, pharmaceuticals and laboratory costs, or any other directly related cost. Indirect economic costs or productivity losses are the costs associated with lost opportunities or an impaired ability to work. ⁸⁰ There is limited information regarding direct and indirect costs of nutrition-related disease in N.Z., with the majority of information sourced from a recent study by Lal et al. ⁸¹ The direct and indirect cost of nutrition-related diseases are presented in **Table 3**. Costs per capita were calculated using population estimates corresponding to the year of source data and the 12-month average exchange rate for the N.Z. Dollar (NZD). Overall the total direct costs for the five types of cancer, CVD, obesity, and diabetes were estimated by Lal et al. ⁸¹ were \$1.6 billion annually.

Source	Direct Costs (Million \$NZD)	Indirect Costs (Million \$NZD)	Total Costs (Million \$ NZD)	Cost per capita*
Cancer				
(2006)				-
Colorectal	58	-	58	Direct cost =\$14
Breast	46	-	46	Direct cost =\$11
Endometrial	5	-	5	Direct cost =\$1
Kidney	8	-	8	Direct cost =\$2
Cardiovascular Disease Lal et al ⁸¹ (2006)				
Ischaemic heart disease	371	-	371	Direct cost =\$89
Hypertensive heart disease	4	-	4	Direct cost =\$1
Obesity Lal et al ⁸¹ (2006) (overweight and obesity)	624	-	-	Direct cost = \$149
New Zealand Government ⁸² (2004)	460	370	830	Direct and indirect costs =\$203
Diabetes Lal et al ⁸¹ (2006) Type 2 only	526	-	526	Direct and indirect costs =\$125
PriceWaterhouseCoopers and Diabetes New Zealand ⁸³ (2008) Type 2 only	600	-	-	Direct Cost = \$140

Table 3	Summary	of	estimates	for	direct	and	indirect	costs	of	nutrition-related
disease	s in New Ze	eala	and							

*Population counts are taken from the estimated populations in June for the corresponding year ⁵⁰ Costs are approximate.

2.5. Summary

The prevalence of nutrition-related disease (cancer, CVD, obesity, and diabetes) and associated risk factors (blood pressure, cholesterol, inadequate fruit and vegetable intake, high salt intake) in N.Z. is already high, but is predicted to rise further. At present there is a vast economic, personal and social burden attributed to nutrition-related disease. Dietary data indicates N.Z. adults are in positive energy balance (increasing in weight), are consuming higher than recommended intakes of saturated fats and sodium, and are consuming inadequate servings of fruit and vegetables, which further contribute to nutrition-related disease. In contrast to these trends, self-reported EI data for N.Z. adults has reduced for males and remained stable for females.

METHODS

OF

3.1. Overview

The following section provides an overview of the traditional methods of dietary assessment and their associated strengths and limitations. This section also includes information on the common sources of error in dietary assessment, validity of the traditional methods of dietary assessment, and discusses the implications of measurement error. The sources for this information were obtained from MEDLINE (1966 to September 2014), Web of Science, Google Scholar, and PubMed (to September 2014). The limitations of the traditional methods of dietary assessment, and implications of measurement error, provide the rationale for investigating the use of wearable cameras to reduce the reporting bias for dietary EI.

This literature overview is structured as follows:

Section 3.2 provides an overview of the traditional methods of dietary assessment

Section 3.4 summarises the common sources of error in dietary assessment

Section 3.5 describes methods used to assess validity, common results from reviews of validation studies, and discusses the implications of measurement error

Section 3.6 summarises sections 3.2 to 3.5

3.2. Dietary assessment

The traditional methods of dietary assessment have been primarily designed to estimate the quantity and/or frequency of food consumption, and are essential to evaluate individuals' or populations' dietary intake and nutritional status. ^{2, 84} Four traditional methods of dietary assessment have been used extensively in nutrition research and in clinical and community settings: 24-hour dietary recall, food record (FR), food frequency questionnaire (FFQ), and the diet history interview (DH). ^{1, 2, 84} Each method has specific strengths and limitations, making them appropriate for use in different situations (discussed in sections **3.2.1** to **3.3.3**). However, all methods rely on self-report without the ability to verify intake objectively (at the time of assessment). Therefore, both unintentional and intentional misreporting can contribute undetected to measurement error. ⁸⁴

Although traditional methods of dietary assessment are primarily designed to record dietary intake, the 24-hour dietary recall and FR can also be used to collect additional limited information on the context of eating episodes (meals and snacks), such as time and location (e.g. food eaten at home vs. at work). ^{85, 86, 87, 88} Context describes the interrelated conditions in which something exists or occurs, ⁸⁹ but at present there is no standard terminology for defining context or classes of contextual variables for dietary intake. ⁴² Both lab-based and real-world studies investigating human eating and nutrition behaviours have shown that many contextual factors affect dietary intake; such as meal time, location, effort required (e.g. distance of food away from table), cost, décor, ambient music, plate size, television viewing, social interaction, positive and negative cues (e.g. suggestions provided regarding the popularity of a food), meal frequency, and meal duration. ^{41, 42} Equally, contextual information can also reveal where and when certain foods are likely to be obtained or consumed. ⁹⁰ As such, information on contexts is now recorded in some large-scale dietary surveillance programmes in conjunction with details of the foods and beverages consumed. ^{85, 86, 91}

3.2.1 24-hour dietary recall

The 24-hour dietary recall is a retrospective method of dietary assessment usually conducted by a skilled interviewer (in person or over the telephone), but it can be completed solely by a participant using an automated website or software package. ^{2, 19-21} This flexibility makes the 24-

hour dietary recall suitable for a range of participants including those with low literacy and people with mental and physical disabilities.¹

Typically, the interviewer follows a standardised format to prompt participants to recall all foods and beverages consumed over the previous 24-hours. The 24-hour period is usually reviewed multiple times to elicit further information from the participant, and this is referred to as the multiple pass 24-hour dietary recall (MP24). As described above, dietary recalls also allow some contextual information on each eating episode to be recorded, such as location (discussed in section **3.2**). ¹¹ The retrospective nature of dietary recalls places a reduced burden on participants compared to the FR, as there is no need to record any information prior to the assessment, and this therefore reduces the possibility of participants to change their dietary behaviour. ¹

Research-specific software is not required to conduct the 24-hour dietary recall interview but is commonly used in large-scale research to help standardise data collection and reduce the burden of analysis. ^{23, 86, 92} The United States Department of Agriculture (USDA) developed the Automated Multiple Pass Method (AMPM) ^{24, 86} for use in The National Health and Nutrition Examination Survey (NHANES), which uses five distinct steps (see **Figure 3**). The first step asks participants to recall all the foods and beverages consumed for the previous day. The second step encourages additional information with a "forgotten foods" list to prompt recall of unreported foods. Successive passes gather additional details, such as time and occasion, food brand, portion size, method of cooking, and condiments added. A final review is then conducted to check all the information is correct.

Nutrition surveillance programmes in N.Z. and Australia incorporate the MP24 and have adopted the method using bespoke software but differ slightly in the number of steps and details collected at each successive step. ²⁵ ^{23, 93} In N.Z., the LINZ24 MP24 software developed by the University of Otago has been used in two previous adult national nutrition surveys (NNS97 and ANS 08/9). ^{23, 94} The LINZ24 system was also adapted for use in the 1995 Australian National Nutrition Survey, ⁹³ but Australia recently developed their own system based on the USDA AMPM for the most recent Australian Healthy Survey 2011/13. ²⁵

Software has also been developed to assist 24-hour dietary recalls for a range of both large and small scale research. An early example is EPIC-SOFT developed for the European Prospective Investigation into Cancer and Nutrition, used to standardise the 24-hour dietary recall procedures across 23 different research centres in 12 European countries. ⁹² Similar to other software, EPIC-

SOFT assists an interviewer to follow the structured method and consists of a quick list followed by detailed steps and probing questions to gather further details. ⁹² For small scale research some nutrient analysis software packages, used to convert foods into energy and nutrients (discussed in section **3.4.1**), include a 24-hour recall feature to assist researchers. More recently, web-based systems have been developed (for both large and small scale research), such as the National Cancer Institute ASA24, which can be easily modified for any country.

3.2.1.1. Limitations of the 24-hour dietary recall

Due to the wide variation in an individual's daily dietary intake, use of a single 24-hour recall is only appropriate to describe the average dietary intake of a group, not individuals. ^{1, 95} As a retrospective method, the portion size of foods and beverages must be estimated, and details regarding the brand and variety may be forgotten or misreported. Moreover, the accuracy of the recall can be affected by interviewer bias and social desirability bias (discussed in section **3.4**). ^{2, 96} Overall, the data obtained from the 24-hour recall method usually under-estimates dietary intake, especially amongst the elderly, women and overweight/obese respondents (specific details regarding the validity of the 24-hour dietary recall are presented in section **3.5**). ^{8, 9, 70, 73, 97}



Purpose

Step one is designed to get a quick report of easily remembered foods consumed over the previous 24-hour. Cues are given to help recall the day's events, but the participant is given an open opportunity to recall foods consumed.

In step two participants are encouraged to think about specific categories of foods that are frequently forgotten e.g. non-alcoholic and alcoholic beverages, sweets, savoury snacks, fruit & vegetables, cheeses, breads and rolls, and any other foods.

Step three is designed to encourage additional recall by helping respondents think about their eating patterns over the past 24 hours. This step also sorts foods into chronological order and eating episodes / meals to assist the detailed review.

Step four records a detailed description of each food consumed, including portion size and additions to the food e.g. sauces or salt. The foods source (e.g. restaurant or food outlet) and location of consumption (e.g. eaten at home) are also recorded. Both the identified eating episode and periods between the eating episodes are reviewed to elicit any additional recall.

Step five provides a final opportunity to recall foods. Participants are encouraged to recall small amounts of food that they may have considered not important or opportunistic (e.g. biscuits offered by work colleague).

Figure 3 Outline of United States Department of Agriculture five step multiple pass 24-hour dietary recall method. ⁸⁶

3.3. Food record

The FR is a self-reported prospective record of all food consumed by a person over a given time period (usually 3 to 7 days), and is used widely in community and clinical settings, and small and large-scale nutrition research. ^{1, 2} Typically, participants are required to record everything they eat or drink for 1 to 7 days. Quantities of foods and beverages recorded are either weighed or estimated (using portion size guides) and training is usually provided to ensure thorough records are kept. ¹ Traditionally, FR's were completed by participants using pen and paper in booklets. An example of the FR booklet used for the National Diet and Nutrition Survey (NDNS) in the United Kingdom (U.K.) is presented in **Figure 5**. ⁹⁸ However, electronic food records have been developed for hand-held devices, personal digital assistants (PDAs), smartphones and tablet computers. ⁹⁹⁻¹⁰¹ More recently consumer friendly FR apps have been developed for smartphones, such as MyNetDiary, Easy Diet Diary, and My Fitness Pal shown in **Figure 4**.

			0 후 🗎 3:44	iPod ᅙ		12:28 pm		* 🗖				£ 🔵	465 🖌 💼 2-52
👖 Meals 💼 🗊			Frid	day 23rd Au	ig					70 V	A 3.33		
<		Today	>			Today			E 🔭 Di	ary			1 +
۲	늘 Breakfast		441 cals 86g carbs	S 8	602 kJ emaining		energy	3909 kJ	Goal	Food	Exercise	Net	Remaining
ര്	💽 Lunch		378 cals 42g carbs						1210	1011	205	192	410
	Dinner		84 cals	BREAKFAST			energy	2972 kJ	•	TODAY			►
ă	Snacks		3g carbs 394 cals 76g carbs	V My Sen 1 Serve	nolina Por 394g	ridge		2606 kJ	BREAKF	AST			318 CAL 📌
X	Nutrition Summary	nmary		Semolina (raw) 1/2 cup				1090 kJ	Nonfat Greek Strained Yogurt				130
6	Nutrient	Consumed	Left	Milk (cow) 1 cup				744 kJ	Fage, 1.0 cup (227 g), 130 cal				
	Calories	1296	718	Sultana 1/4 cup				772 kJ	Strawberries - Raw			49	
-	Fat, g	32	12	Banana (fresh)				000	Pread - Whole wheat togeted				
~	Carbs, g	207	20	1 medium 101g				366 kJ	2.0 slice, 138 cal				139
1	Protein, g	59	117	~		<u> </u>			LUNCH				241 041 6
	Trans Fat, g	0	3	Q		ΘΞ			LUNCH				241 GAL
	Saturated Fat, g	7	11						Guacamole Medium				50
MyNe	Poly Unsat Fat, g	3	22	 LUNCH 	4		energy	314 kJ	2.0 Tbsp, 5	0 cal			
ଁ	Mono Unsat Fat, g	3	22						Cilantro L	ime Vinaig	rette Dressi	ng	180
	Cholesterol, mg	95	105	DINNER			energy	0 kJ	liceme Sticke				
	Sodium, mg	1508	-8	SNACKS & DRINKS				623 kJ	JICAMA STICKS Balev's, 1,0 oz., 11 cal			11	
-	Net Carbs, g	180				NKS	energy						120.041 6
A	Fiber, g	27	6						SNACK				120 CAL
~	Sugars, g	127	21	• EXERC	ISE		energy	1311 kJ	Triscuit R	educed Fa	t		120
													120

Figure 4 Screenshots food record software applications for handheld devices, MyNetDiary, Easy Diet Dairy, and My Fitness Pal.

Chapter 3. The traditional methods of dietary assessment

As the FR is the only method of dietary assessment which attempts to record dietary intake prospectively (at the time food is consumed), the FR has been referred to as the gold standard measure (seven-day weighed FR) and is used as a reference method to validate other methods of dietary assessment. ¹ However, the data obtained from the FR is far from optimal, as discussed in section **3.3.1.1**.

As described above in section **3.2**, the prospective nature of the FR makes it suitable to record the context of eating episodes, and is the recommended choice in research of human eating and nutrition behaviours in free-living settings. ^{85, 87, 88} Moreover, since 2008 the NDNS has included an extra column in the FR booklet for participants to record contextual information on the location (where?), whether there was social interaction (with whom?), whether television was viewed (TV on?), and seating position (at table?) of the eating episode (see **Figure 5**). ⁸⁵

Day: Thurs		Date: 29 th June 2014							
Time Where? With whom? TV on? At table?		Food & drink Description	Brand Name	Portion size or quantity <u>eaten</u>					
6am to 9am									
6:30am	Kitchen Alone No TV Not at Table	Filter Coffee, decaffeinated Milk (fresh, trim) Sugar white	Nescafe Anchor	250ml 10ml 1 tsp					
7:30am	Kitchen Partner TV on At Table	Filter coffee with milk and sugar Cornflakes Milk (fresh, trim) Toast, Multigrain Marmite	As above Kellogg's Anchor Tip Top Sanitarium	As above Bowl (3/4 cup) ½ cup 1 Slice 1 heaped tsp					
12:00	At work Alone No TV At Desk	Banana Yoghurt Coke	Fresh Yoplait – Lite berry Zero	Medium 1 tub 1 x355ml can					
9am to 12 noon									

Figure 5 Example of an estimated food record booklet, adapted from the NDNNS with an additional column for contextual information: Where? With whom? TV on? At table? ⁸⁵



3.3.1.1. Limitations of the food record

Similar to the 24-hour dietary recall the FR is self-reported without any means to objectively verify dietary intake (at the time of assessment). Moreover, 1-day FRs like 24-hour dietary recalls are only appropriate to describe the average dietary intake of a group, not individuals, due to the wide variation in an individual's daily dietary intake. ^{1, 95} A selection bias (non-response) is possible as participants must be literate, and a high burden is placed on participants due to the level of detail, concentration, and time required for completion of the FR correctly. ^{1, 102} Investigations have demonstrated that the quality of the data obtained from FRs diminishes over successive days of assessment. ^{2, 102} Early studies using electronic food records revealed the majority of participants record their meals between 3-6 hours after consumption, thus the FR can still be prone to similar limitations of retrospective methods i.e. memory lapses, and incorrect estimation of portion sizes. ¹⁰³ The prospective design of the FR is also prone to respondent biases such as changes in dietary behaviour, under-recording of food intake, and forgetting to record foods. ¹⁰⁴ Together, these limitations affect the quality of the data obtained when using FRs, and often result in underestimation of dietary intake. ^{8, 9} Specific details on validity of the FR method are presented in section **3.5**.

3.3.2 Food frequency questionnaires

Food frequency questionnaires assess the frequency with which certain foods are consumed over a standard period of time (e.g. one month), using questions such as "how often over the past month did you eat fish"? A limited number of possible responses are available. The FFQ was traditionally an interviewer-led or self-administered paper-based questionnaire, but can be completed by a parent or caregiver proxy, or conducted over the telephone, or the internet. ² A FFQ designed to assess the full diet usually contains approximately 120-180 pre-specified food items (questions) but can be shortened to focus on specific nutrients and food groups. ^{105, 106} The frequency at which certain foods are consumed can be used to rank individuals according to nutrient intake, or monitor changes in usual intake, and semi-quantitative FFQs that incorporate portion size questions also allow estimates of energy and nutrient intakes to be derived. ^{1, 107} The strength of the FFQ is therefore its ability to assess a participant's habitual intake of foods, and is also better suited to capture intake of episodically consumed foods/seasonal foods such as, strawberries or mandarins. Further, being a retrospective method, it does not influence dietary

behaviours like the FR (though participants can still misreport intake). ¹ Moreover, FFQ has a low participant burden compared to the FR, and the limited number of pre-specified responses make the FFQ easy to analyse compared to other methods. ^{1, 2} These attributes make the FFQ a popular choice in dietary surveillance programmes, diet-related intervention studies, and prospective and retrospective research investigating diet and disease relationships. ^{2, 107}

3.3.2.1. Limitations of food frequency questionnaires

Food frequency questionnaires lack accuracy compared to other methods, as many details regarding dietary intake are not captured by the limited possible responses. Respondent biases and memory lapses are problematic, as it can be challenging to estimate how frequently certain foods are consumed, especially with foods only eaten occasionally. ¹ Further inaccuracies result from incomplete lists of foods in the questionnaire, and FFQ must be tailored and validated for specific population groups, otherwise it may not contain the most appropriate and commonly consumed foods for the population assessed. Therefore, data obtained from the FFQ method should only be considered an approximation. ^{1, 2} Additionally, information on context cannot be recorded as the FFQ captures information on habitual dietary intake, not specific eating episodes.

3.3.3 Diet history

The DH method is usually conducted by a skilled interviewer and comprises a range of questions, which can vary depending on the exact technique used, to assess an individual's historical and habitual dietary intake. ¹⁰⁸⁻¹¹⁰ However, computer-assisted systems for automated self-report have also been developed. ¹¹¹ The DH typically obtains information on commonly consumed meals and snacks, cooking methods, and the frequency with which certain foods and food groups are consumed. ^{1, 2} The major strength of the DH is the assessment of habitual intake, rather than daily intake, as assessed by the 24-hour recall and FR. Therefore, the DH is a popular method for dietitians and nutritionists in clinical and community settings, ¹ but in recent years has become less common in nutrition research, and is not commonly used in dietary surveillance programmes due to the limitations discussed below.

3.3.3.1. Limitations of a diet history

Substantial interviewer biases may arise due to the variability of skills and techniques used to obtain dietary intake information, and the lack of standardisation can make comparisons between

studies challenging. ^{1, 96} Respondent biases are also problematic as many judgements are required regarding the frequency of consumption of different meals or food. ¹ Social desirability biases may affect reported intake during the interview, and, as it is a retrospective method, lapses in memory may cause further errors. ^{2, 96} The validity of different DH techniques is also hard to assess as there is no independent reference measure for habitual intake. ^{1, 28} Additionally, information on context cannot be obtained as the DH assesses habitual dietary intake, not details of specific eating episodes.

3.4. Sources of error in dietary assessment

There are many potential sources of error in dietary assessment that can be random or systematic.⁹⁶ Random errors occur with all respondents/participants but can be minimised by increasing the number of observations and/or days diet is assessed. ^{8, 84} In contrast, systematic errors may be associated with particular respondent/participant characteristics (e.g. among people with obesity), with certain foods (e.g. socially undesirable snack foods), or with specific interviewers conducting the dietary assessment. ⁸⁴

Extensive work has been conducted to help minimise the potential sources of error in dietary assessment. These include standardising data collection with software and websites, ¹⁹ and developing sophisticated portion size guides ^{22, 27} or food atlases (books that contain images of commonly eaten foods in various portion sizes). ¹¹² However, the traditional methods still rely on self-report without the ability to verify intake, ^{112, 113} therefore systematic biases in dietary studies are problematic. ^{1, 84, 114} The common sources of measurement error are briefly described below.

Non-response bias

Non-response or selection bias may arise from use of a sample that does not reflect the population of interest. This can occur due to restraints of the research design, recruitment method, or resources. ⁹⁶ Moreover, the associated burden of dietary recording for participants can cause a reduced response rate, ^{115, 116} and therefore people who do choose to participate may not reflect the population of interest.

Respondent bias

Respondent bias may occur if a participant provides socially desirable answers to avoid criticism (social desirability and approval biases), under or over-reports food intake, or changes dietary

behaviours during the period of assessment, thereby altering dietary intake. The psychosocial and behavioural characteristics related to respondent bias have been extensively reviewed with a range of characteristics commonly associated with bias: lower leisure physical activity, increased social desirability, fear of a negative evaluation, body size dissatisfaction, recent weight loss, fluctuation of body weight, attempted weight loss in previous 12 months, eating restraint, and eating disinhibition (overeating/loss of self-control over hunger).²⁸

Interviewer bias

Interviewer bias may result if different interviewers use different techniques to probe for information to varying degrees, intentionally omit questions (to save time), or record responses incorrectly. ⁹⁶ Thorough training regimens and audits of interviewer techniques can be used to ensure interviewer bias is minimised. ¹¹⁷ Furthermore the development of computer-assisted methods commonly used in large-scale research has helped to minimise interviewer bias. ^{26, 86}

Incorrect portion size

Errors in portion size can arise in all methods of dietary assessment as participants may fail to record estimate and/or record the portion size correctly. Other errors may arise due to misconceptions of what is considered a "normal" portion size, and differences in food characteristics and volume make some foods more difficult to estimate accurately than others, such as single foods items and complex mixed dishes. ^{96, 118 27} Reduced accuracy in portion size estimation has also been shown among obese people. ¹¹⁹

Supplement usage

Food and nutrient databases usually contain a limited number of dietary supplements, but due to the number of products on the market and the large differences in dosage and ingredients, supplement intake is inherently hard to assess. The errors associated with supplement intake affect both macronutrients (e.g. protein powders or oil-based supplements) and micronutrients (e.g. vitamin and mineral products).

Coding errors and mixed dishes

Coding errors are errors that arise when converting foods into nutrients due to the limited number of foods present in nutrient databases (the coding errors are discussed below in section **3.4.1.1**.). Mixed dishes are problematic to assess as the specific ingredients and portions may be unknown.

3.4.1 Converting food to nutrients using a food composition database

The dietary intake data (e.g. foods and beverages) recorded during dietary assessments is converted into energy and nutrient values using food composition databases. Food composition databases are usually country specific and provide the macronutrient (energy protein, carbohydrate, and fat) and micronutrient (e.g. vitamins and minerals) values of local foods. The values for every food and nutrient are derived by chemical analysis or can be estimated from other databases. ¹²⁰ The first nutrition composition tables (paper-based) were published in 1896 by Atwater et al. ¹²¹ More recently, electronic formats are available for many countries and are often freely available to view online, or can be accessed and manipulated using nutrition-specific software for nutrient analysis.

There are a range of software packages available to analyse dietary intake data, but these are usually region-specific or only have access to a select number of food composition databases. The software accesses the appropriate database and codes foods and beverages weight/volume with the corresponding macronutrient and micronutrient values. In N.Z., the most common nutrient analysis software is FoodWorks (Xyris software, Queensland, Australia) and was used for the PICTURE study reported in Chapter 7. FoodWorks accesses the electronic version of the N.Z. Food Composition Database (NZFCD) called FoodFiles. ¹²² The NZFCD is N.Z.'s most comprehensive food composition database and contains information on nutrient data for 59 nutrient components of 2710 commonly consumed foods. The NZFCD is maintained by the N.Z. Institute for Plant and Food Research Limited in partnership with the Ministry of Health. In the U.K., WISP (Tinuviel Software, Warrington, United Kingdom) is a common nutrient analysis software and was used for the UK-based feasibility study reported in Chapter 6. WISP accesses the Composition of Foods Integrated Data Set, which is maintained by the Institute of Food Research in partnership with the Biotechnology and Biotechnology Sciences Research Council. The dataset contains nutrient information for 3423 foods and was last updated in 2002. ¹²³ An updated version is scheduled for 2014. 123

3.4.1.1. Sources of error associated with food composition databases

Although food composition databases are the only method researchers can use to derive nutrient intakes from dietary data, there are several limitations of food composition databases that need to be acknowledged. Firstly, food databases only contain a limited number of foods, and thus may not contain all foods consumed by respondents. In such circumstances a similar food must be chosen. Secondly, there is a constant change in the food supply as manufacturers produce new products and reformulate existing products. Therefore, it is a challenge to keep the databases current and many manufactured foods are not included. Thirdly, seasonal variations, changes to agricultural practices and natural variability in natural and manufactured products can change the nutrient composition of foods. ^{120, 124} Fourth, coding errors may result when mixed dishes are reported (or recorded) due to the ingredients in the mixed dishes being different than the ingredients used in the food composition database. Lastly, different food composition databases vary in how foods are described, grouped and analysed. Therefore, it is difficult to make comparisons across databases and/or countries. ¹²⁵.

3.5. Validity in dietary assessment

Validity describes the degree to which a method of dietary assessment measures what it is intended to measure. Errors that affect the validity of dietary assessment methods are usually systematic. ^{84, 126} The relative or concurrent validity of a method can be assessed by comparing the method of interest with another method of dietary assessment, usually FR and 24h-hour dietary recall. ^{84, 127} However, caution must be taken when interpreting results from concurrent validation studies as no method of dietary assessment can accurately assess dietary intake in free-living settings, therefore correlated errors are almost certainly present. ^{8-10, 18}

3.5.1 Biomarkers

Biomarkers provide an independent and objective measure to assess the validity of specific nutrients as assessed by dietary assessment. ^{144, 191} At present, the DLW method to assess total energy expenditure (discussed further in section **3.5.2** below), 24-hour urinary nitrogen to assess protein intake, and 24-hour urinary potassium to assess potassium intake are routinely used to validate the results of dietary assessment methods. ¹²⁸ 24-hour urinary nitrogen is the most well-known biomarker, and is used to validate protein intake at the group level. ¹²⁹ Metabolic studies

have demonstrated a moderate correlation between protein intake, and urinary nitrogen excretion when dietary intake is controlled. ¹³⁰ However, its use relies on the assumption that participants are in nitrogen balance. In other words, nitrogen is not accumulated in the body due to the growth or repair of tissue, or lost due to starvation. Due to the daily variation in nitrogen excretion, multiple 24-hour periods of urine collection are usually conducted to validate protein intake estimated from dietary assessment. ^{130, 131} 24-hour urinary potassium is a suitable biomarker to assess dietary data as it is abundant in a wide range of foods and 24-hour urinary potassium is highly correlated with dietary intake. ¹²⁸ Other urinary biomarkers (sodium and iodine), plasma biomarkers (Vitamin C, β -carotene, Vitamin E, and Vitamin D), serum selenium, and folacin are also used. However, these biomarkers are only weak to moderate correlates of intakes and some of the plasma biomarkers can also be affected by smoking and alcohol intake, particularly for those which are prone to oxidation (e.g. Vitamin C, tocopherols, β -carotene, folate), and are therefore better used as a measure of nutrition status. ¹³²

3.5.2 Use of doubly labelled water to assess total energy expenditure

Doubly labelled water contains two (doubly labelled) isotopes ²H and ¹⁸O. The isotopes occur naturally in the environment ¹³³ but are concentrated for use in the DLW method, which allows them to be measured as they are turned over. Special attention to DLW water is given in this section as it is the biomarker used to validate EI in the PICTURE study (Chapter Seven). Total energy expenditure is usually assessed for a period of 7 to 15 days using the DLW method, though longer durations are possible. ¹³⁴ A baseline urine sample is collected before a weight-specific loading dose of DLW (²H₂O and H₂¹⁸O) is ingested orally by participants (approximately 60 to 130ml). The baseline urine sample is used to control for background levels of the isotopes ²H and ¹⁸O that occur naturally in the environment. ¹³³ After the loading dose, urine samples are collected periodically (approximately 3 to 7 samples collected) over the testing period to measure the isotopes as they are eliminated. The isotopes are measured using ratio isotope mass spectrometry. ^{8, 133}

The labelled hydrogen deuterium (²H) exists in the body as water (²H₂O), and the labelled oxygen (¹⁸O) exists in the body as water (H₂¹⁸O) and carbon dioxide (C¹⁸O₂). ¹³⁵ ¹³³ The ²H is eliminated through water losses in urine, respiration and sweat, and the ¹⁸O is eliminated through the same water losses (urine, respiration and sweat) and also as carbon dioxide in expired air (**Figure 6**).
Chapter 3. The traditional methods of dietary assessment

¹³³ Total energy expenditure can be derived as the isotopes (²H and ¹⁸O) are turned over at different rates. The rate of turnover is used in calculations to determine carbon dioxide production, which in turn is used to derive total energy expenditure. ¹³⁵ ¹³³ ¹³⁶ Total energy expenditure can be used to validate EI data at a group level as the first law of thermodynamics states that energy is conserved. Therefore the energy put into a system (EI) is equal to the energy used (energy expenditure) and/or stored by the system (change in body mass). ¹³⁷ More specific details of the methods and calculations used in this thesis are presented in the methods section of the PICTURE validation study (see section **7.5**).

The DLW method was first developed in the early 1950's ¹³⁸ but its application as a method to objectively assess total energy expenditure in free-living individuals was not demonstrated until 1982 by Schoeller and Van Santeen. ¹³⁹ The DLW method was later validated, with comparisons to direct (whole-room calorimetry) and indirect calorimetry demonstrating non-significant differences (<1%) between methods. ¹⁴⁰ It is therefore considered an accurate method to assess total energy expenditure in free-living individuals. ^{137, 141, 142} The DLW method has been used extensively as the gold standard method to evaluate EI data at a group level in nutrition research. ^{8-10, 143} However, DLW is very costly (approximately \$1000 NZD per participant excluding analysis) so is predominantly only used in studies with small of sample sizes between 5 - 50 participants; ⁹ although some larger DLW studies (N = 450+) have also been conducted. ^{26, 131}

Chapter 3. The traditional methods of dietary assessment



Figure 6 Elimination of doubly labelled water to assess total energy expenditure

3.5.3 Validity of dietary energy intake data

Studies that have evaluated EI data have demonstrated both over- and under-reporting of energy; However, only under-reporting is of major significance (only 3-4% of participants over-report). ^{8-10,} ¹⁴⁴ Under-reporting is systemic among all methods of dietary assessment and population groups, however, people of older age, ^{70, 73, 145, 146} women, ^{70, 145, 147-149} indigenous ethnicities, ^{73, 150, 151} and people with higher body weight, BMI, or waist circumference, ^{73, 146, 152-155} and people with higher energy expenditure (e.g. athletes) tend to under-report to a greater extent than others. ^{134, 156-158}

Five reviews between 1990 and 2008 have extensively compared EI data from dietary assessment methods to the DLW method, and have repeatedly demonstrated widespread underreporting across all methods of dietary assessment. ^{8-10, 137, 143} The first review was conducted by Schoeller et al ¹³⁷ in 1990, and included ten DLW studies. Large differences between studies in the under-reporting bias was found (+25% to -75%), especially in participants with obesity; they concluded that traditional methods of dietary assessment are not appropriate for obesity research. In 2001 Trabulsi et al ¹⁴³ reviewed 23 DLW studies, which revealed under-reporting reporting was greater than 20% of total EI in 16 of the 23 studies and greater than 25% of total EI in 12 of the 23 studies. Compared to the DLW, no method of dietary assessment was more accurate than others. Hill et al ¹⁰ also conducted a review in 2001 and examined 40 DLW studies of adults, obese people, and athletes. Similarly, substantial levels of under-reporting were evident for all methods of dietary assessment for most population groups, but participants with obesity and athletes typically under-reported to a greater extent. ¹⁰

Livingstone et al ⁸ reviewed 63 DLW studies in 2004, and also found a substantial under-reporting of El. A comparison between the traditional methods of dietary assessment found all methods under-estimated El to a similar extent. In 22 studies using weighed FR, 25 studies using estimated weighed food records, four using the DH, and six studies using 24-hour dietary recalls, El was under-estimated by 16% (± 10%, ± 14%, ±14%, respectively). ⁸ Additionally, in six studies using FFQs, El was underestimated by 13% (±12%). ⁸ Livingstone et al ⁸ concluded that studies in which El was reported and EE was measured using the doubly labelled water technique conclusively demonstrate widespread bias to the underestimation of El. Polsuna et al ⁹ conducted the most recent review in 2008 and compared both the prevalence of under-reporting of El (16 studies) and the magnitude of El was underestimated (11 studies) using the 24-hour recall, weighed FR, and estimated FR. No differences were found between the 24-hour dietary recall,

weighed FR, and estimated FR for the prevalence of underreporting (31.0%; range 21.5%-67.5%, 33.3%; 14.0%-38.5%, and 31.0%; 11.9%-44%, respectively), or underestimation of EI (13.4% range 12.8%-14.0%; 18.0% range 10.4%-20.2%, and 12.2% range 7.2%-20.%, respectively). ⁹

3.5.4 Implications of measurement error

In epidemiology the terms "measurement error" and "misclassification" refer to a discrepancy between the true value of a variable and its measured value. ¹⁵⁹ There are a number of implications of measurement error: (1) nonsignificant associations in studies of diet and disease relationships (masked by attenuation and discussed further below), ^{18, 34, 159, 160} ¹⁶¹ (2) under-reporting that results in serious over-estimates of nutrient inadequacies, ¹⁶² and (3) differential under-reporting, which hinders the usefulness of dietary guidelines. ¹⁶³

Gibson ^{96(p121)} states that "the existence of both random and systematic measurement errors in dietary assessment is a major challenge to the design of all types of nutritional assessment. The existence of such errors in dietary assessment can have serious consequences when interpreting data". With respect to the validity of dietary intake data presented in section **3.5.3** above, Willet ^{164(p4)} notes "we assume that all errors apply equally to all cases and non-cases in an epidemiological study, that is, errors are random in relation to disease. Systematic differences in measurement error between these two groups, that is measurement errors that are biased with respect to disease, have serious consequences that are usually not amenable to correction".

Moreover, measurement error with respect to EI requires special consideration, as an inaccurate assessment of EI can have important implications including ^{8, 165} (1) the level of EI may be the primary determinant of disease, (2) most nutrients are positively correlated with EI thus differences in individual EI can cause variation in the intake of specific nutrients unrelated to dietary composition, and (3) EI may be associated with a disease but not its direct cause, thus the effects of specific nutrients may be distorted or confounded by EI.¹⁶⁵

An example that demonstrates the implications of measurement error has been presented by Kipnis et al. ¹⁸ Data from a U.K. FFQ validation study were used to model the attenuation effect of measurement error; the analysis revealed that measurement error in the FFQ could lead to a 51% attenuation of the true nutrient effect for protein, and require a 2.3 times larger sample size than previously calculated i.e. the measurement error would substantially increase the chance of null findings. Kipnis et al ³⁴ further demonstrated the attenuating effect of measurement error in the

Observing Protein and Energy Nutrition Study (OPEN), a large biomarker validation study (n=484) in part designed to explore the implications of measurement error. The analysis revealed that the FFQ led to severe attenuation (reduced statistical power), and that the use of a 24-hour recall for a reference method for an FFQ could under-estimate attenuation by up to 60%.

Due to the implications of measurement errors, statistical approaches in handling the data and adjusting EI have been developed and intensively studied. ^{161, 166-176} However, the relative merits of the different method of energy adjustment have been vigorously debated, as assumptions regarding the nature of the data must be made, and adjustments cannot eliminate selective/systematic underreporting of foods. ^{177 164 178-183}

3.6. Summary

There are many potential sources of error in dietary assessment, from the initial recording of intake to the conversion of food intake data into nutrient data. Traditional methods of dietary assessment are used widely in research but are self-reported without any objective means to verify intake. Findings of DLW studies have consistently demonstrated systematic under-reporting of EI, which is systemic among all methods of dietary assessment and population groups (though certain groups misreport to a greater extent). Such measurement error increases the chance of null findings in nutrition research towards the null hypothesis thereby masking relationships between nutrients and disease. Despite the development of sophisticated software and structured methodology to ensure consistent and thorough assessment of diet, the primary methodologies to assess dietary intake have remained the same. Livingstone and Black note that if dietary assessment is to be improved, we need to understand which foods and meals are misreported. ⁸ Therefore, research needs to address the problems of self-reporting bias, and not just improve the procedures of data collection; otherwise, the issues associated with dietary assessment will remain.

CHAPTER 4. UNDER-REPORTING REMAINS Α LIMITATION SELF-REPORTED OF KEY ANALYSIS OF DIETARY INTAKE: AN THF 2008/09 ZEALAND ADULT NUTRITION NEW **SURVEY**

4.1. Introduction to publication

The findings of the Adult nutrition Survey (ANS 08/9) published in November 2011 indicated the level of under-reporting of dietary EI may have increased substantially compared to the 1997 National Nutrition Survey (NNS97). Measured body weights increased by 4-5kg for men and women, but self-reported Els were lower for men (10.7MJ, 95%Cl 10.4-11.1MJ vs. 12.0MJ, 95%CI 11.7-12.2MJ) and women (7.6MJ, 95%CI 7.5-7.8MJ vs. 8.0MJ, 95%CI 7.8-8.1MJ) compared to the NNS97. ⁶⁵ Determining the prevalence of under-reporting for dietary EI in the general population would reveal the importance of improving dietary assessment methods. This chapter (Chapter Four) is a reformatted version of the manuscript entitled "Under-reporting remains a key limitation of self-reported dietary intake: an analysis of the 2008/09 New Zealand Adult Nutrition Survey" in the European Journal of Clinical Nutrition, 2013, Volume 68, Issue 2, Pages 259-264. DOI: 10.1038/ejcn.2013.242. The most recent impact factor available was 2.756 (2013). A few minor revisions were made to this reformatted manuscript at the request of the thesis examiners. The chapter presents results from a retrospective statistical analysis undertaken on the datasets of the 2008/09 New Zealand Adult Nutrition Survey. The study was conducted to address Objective One of this thesis, which was to evaluate self-reported dietary EI in the 2008/9 Adult Nutrition Survey and determine the prevalence of low energy reporters.

4.2. Author contribution

Luke Gemming was involved in developing the research question, application for access to ANS 08/9 datasets, development of methods, and procedures completed to analyse the data. He also wrote the paper for publication.

UNDER-REPORTING REMAINS A KEY LIMITATION OF SELF-REPORTED DIETARY INTAKE: AN ANALYSIS OF THE 2008/09 NEW ZEALAND ADULT NUTRITION SURVEY

^aLuke Gemming MSc

^aYannan Jiang PhD

^bBoyd Swinburn MD FRACP

^bJennifer Utter PhD

^aCliona Ni Mhurchu PhD

^aPostal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

^bSchool of Population Health, University of Auckland, Private Bag 92019, Auckland, 1142, New Zealand

Address for correspondence:

Luke Gemming

Postal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

l.gemming@auckland.ac.nz

4.3. Abstract

BACKGROUND/OBJECTIVES: The most recent New Zealand Adult Nutrition Survey 2008/09 (ANS 08/9) revealed a decrease in reported EIs compared to the previous 1997 National Nutrition Survey (NNS97). Conversely, measured body weights increased over the same period. We conducted an analysis on the ANS 08/9 datasets to evaluate reported EIs.

SUBJECTS/METHODS: Analysis was conducted on data from 3919 (1715 men and 2204 women aged \geq 15yrs) survey participants who completed the 24-hour dietary recall in the ANS 08/9. Under-reporting was assessed using the ratio of reported EI to estimated resting metabolic rate (EI:RMR_{est}), and a cut-off limit of <0.9 (EI:RMR_{est}) was used to identify low energy reporters (LERs). Results were examined by gender, body size, age and ethnicity.

RESULTS: The mean EI:RMR_{est} (SEM) was 1.34 (0.02) for men, and 1.23 (0.02) women. Overall, 21% of men and 25% of women were classified as LERs. There was a greater prevalence of LERs among people with overweight (25%), or obesity (30%) than people with normal body weight (16%, p<0.001). The oldest age group (\geq 65yrs) had a greater prevalence of LERs (33%) compared to all other age groups (19-24%, p<0.001). Pacific people had a greater prevalence of LERs (33%) compared to Maori (26%, p=0.007) and European (23%, p<0.001). Compared with the NNS97, a substantial increase in the prevalence of LERs was evident in most subgroups.

CONCLUSIONS: Under-reporting of EI will continue to be a major limitation of nutrition surveys without technological innovation. Care should be taken when interpreting EI data.

KEYWORDS Nutrition Assessment, Low Energy Reporters, Under-Reporting, Dietary Surveys, 24-hour Dietary Recall

4.4. Introduction

Under-reporting of total energy intake (EI) is a common and acknowledged source of measurement error in dietary assessment. ^{8, 9} Many factors contribute to under-reporting, but respondent biases (e.g. social desirability) and memory lapses are probable sources not easily addressed.

The gold standard method to assess the validity of EI data post hoc is with the doubly labelled water (DLW) technique to accurately measure total energy expenditure. ^{8, 135, 140} However, due to the prohibitive expense of DLW, under-reporting is most commonly assessed in large samples by estimating basal metabolic rate (BMR) and applying Goldberg cut-off values. ^{9, 184} The Goldberg cut-offs were derived using evidence from whole-body calorimetry and DLW studies, and take into account typical physical activity levels (PAL) to assign minimum EI to BMR values (EI:BMR). ¹⁸⁴, The Cut-off values assess if the EI data recorded is a plausible measure of EI during the measurement period and can be used to estimate the prevalence of low energy reporters (LERs).

In 2011, findings of The New Zealand Adult Nutrition Survey 2008-09 (ANS 08/9), which included a nationally representative sample (n=4258) of adults aged 15yrs and over, were released. The survey used a multiple pass 24-hour dietary recall and, compared with the previous 1997 National Nutrition Survey (NNS97), reported a significant decrease in self-reported EI for men 10.7MJ (95%CI 10.4-11.1MJ) vs. 12.0MJ (95%CI 11.7-12.2MJ), and a non-significant decrease for women 7.6MJ (95%CI 7.5-7.8MJ) vs. 8.0MJ (95%CI 7.8-8.1MJ), respectively. ⁶⁵ Conversely, a significant rise was recorded in measured mean body weights of approximately 4-5kg for men and women, of all ethnicities, compared with the NNS97.

Controlled feeding, metabolic, and modelling studies have demonstrated a strong relationship between body weight and EI. ¹⁸⁶⁻¹⁸⁹ Therefore, to investigate the paradox of decreased EI in parallel with increased body weight we undertook analysis of the ANS 08/9 data using the Goldberg cut-offs, and estimated the prevalence of LERs by gender, body size (normal, overweight, obese), age, and ethnicity. These findings were compared to a previous analysis of the NNS97 dataset, to observe time trends in under-reporting. ⁷³

4.5. Methods

The ANS 08/9 was conducted by the University of Otago, using a computer-based interviewerassisted three pass (multiple-pass) 24-hour dietary recall method. ^{23, 65} The study was completed between October 2008 to October 2009, and a total of 4721 participants (2066 men and 2655 women) aged 15yrs and over completed the survey. ^{23, 65} To permit equivalent comparisons between the ANS 08/9 and the NNS97 we followed a similar methodology to Pikholz et al ⁷³ to evaluate reported EIs and estimate the prevalence of LERs (surveys compared in the discussion).

The ANS 08/9 datasets which consisted of participant characteristics, anthropometry, and 24-hour recall nutrient data were first merged. Participants were excluded if data from key variables, such as height or weight were missing. Chinese, Indians and the "Other" (ethnic) group (Dutch, Japanese, Tokelauan) were also excluded due to small participant numbers and mixed ethnicity. After these exclusions a total of 3919 participants (1715 men and 2204 women) remained for the analysis.

A body size variable was created by grouping BMI into three categories: normal weight, overweight, and obese. BMI ranges used were those recommended by the World Health organization (WHO) for all adults over 18 years of age as follows: normal weight = BMI<25 kg/m2, overweight = BMI \geq 25 kg/m2 to BMI <30 kg/m2, obese = BMI>30kg/m2.¹⁹⁰ The International Obesity Taskforce (IOTF) cut-offs for children and adolescents were applied for all participants aged 15-18yrs.^{191, 192} A second body size variable was created to allow comparison with the NNS97 analysis of under-reporting which used the following ethnic-specific cut-offs for (1) NZ Europeans: normal weight = BMI<25 kg/m2, overweight = BMI \geq 25 kg/m2 to BMI<30 kg/m2, obese = BMI \geq 30kg/m2, and (2) for Maori and Pacific people: normal weight = BMI<26 kg/m2, overweight = BMI \geq 26kg/m2 to BMI <32 kg/m2, obese = BMI \geq 32 kg/m2.¹⁹³ Prior to 2006 ethnic specific cut-offs were commonly used in New Zealand as dual x-ray absorptiometry (DEXA) revealed Polynesians (Maori and Samoans) had a significantly greater ratio of lean muscle mass : fat mass compared to New Zealand Europeans.¹⁹⁴ Post 2006 New Zealand aligned national BMI classifications with WHO recommendations to allow comparison with other nations.¹⁹⁵ Ethnicity was self-identified.²³

Several steps were performed to estimate resting metabolic rate (RMR_{est}). Using equations derived by Swinburn et al ¹⁹⁴ fat mass (FM, in kg) was calculated from BMI for N.Z. European,

Maori, and Pacific males and females (Samoan equations were used for the whole Pacific ethnic group). Fat free mass (FFM, in kg) was then calculated by subtracting fat mass from weight. Lastly, an equation taken from Bogardus et al ¹⁸⁶ was used to calculate RMR_{est} (RMR_{est} (kilocalories per day) = (22.8 x FFM) + 489). The unit for energy (kilocalories per day) was converted into kilojoules per day by multiplying by the conversion factor 4.184.

The ratio between reported EI and RMR_{est} (EI:RMR_{est}) was calculated by dividing EI by RMR_{est}. Cut-off values taken from work conducted by Goldberg et al ¹⁸⁴ were used to evaluate EI data. Goldberg et al ¹⁸⁴ used estimated basal metabolic rate (BMR_{est}) predicted from the Schofield equation, ¹⁹⁶ where this analysis substituted RMR_{est} for BMR_{est} as they are nearly identical. EI:BMR_{est} cut-off values vary according to the sample size and number of days dietary intake is measured. This analysis used the 95th percentile lower cut-off value for one-day of dietary intake (as data were from a single 24-hour dietary recall) to classify LERs for individuals and population subgroups (gender, body size, age, and ethnicity). The cut-off values range from 0.9 for one person, to 1.53 for 2000 people (based on one day of dietary intake). ¹⁸⁴ Thus the cut-off limit of <0.9 EI:RMR_{est} was used to classify individuals as LERs and higher cut-off values (1.50-1.53 dependant on *n*) were used to assess the mean EI:RMR_{est} for subgroups. Individuals with a EI:RMR_{est} ≥0.9 were considered adequate reporters for the purpose of this analysis, but likely comprise a mixture of possible LERs, adequate energy reporters (AER), and high energy reporters (HER).¹⁴⁴

The ANS 08/9 used a multi-stage, stratified, probability-proportional-to-size sample design, and over-sampled Maori, Pacific people, and some age groups with a three-step selection process by meshblocks, dwellings from within each meshblock, and respondents within households. The weights were calculated for every survey participant to ensure no group was under - or over - represented in estimates from the survey. 100 replicate weights were produced for every respondent in the sample. ²³ The standard error of the population estimate is based on the variation of the replicate estimates. Weighted means and standard errors of the mean (SEM) EI:RMR_{est} were calculated using linear regression models with both unadjusted and adjusted analyses, controlling for the effect of important confounders (body size, age, ethnicity). Same survey weighted estimates were calculated for the percentages of LERs, taking into account the unequal selection probabilities. Differences between subgroups were tested using the standard t-test. Since this was a national survey to generate population level estimates, no multiple

comparisons were considered. Statistical package SAS Version 9.3 (SAS Institute Inc, Cary, NC) and R version 2.15 (R Foundations for Statistical Computing, Auckland, New Zealand) were used. Threshold for significance was set at $\alpha < 0.01$.

4.6. Results

Table 4 presents the population estimates of important participant characteristics by subgroups, using the data collected from all 3919 participants. The clear differences in age, body weights and BMI between ethnicities have been previously reported. ⁶⁵ Though, noteworthy ethnic differences include, Europeans 8 - 10yrs older on average than Maori and Pacific people, and more than double the prevalence of obesity among Pacific people compared to Europeans.

The mean EI:RMR_{est} (SEM) values are presented in **Table 5** for the total population and subgroups. The mean EI:RMR_{est} for all participants of 1.28 (0.01) were below the suggested cutoff value of 1.53 (measurement of one-day;1 x 24-hour dietary recall) and all subgroups were below the suggested cut-off values which ranged from 1.50 to 1.53 (see **Table 4** legend and Goldberg el al¹⁸⁴ for suggested cut-off values).

Regarding body size, no differences were observed among men but women with overweight, and obesity had a lower mean EI:RMR_{est} of 1.18 (0.03), and 1.13 (0.03) compared to women with normal weight 1.28 (0.03) respectively (p=0.003 and p<0.001, adjusted for age, and ethnicity). Regarding age, the mean EI:RMR_{est} for men 1.08 (0.04) and women 1.05 (0.03) aged \geq 65yrs were lower than all other age groups, 1.26 (0.05) to 1.41 (0.05), and 1.19 (0.03) to 1.31 (0.04) respectively (p \leq 0.002 and P \leq 0.001, adjusted for ethnicity and body size). With respect to ethnicity there were no significant differences in the group mean EI:RMR_{est} among men, or among women.

Characteristics		Gender	European	Maori	Pacific	All
Number		Male	1013	388	314	1715
		Female	1273	588	343	2204
Age (yrs), mean (SEN	1)	Male	46.0 (0.37)	36.7 (0.27)	36.5 (0.38)	44.3 (0.28)
		Female	47.1 (0.31)	37.8 (0.29)	37.9 (0.39)	45.3 (0.25)
Height (cm), mean (S	EM)	Male	176.1	175.0	176.2	176.0
			(0.29)	(0.37)	(0.50)	(0.23)
		Female	162.5	163.4	163.7	162.7
			(0.25)	(0.33)	(0.37)	(0.21)
Weight (kg), mean (SEM)		Male	84.6 (0.68)	91.6 (1.26)	98.0 (1.50)	86.3 (0.57)
		Female	71.7 (0.59)	81.8 (1.04)	88.3 (1.29)	74.0 (0.52)
BMI (kg/m ^² , mean (SEM)		Male	27.3 (0.21)	29.9 (0.37)	31.5 (0.45)	27.9 (0.18)
		Female	27.1 (0.23)	30.7 (0.40)	33.0 (0.45)	28.0 (0.20)
Body Size WHO/IOTF [§] n						
(70) N	lormal	Male	356 (35.1)	74 (19.1)	[†] 27 (8.6)	457 (26.6)
		Female	525 (41.2)	130 (22.1)	[†] 43 (12.5)	698 (31.7)
Overv	weight	Male	421 (41.6)	153 (39.4)	89 (28.3)	663 (38.7)
		Female	440 (34.6)	170 (28.9)	89 (25.9)	699 (31.7)
	Obese	Male	236 (23.3)	161 (41.5)	198 (63.1)	595 (34.7)
		Female	308 (24.0)	288 (49.0)	211 (61.5)	807 (36.6)
Body Size [§] n (%) **						
Swinburn No	ormal	Male	369 (36.4)	104 (26.8)	[†] 43 (13.7)	516 (30.1)
		Female	530 (41.6)	167 (28.4)	58 (16.9)	755 (34.3)
Overv	weight	Male	411 (40.6)	163 (42.0)	102 (32.5)	676 (39.4)
		Female	437 (34.3)	201 (34.2)	112 (32.7)	750 (34.0)
	Obese	Male	233 (23.0)	121 (31.2)	169 (53.8)	523 (30.5)
		Female	306 (24.0)	220 (37.4)	173 (50.4)	699 (31.7)

Table 4 Baseline characteristics of participants*

BMI = Body mass index, SEM = standard error of the mean.

*The unequal selection probabilities have been taken into account.

**Percentages relate to body size in each ethnic and gender subgroup (i.e. male: % normal + % overweight + % obese = 100%).

+Limited sample size within that cell, n<50, and data should be interpreted with caution.

§ See methods for definitions.



Characteristics		Male	Female	All
All: mean(SEM)		1.34 (0.02)	1.23 (0.02)	1.28 (0.01)
Ethnicity: †	European:	1.31 (0.02)	1.22 (0.02)	1.27 (0.02)
mean (SEM)	Maori:	1.36 (0.04)	1.17 (0.03)	1.26 (0.03)
	Pacific:	1.21 (0.06)	1.19 (0.04)	1.20 (0.03)
Age (yrs):‡	15-29:	1.41 ^a (0.05)	1.31 ^a (0.04)	1.35 ^a (0.03)
mean (SEM)	30-39:	1.40 ^a (0.05)	1.21 ^ª (0.04)	1.29 ^{a,b} (0.03)
	40-49:	1.38 ^a (0.05)	1.21 ^ª (0.04)	1.29 ^{a,b} (0.03)
	50-56:	1.26 ^a (0.05)	1.19 ^a (0.03)	1.22 ^b (0.03)
	≥65:	1.08 ^b (0.04)	1.05 ^b (0.03)	1.06 ^c (0.02)
Body Size: IOTF §	normal:	1.34 (0.05)	1.28 ^a (0.03)	1.32 ^a (0.03)
mean (SEM)	overweight:	1.30 (0.04)	1.18 ^b (0.03)	1.24 ^{a,b} (0.02)
	obese:	1.23 (0.03)	1.13 ^b (0.03)	1.18 ^b (0.02)
Body Size:	normal:	1 22 (0.05)	1.26 ⁸ (0.02)	$1.20^{2}(0.02)$
Swinburn§		1.32 (0.03)	1.26 (0.03)	1.30 (0.03)
mean (SEM)	overweight:	1.30 (0.03)	1.17 ^a (0.02)	1.23 ^{a,b} (0.02)
	obese:	1.23 (0.04)	1.13 ^b (0.03)	1.18 ^b (0.02)

Table 5 Ratio of reported energy intake to estimate resting metabolic rate (EI:RMR_{est})*

*The unequal selection probabilities have been taken into account.

+Adjusted for age and body size (and gender for the last column 'all').

‡Adjusted for ethnicity and body size (and gender for the last column 'all').

§ Adjusted for ethnicity and age (and gender for the last column 'all'). See methods for definitions.

a, b, c, and d = values for each population characteristic in the same column with different superscript letters are significantly different from each other, P<0.01.

Suggested cut-off values taken from Goldberg et al¹⁸⁴ for *n* and one day of dietary intake with 95% confidence limits (Male 1.52; Female 1.52; European 1.52; Maori men 1.51; Maori women 1.52; Pacific people 1.50).

Using the cut-off limit <0.9 applied to individual EI:RMR_{est} values the overall prevalence of LERs was 23.0%; with men 21.0%, and women 25.0%. The proportions of LERs by subgroups (ethnicity, age and body size) are presented in **Figure 7** to **Figure 9**. **Figure 7** presents the prevalence of LERs using the WHO IOTF classifications for body size (see methods). The prevalence of LERs among men with obesity was 12.7% greater than men with normal body weight (p<0.001), and the prevalence of LERs among women with overweight, and obesity was 9.4% and 14.6% greater than women with normal body weight, respectively (both p<0.001). **Figure 8** shows the prevalence of LERs among men aged ≥65yrs was 17.5% and 20.6% greater than among men aged 15-29yrs and 30-39yrs, respectively (both p<0.001). Furthermore the prevalence of LERs among women aged ≥65yrs was ≥11.8% greater than all other age groups (p≤0.007). **Figure 9** shows the prevalence of LERs among Pacific men was 11.7% (p=0.001) and 14.0% (p=0.007) greater than among Maori and European men, respectively. Moreover, the prevalence of LERs among Pacific and Maori women was 10.7% and 8.2% greater than among European women (both p=0.003).



Figure 7 Percentage of low energy reporters by body size using the WHO & Obesity International Taskforce BMI cut-offs ¹⁹¹ (unadjusted; unequal selection probabilities taken into account). a, b = values for each population characteristic in the same column with different superscript letters are significantly different from each other, P<0.01.



Figure 8 Percentage of low energy reporters by age group (unadjusted; unequal selection probabilities taken into account). a, b, c, = values for each population characteristic in the same column with different superscript letters are significantly different from each other, P<0.01.



Figure 9 Percentage of low energy reporters by ethnicity (unadjusted; unequal selection probabilities taken into account). a, b = values for each population characteristic in the same column with different superscript letters are significantly different from each other, P<0.01.





Figure 10 Percentage change of low energy reporters in the NZ Adult Nutrition Survey 2008-9 (ANS 08/9) by body size, age, and ethnicity compared to the 1997 NZ National Nutrition Survey (NNS97) (unadjusted; unequal selection probabilities taken into account).

NNS97 data taken from Pikholz et al (2004). Swinburn et al (1998) classifications for body size used.

4.7. Discussion

This analysis of the 2008/9 New Zealand Adult Nutrition Survey revealed substantial underreporting of EI. The mean EI:RMR_{est} values for the population were substantially below suggested cut-off values and lower than those reported in the NNS97 for both men and women, 1.34 (0.02) vs. 1.51 (0.02), and 1.28 (0.01) vs. 1.40 (0.01) respectively. ⁷³ The greater prevalence of LERs observed among priority ethnic groups, older age groups, and people with overweight and obesity have been reported elsewhere. ^{70, 145, 147, 197}

Compared with the NNS97 analysis the proportion of LERs increased in nearly all subgroups (differences between surveys presented in **Figure 10**). The overall increase in the prevalence of LERs paralleled an increase in mean population body weight for men and women, for all ethnicities. The increase in mean body weight (~4-5kg) can be calculated to reflect an approximate increase of 400-500kJ in daily EI. ¹⁹⁸ This is in marked contrast to the reported decrease in daily EI for men (1300KJ) and women (400KJ) between 1997 and 2008/9.

Notably the proportion of LERs more than doubled among men classified as normal body weight (6.1 to 14.7%), and increased in women with normal body weight from 14.4% to 18.6%. Considering the similarities between surveys this finding suggests the increase in LERs may be due to an increased influence of psychosocial factors (e.g. social desirability) and other behavioural characteristics (e.g. body dissatisfaction) among individuals with normal body weight (in addition to people with overweight and obesity). The relationship between these factors and under-reporting is well established, thus sociocultural changes and events within New Zealand society between surveys offer possible explanations for the differences observed.²⁸

Over time, there has been an increase in screen-based activities and a substantial shift in the genre and quantity of television shows and media advertising, many of which portray a slim-body image or health-related content (e.g. reality television focussed on weight loss, cosmetic surgery, makeover, modelling etc.) which are known to influence factors, such as body dissatisfaction, self-esteem, depression, and eating behaviours (all related to under-reporting). ^{28, 199, 200} There were also two widely publicised nation-wide government led health campaigns (Healthy Eating Healthy Action, 2003; Mission On, 2006) that promoted a healthy lifestyle which, in conjunction with an increased public awareness of nutrition-related health and increased dieting practises may have

influenced people's self-perceptions, and thus increase the likelihood of dietary under-reporting.

The absence of a method to verify the self-report or assist memory in dietary recalls (during the assessment) dictates that pervasive under-reporting is likely to remain a key issue in large scale dietary surveys. ^{69-71, 73} However, the recent development of wearable cameras may provide new solutions to enhance self-report and improve accuracy. Two studies recently demonstrated a significant increase self-reported EI when wearable cameras were used data to complement food records, ²⁰⁴ and 24-hour dietary recalls. {Gemming, #983} The images from the wearable camera revealed unreported/records foods and misreporting/reporting errors that substantially added to the estimated energy intake from the traditional method. Moreover, the design of bespoke wearable cameras for the passive and objective assessment of dietary intake and physical activity are under development.²⁰⁵ If feasible, such technologies would be a welcome addition to improve dietary assessment,¹⁵ though, further testing and validation would be required to evaluate their utility for large scale dietary surveys.

These analyses have several limitations that need consideration. The present study used the same methods used as those utilised by Pikholz et al ⁷³ to allow comparison between New Zealand's Adult Nutrition Surveys. However, possible selection bias and differences between surveys sample designs must be taken into account. ^{23, 94} Participants in the NNS97 survey were recruited after participating in the linked New Zealand Health Survey with a response rate of 50%, where participants in the ANS 08/9 survey were recruited independently, had a greater percentage of Maori and Pacific people sampled, and a somewhat higher response rate (61%). Other limitations comprise both limitations of the ANS 08/9 survey methodology, and methods used in the present study to analyse the data. The ANS 08/9 used a single 24-hour dietary recall as primary method to collect nutrient data for the full sample. A single 24-hour recall cannot capture daily, weekly or seasonal intra-individual variation in food intake which must be considered when interpreting data. Regarding the methods used in this analysis, estimated resting metabolic rate was derived from estimated fat-free mass and the Goldberg cut-offs were applied, where the gold-standard for the assessment EI data is with the use of the doubly labelled water technique. ^{8, 135}

The use of RMR_{est} was justified in this analysis as the Schofield equation commonly used to estimate BMR was developed for a population of normal weight (up to 84kg). Thus, the Schofield equation could not be assumed valid for the current New Zealand population (due to a high proportion with a BMI \geq 35kg/m²). ²⁰⁶ Furthermore about 80% of the variance between individual BMR can be explained by fat-free mass, ^{187, 207, 208} thus New Zealand specific equations for estimation of fat-free mass were used. ¹⁹⁴

Limitations of the Goldberg cut-offs must also be considered. Cut-off values alone cannot distinguish dieting from LERs (if below the cut-offs), and the cut-offs only identify extreme degrees of low energy reporting. ¹⁸⁵ Moreover, use of a single cut-off value can be conservative for physically active populations, as people can under-report but not fall below the cut-offs. ^{144, 185} Information regarding PAL was not recorded in the ANS 08/9, but over half of New Zealand Adults meet physical activity guidelines (self-reported) suggesting the cut-off values used were likely too conservative. ⁵⁸

4.8. Conclusions

This analysis highlights a systematic bias in self-reported EI data and, the need to interpret EI data with substantial caution. Without technological innovation, under-reporting of EI will continue to be a major limitation of 24-hour dietary recall method used for large scale nutrition surveys.

Acknowledgements

Access to the data in this study was provided by Statistics New Zealand in accordance with security and confidentiality provisions of the Statistics Act 1994. The results in this paper have been confidentialised to protect individuals from identification.

Conflict of interest

There were no conflicts of interest.

CHAPTER 5. IMAGE-ASSISTED DIETARY ASSESSMENT: A SYSTEMATIC REVIEW OF THE EVIDENCE

5.1. Introduction to publication

There are concerted efforts to use technologies to enhance dietary assessment. ^{16, 31} One approach is to use images of eating episodes to assist self-report used in traditional methods, or use the images as the primary record of dietary intake. Several of these image-assisted methods have been developed, pilot tested or validated. ²⁰⁹⁻²¹⁴ However, no review had examined studies that have evaluated or validated image-assisted methods of dietary assessment among users. This chapter (Chapter Five) is a reformatted version of the manuscript entitled "Image-assisted dietary assessment: A systematic review of the evidence" was published in the Journal of the Academy of Nutrition and Dietetics, 2015; Volume 115, issue 1, pages 64-77 DOI: 10.1016/j.jand.2014.09.015. The latest impact factor available was 3.797 (2013). A few minor revisions were made to this reformatted manuscript at the request of the thesis examiners. The chapter presents the results from the first systematic review to examine the evidence for image-assisted methods of dietary assessment. The review was completed to address Objective Two of this thesis, which was to undertake a systematic review of existing evidence for image-assisted methods of dietary assessment.

5.2. Author contribution

Luke Gemming was involved in developing the research question, search strategy, selecting the papers, and extracting the data. He also summarised the information and wrote the paper for publication.



IMAGE-ASSISTED DIETARY ASSESSMENT: A SYSTEMATIC REVIEW OF THE EVIDENCE

^aLuke Gemming MSc

^bJennifer Utter PhD

^aCliona Ni Mhurchu PhD

^aPostal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

^bSchool of Population Health, University of Auckland, Private Bag 92019, Auckland, 1142, New Zealand

Address for correspondence:

Luke Gemming

Postal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

l.gemming@auckland.ac.nz

5.3. Abstract

Images captured during eating episodes provide objective information to assist in the assessment of dietary intake. Images are captured using handheld devices or wearable cameras, and can support traditional self-report or provide the primary record of dietary intake. A diverse range of image-assisted methods have been developed and evaluated but have not been previously examined together. Therefore a review was undertaken to examine all studies that have evaluated or validated image-assisted methods of dietary assessment for assessing dietary EI. Identified image-assisted methods that employ similar methodologies were grouped for comparison. English language full-text research articles published between November 1998 and November 2013 were searched using five electronic databases. A search of reference lists and associated websites was also conducted. Thirteen studies, that evaluated ten unique imageassisted methods among adults aged 18 to 70 years, were included. Ten studies used handheld devices and three studies used wearable cameras. Eight studies evaluated image-based food records, two studies explored the use of images to enhance written food records, and three studies evaluated image-assisted 24-hour dietary recalls. Results indicate images enhance selfreport by revealing unreported foods and identify misreporting errors not captured by traditional methods alone. Moreover, when used as the primary record of dietary intake, images can provide valid estimates of energy intake. However, image-assisted methods that rely on image analysis can be prone to underestimation if users do not capture images of satisfactory quality before all foods are consumed. Further validation studies using criterion measures are warranted. The validity among children, adolescents, and the elderly as well as the feasibility of using imageassisted methods in large samples needs to be examined. Additional research is also needed to better understand the potential applications and pitfalls of wearable cameras.

5.4. Introduction

There is a clear need for improved methods of dietary assessment. ^{15, 215} Despite advances in computer technologies to standardize methods and streamline analysis through software, traditional methods are still prone to substantial error and bias. ^{26, 72, 73} A main contributor to bias is reliance on self-report without the ability to verify the dietary information, which typically leads to under-reporting of energy intake (EI), especially problematic in overweight and obese populations. ^{26, 131, 152-154, 216-219} However, despite the bias, self-report obtains valuable information about the foods people consume.

Due to the limitations of self-report and the perception that images may increase objectivity (a picture is worth a thousand words), various methods of image-assisted dietary assessment have been developed, pilot tested or validated. ²⁰⁹⁻²¹⁴ Image-assisted dietary assessment refers to any method that uses images/video of eating episodes to enhance self-report of traditional methods, or uses images/video as the primary record of dietary intake. The images of foods can be captured using any device but two distinct approaches for capturing the image, 'active' and 'passive', have been explored.

Active methods typically require individuals to capture images of foods with handheld devices, such as digital cameras or smartphones. Generally images are captured before and after eating episodes (to record wastage) and a reference marker is placed near the foods to assist image analysis techniques (manual or automated). ²²⁰⁻²²² Often the images of foods are supported by supplementary text or voice recordings describing the foods, or require user input to confirm details extracted from the image (within a software application), such as food type or portion size. ^{220, 223, 224} The active approach helps ensure the images obtained are relatively consistent for image-analysis, but relies on users to remember to use the camera at every eating episode.

The passive approach uses wearable cameras to automatically capture point-of-view images of daily events, including eating episodes, with virtually no user input. Thus passive image capture does not rely on users to capture images of foods; however, the images captured are not directed specifically at foods, nor do they contain a reference marker to assist analysis. A novel aspect of passive image capture, in comparison to active methods, is the ability to aid memory recall during retrospective assessment without the need for the user to manually record dietary intake during the assessment period. ^{32, 213}

Due to a variety of technologies suitable for use in image-assisted methods, and the differences between the active and passive approach, there is a diverse range of methods not previously examined or easily compared. Stumbo et al ³¹ have detailed the methods employed in selected image-assisted methods in development (yet to be validated) and Illner et al ²²⁵ have examined the strength and weaknesses of several innovative technologies in dietary assessment. However, to date no review has examined the current evidence regarding the use of image-assisted dietary assessment methods. The aim of this review was to examine all studies that have evaluated or validated an image-assisted method of dietary assessment compared to a reference method for assessing dietary EI. Due to the diversity of image-assisted methods identified, we grouped and categorized methods that employ similar methodologies for comparison.

5.5. Methods

5.5.1 Eligibility criteria

All studies that evaluated or validated an image-assisted method of dietary assessment compared to a reference method for assessing dietary EI were included. Technical reports associated with the studies and methods of image-assisted dietary assessment included were only used to support the description of the method and supporting systems used.

5.5.2 Exclusion criteria

Studies that did not report EI or compare EI to a reference method were excluded. Methods of image-assisted dietary assessment under development that have not been evaluated among users described in technical reports were excluded. Studies that used pre-captured images or image databanks to assist portion size estimation in traditional methods of dietary assessment were also excluded.

5.5.3 Information sources and search strategy

Five electronic databases were searched: MEDLINE; PubMed; Web of Science; CINAHL Plus; and ProQuest. The searches were conducted in November 2013. A search strategy was developed using a combination of Medical Subject Headings and keywords. The search string was modified where appropriate for use in the other databases. Search limiters included English language, human participants, and studies reported between 1998 and the search date to ensure all technologies evaluated in image-based methods were identified. (See **Figure 11** for example search strategy for MEDLINE database). A manual search of included articles reference sections, and associated websites, supplemented the search of electronic databases. Corresponding authors of identified image-assisted methods in development (not evaluated or validated among users) were contacted to identify any additional studies. The search results from all databases and the manual search were imported into a reference software package EndNote (version 16, released August 6, 2012, Thomson Reuters, Philadelphia, Pennsylvania). After the removal of duplicates, the title and abstracts were screened by one reviewer [LG]. The full-text studies that appeared relevant were then obtained and screened. Manuscripts potentially eligible for inclusion were discussed and their inclusion or exclusion were agreed upon by two authors [LG and CN].

Example Search Strategy				
Source: MEDLINE ^a				
1. exp ^b Technology/ ^c				
2. exp Cellular Phone/				
3. smartphone\$ ^d .mp. ^e				
4. mobile phone\$.mp.				
5. mobile telephone\$.mp.				
6. personal digital assistant.mp.				
7. PDA.mp.				
8. exp Computers, Handheld/				
9. tablet computer.mp.				
10. device.mp.				
11. life-logging.mp.				
12. exp Video Recording/				
13. video.mp.				
14. image\$.mp.				
15. digital camera.mp.				
16. wearable camera.mp.				
17. sensecam.mp.				
18. wearable sensor.mp.				
19. camera.mp.				
20. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8 or 9 or 10 or 11				
or 12 or 13 or 14 or 15 or 16 or 17 or 18 or 19				
21. exp Nutrition Assessment/				
22. exp Dietetics/mt [Methods]				
23. dietary assessment.mp.				
24. exp Diet/is, mt [Instrumentation, Methods]				
25. exp Diet Surveys/				
26. exp Nutrition Surveys/				
27. 21 or 22 or 23 or 24 or 25 or 26				
28. 20 and 27				
29. limit 28 to (english language and humans				
and yr="1998 -Current")				

Figure 11 MEDLINE search strategy for the systematic review examining the evidence for image-assisted methods of dietary assessment.

5.5.4 Data extraction

Data extraction was conducted by one reviewer [LG] using a custom data extraction form to extract general study details as follows: participant characteristics (sex, age, BMI); inclusion/exclusion criteria; study setting; method of image-assisted dietary assessment; study design and duration; reference method used; mean El/energy expenditure; statistical analysis; feedback regarding method/technology, and study limitations. When further information was sought, corresponding authors were contacted via email. One follow-up email was sent if no response was received to the first. Because of the substantial heterogeneity between image-assisted methods of dietary assessment used, study designs, durations, and populations only a narrative review was performed in this systematic review.

5.5.5 Quality assessment

The majority of studies published in this field to date have been pilot or feasibility studies. Therefore a formal assessment of study quality was not undertaken as the assessment would not provide meaningful outcomes. However, the study designs were examined to assess risk of bias regarding the review and analysis of captured images to obtain EI data. (There is potential for bias if the image analysis is conducted without independent image analysts/researchers blinded from the reference method).

5.6. Results

5.6.1 Search results

Please refer to **Figure 12** for a flow diagram depicting the search and inclusion of the studies. Of the 1278 potentially eligible articles initially identified, 13 met the inclusion criteria. Ten studies used active image capture ^{209-211, 222-224, 226-229} with handheld devices and three used passive image capture with wearable cameras. ^{212, 213, 230} Eight studies evaluated five different image-based food records captured using handheld digital cameras, personal digital assistants (PDAs) and smartphones. ^{209, 211, 222-224, 226-228} Two studies evaluated the use of images to assist traditional written food records; one used a single-use/disposable camera, ²¹⁰ and the other used a wearable camera. ²³⁰ Three studies evaluated the use of images to assist self-report during 24-hour dietary recalls. Two of the studies used wearable cameras^{212, 213} and one study used a

handheld digital camera. ²²⁹ Three other image-assisted methods under development were identified by the search strategy but had not been evaluated among users assessing dietary energy intake. ^{214, 231, 232}



Figure 12 Flowchart and inclusion process for the systematic review examining the evidence for image-assisted methods of dietary assessment.

5.6.2 Characteristics of included studies

The 13 included studies are presented in Table 7. The sample sizes of the studies were small. Six studies had fewer than 20 participants ^{210, 212, 213, 224, 227, 228} six studies had between 20 and 50 participants, ^{209, 222, 223, 226, 229, 230} and one study had 75 participants.²¹¹ Seven studies recruited healthy adults, ^{211-213, 222, 227-229} two recruited university students, ^{223, 226} two recruited adults with overweight and obesity, ^{209, 210} one study recruited adults with type 2 diabetes, ²²⁴ and one study recruited a combination of athletes and physically active university students. ²³⁰ The mean age was available for 11 studies, ^{209, 211-213, 222, 224, 226-230} with the mean age ranging from 18 to 65 years. No study had participants below 18 years or above 70 years of age. Sex was identified in all studies. Three studies recruited female participants only. ^{223, 226, 229} Five studies used criterion reference methods to evaluate EI data. ^{209, 212, 222, 228, 229} Two of the studies used doubly labelled water (DLW) to assess total energy expenditure (TEE),^{209, 212} and three used weighed meals. ^{210, 211, 213, 222, 224, 226, 227, 230} Four studies used weighed food records, ^{211, 223, 226, 227} three used estimated food records, ^{210, 224, 230} and one study used the 24-hour dietary recall. ²¹³

5.6.3 Image-based food records

For the purpose of this review an image-based food record is any method where images of foods captured during eating episodes provide the primary record of dietary intake to determine energy and nutrient content. The first attempt to validate an image-based food record used cameraenabled PDAs in the Wellnavi method. ^{223, 226} The Wellnavi method required users to capture images of foods, at a 45 degree angle, before and after eating episodes. Foods were placed on a table, and the PDAs stylus was placed beside foods as a visual reference for portion size estimation. After the images were captured, users were required to describe the foods and provide ingredients with written text on the screen (using the stylus), especially when the foods were transmitted wirelessly to a server for manual image analysis by dietitians. To aid analysis, a brief questionnaire was used to obtain additional information on dietary behaviors such as added sugar to beverages, and typical condiment use.

Two pilot studies were conducted among female nutrition students (N=20, and N=28) who simultaneously recorded dietary intake using the Wellnavi method and weighed food record.^{223,}

²²⁶ In the first study, diet was recorded for one-day, ²²³ and in the latter participants recorded diet for one-day in June and November. ²²⁶ Both studies found no significant differences in EI or macronutrients between the two methods. ^{223, 226} A larger validation study was then conducted by Kikunaga et al ²¹¹ among adults of the general Japanese population (N=75) who simultaneously recorded their diet for 7 days using the Wellnavi method and weighed food record. Compared to the food record the Wellnavi method underestimated mean EI by 13.1% (1977±405 kcal vs. 1718±361 kcal, P<0.001) and significantly under-estimated all macronutrients. The authors noted that a high proportion of images did not contain any text describing the foods, which made accurate image analysis challenging for complex and traditional Japanese dishes.¹⁶ Participant feedback in the pilot study indicated it was difficult to write text on the small screens (using the stylus), ²³ which was likely a factor that contributed to the low compliance and subsequent underestimation of EI. ²¹¹ Moreover, limitations of the PDA technology including the bulk of the device, poor battery life and image quality were apparent.

Lassen et al ²²⁷ conducted a study using a stand-alone handheld digital camera to evaluate the potential of the "Digital Method" image-based food record. While seated, users were required to capture images of foods on a table, at a 45 degree angle, before and after eating episodes. Users were required to separate different meal components on the plate to assist analysis and a rulers was placed beside foods as a reference for portion size estimation. A notebook was provided to record the recipes and ingredients in grams or common household measures. ²²⁷ Manual image analysis was conducted by two image analysts trained using a reference database of commonly consumed foods developed from a feasibility study. ²²⁷

A sample of healthy adults (N=19) simultaneously recorded their dinners for five nights (excluding beverages) using the Digital Method and weighed food records. Compared to the food record the Digital Method underestimated mean EI by 11.3% (dinner only; 526±178 kcal vs. 471±167 kcal, P<0.001). ²²⁷ Participant feedback revealed difficultly remembering to record intake, particularly when eating out, or for rapidly consumed items not eaten from a plate. Moreover, some participants noted it was awkward to separate foods on the plate before capturing an image. ²²⁷

Rollo et al ²²⁴ conducted a pilot study to test Nutricam an image-based food record application on a mobile phone. Similar to other methods before and after images were captured of foods on tables, at a 45 degree angle, before and after eating episodes. Additional images were captured if necessary to ensure the images were clear for analysis. A reference card was placed next to

Chapter 5. Image-assisted dietary assessment: a systematic review of the evidence

foods for portion size estimation, and provided prompts for a brief voice recording (≤30 seconds) to describe the food name, type, brand/product name and preparation/cooking method of each food item. The images and associated voice recordings were assessed independently by a dietitian. ²²⁴

Adults with type 2 diabetes (N=10) simultaneously recorded their dietary intake using Nutricam and estimated food record for three days. Compared to the food record Nutricam underestimated mean EI by 9.3% (1660±439 kcal vs. 1505±469 kcal, P<0.05). Only 71% of Nutricam entries included an image of adequate quality for analysis, and only 66% of the entries included a voice recording, which explained the underestimation. The authors also noted it was the difficult to analyze complex dishes. Participant feedback revealed memory failure as the most common reason why they did not capture an image of the foods consumed.²²⁴

The Remote Food Photography Method (RFPM), and the mobile phone Food Record (mpFR) are more sophisticated methods that incorporate automated image analysis techniques into comprehensive dietary assessment systems.^{233, 234, 20, 41} The RFPM was adapted from a validated method to assess EI using images in cafeteria settings (not self-captured).^{233, 234} Like other methods, users are required to capture images of foods on a table at 45 degrees, before and after eating episodes.²²² A reference card with a printed pattern is placed next to foods to correct for color and assist in the estimation of the food's area.²³⁵ The images captured are transmitted wirelessly to a server in near real-time for analysis using a custom program. Features from the images are extracted for each food identified for food classification, ²³⁵ and the program compares the foods with a searchable image archive of foods and portion sizes matched to the Food and Nutrient Database for Dietary Studies. ^{222, 236} The food area is converted to grams based on the association between food area and weight for each respective food. ²³⁵ Dietitians review the data and make changes as required and can contact users immediately if the images/data are of poor quality. ²³⁵ Users are also instructed to record dietary intake using pen and paper or leave a voice message describing the foods if they forget to capture images. To remind participants to record dietary intake, the RFPM incorporates the use of ecological momentary assessment (EMA); automated prompts at meal times requiring a user response.^{209, 222}

The validity and reliability of the RFPM was initially assessed using weighed meals and a manual procedure of image analysis. ²²² A sample of healthy adults (N=50) recorded their diet for three days using the RFPM. The dine-in group (n=25) recorded their lunch and dinners in the

laboratory, while the take-out group (n=25) recorded their lunch in the laboratory and dinner in simulated free-living conditions (pre-weighed foods provided in coolers). ²²² Compared to the weighed meals the RFPM under-estimated the mean EI by 4.7% (dine-in group), and 5.5% (take-out group) in laboratory conditions, and by 6.6% in simulated free-living conditions (takeout-group). ²²² An assessment of reliability between three dietitians analyzing the images demonstrated good agreement for EI ICC=0.88 (95% CI 0.81 – 0.91) and food type selection 0.99 ICC (95% 0.99, 0.99). ²²²

The RFPM was refined before a further development study among overweight and obese participants (N=40) evaluated different EMA approaches; standard EMA (n=24) and customized EMA (n=16). ²⁰⁹ Participants recorded their diet for one week during a two-week DLW protocol to assess total energy expenditure (EE), and were provided standard EMA (set meal times) or customized EMA (individualized meal times). Using standard EMA the mean EI:EE was underestimated by 36.3% compared to only 12.4% in the customized EMA group.²⁰⁹ After further refinement the RFPM was validated in a sample of overweight and obese participants in both laboratory (N=49, two weighed buffet meals) and free-living conditions (N=42, RFPM recorded diet for one week of a two-week DLW protocol). ²⁰⁹ Compared to weighed meals, mean EI estimated by the RFPM was very similar (587±209 kcal vs. 583±190 kcal, P=0.67), and compared to DLW in free living conditions the mean EI:EE was underestimated by 6.4% (2360±626 kcal vs. 2208±665 kcal, P=0.16). Participant feedback indicated the majority were satisfied with the method and indicated the method was easy to use compared to written records.

The mpFR uses a similar procedure to the RFPM with users required to capture images of foods on a table, at a 45 degree angle, before and after eating episodes. The images captured are transmitted wirelessly to a server for analysis using a custom program. A checked fiducial marker (reference marker) is placed next to foods to assist the automated system to estimate food volume. ²²⁰ The foods are segmented into individual food items using a series of techniques ²³⁷⁻²⁴⁰ before classification and volume estimation using calculations based on the food's shape. ²⁴⁰

Using a different approach to the RFPM, images of foods are labelled and the results are sent back to the user to confirm or modify the foods and portion size determined by the automated system (rather than by researchers). After user adjustments and confirmation the foods are indexed with the Food and Nutrient Database for Dietary Studies ²³⁶ before results are sent to

researchers/dietitians. A backup electronic food record is built into the smartphone application when users forget to capture images.

At the time of writing, only a pilot study (within a PhD thesis) designed to inform a larger validation of the mpFR has been reported which assessed energy intake among users. ²²⁸ The pilot used manual image analysis with trained image analysts (not the automated system described above). ²²⁸ A sample of adults from the campus community (N=12) were provided with a range of foods (portions known and excess to their energy requirements) on three non-consecutive days to eat in both laboratory and free-living conditions, and instructed to record all eating episodes with the mpFR. Compared to the presumed EI (determined from returned uneaten foods) the mean EI was under-estimated by 6.4%. (Note image analysts had knowledge of the foods and portions provided). Participant feedback was not reported, but other research evaluating usability of the mpFR indicated users find the method easy to use and is preferred over a traditional food record.^{241, 242}

5.6.4 Image-assisted food records

For the purpose of this review an image-assisted food record is any method where images captured during eating episodes are used to enhance or supplement a traditional text-based food record (written or electronic). Gregory et al ²¹⁰ conducted a feasibility study among adults with obesity (N=9) to explore the use of a handheld disposable/single-use film camera to enhance an estimated food record over three non-consecutive days. Participants were required to capture a picture at arm's length from the table, and placed a 15cm ruler as a reference for plate size. Foods were then recorded into a booklet provided. After the testing period the food record alone, reviewing the photographs increased the mean EI by 8% for dietitian #1 and by 5.2% for dietitian #2 but this was not statistically significant (P=0.87). No data pertaining to how the image review specifically changed EI was provided, but identification of misreporting errors was noted by the authors.

O'Loughlin et al ²³⁰ conducted a study to assess if images can enhance the food record using the wearable camera (SenseCam). The SenseCam is a wearable camera worn around the neck on a lanyard with a wide-angled lens, ^{243, 244} and images are captured passively at approximately 20 second intervals (~2000 to 3000 images per day). Internal flash memory is sufficient for one week
and battery capacity is adequate for a typical 12-16 hour day. Once turned on, the SenseCam operates continuously until the camera is switched off (a privacy button can be activated to cease image capture temporarily).

For one day, trainee jockeys (n=17), Gaelic footballers (n=15), and physically active university students (n=15) wore the SenseCam and recorded their diets using an estimated food record. After the testing period the food record was reviewed by the participant and dietitian for ambiguous information before viewing the SenseCam images. Compared to the food record alone, viewing the SenseCam images significantly increased mean EI by 12%, 23% and 11% for the trainee jockeys, Gaelic footballers, and physically active university students, respectively (P≤0.001, P≤0.001, and P≤0.01). ²³⁰ In all three groups only one participant's food record remained unchanged after image review. No data pertaining to exactly how the image review altered the EI was provided, but unreported foods and misreporting errors were noted by the authors. The study design had a risk for interviewer bias as changes to dietary intake were made by the same dietitian. Technical problems with SenseCam devices and user error resulted in incomplete data for 13 participants (28%), and were not included in the analysis. The authors also noted poor image quality in low light environments and devices were frequently not worn correctly affecting image quality.

5.6.5 Image-assisted 24-hour recalls

For the purpose of this review an image-assisted 24-hour dietary recall is any method where images captured during eating episodes are used to self-report during the 24-hour dietary recall method. Arab et al ²¹² first tested the feasibility of a web-based, self-administered, image-assisted 24-hour dietary recall (Image-DietDay) in a sample of healthy adults (N=14). For 6-10 days of a 15-day DLW protocol participants wore a customized mobile phone around the neck (using a lanyard) that captured images every 10 seconds during eating episodes. The images were transmitted wirelessly to a server for processing (blurry and dark images removed), and a selection of images (<100) were presented in an image-viewer to assist participants during three Image-Diet Day 24-hour recalls. Compared to DLW, Image-DietDay overestimated the mean EI:EE intake by 7%, 2711±1225 kcal vs. 2519±609 kcal, respectively. The study design did not permit any analysis regarding how the images assisted the recall but participant feedback indicated that most found the images helpful. However, in some cases wearing the phones may

have altered usual eating behaviors. The authors ²¹² also noted the battery life was not always sufficient to last an entire day and the phone's narrow field of view was not ideal for dietary assessment. ²¹² Other testing of the system also reported an imaging frequency of 10 seconds captured few images of rapidly consumed foods (e.g. fruit), and no images of socially undesirable foods, such as candy, or chips.²⁴⁵

Gemming et al²¹³ tested the feasibility of a SenseCam-assisted interviewer-administered 24-hour dietary recall in a sample of healthy adults (N=10). Participants wore the SenseCam for two days, while conducting their usual daily activities. Day one familiarized participants with the SenseCam and the images from day two were used to assist participants' self-report after the final pass of the 24-hour dietary recall. To reduce the potential for interviewer bias the researcher did not suggest changes or scrutinize self-reported intakes, but queried unreported food items present in the images. Compared with 24-hour dietary recall alone the images increased mean EI by 12.5% (2738±502 kcal vs. 3080±712 kcal, p=0.02). ²¹³ The increase was primarily due to 41 unreported food items across the sample. The unreported foods were from a range of food groups, and included both snack foods and more substantial food items. Eight changes to portions size and 12 misreporting errors were also identified by participants but together these changes had little impact on EI.²¹³ Participant feedback indicated the images were helpful and enabled participants to provide more accurate information, but some indicated they felt uncomfortable in public situations, such as riding the bus or purchasing foods. Limitations regarding SenseCam were also apparent. The imaging frequency was too slow to capture rapidly consumed foods, images were poor quality in low-light environments, and posture and body shape affected lens angle resulting in some non-useful images. ²¹³

Lazarte et al ²²⁹ examined the use of handheld digital cameras to enhance self-report in the Food Photography interviewer-administered 24-hour Multiple Pass Recall method (FP 24-hR). The FP 24-hR was validated among a group of healthy Bolivian women (N=43) for a single 24-hour period. Participants were provided a photo kit that contained a camera and mat with a 1.5cm grid and captured two images at 50cm distance from the table (at 90 and 45 degree angles) before and after eating episodes. ²²⁹ Researchers visited the participant's home during the testing period and weighed all meals. The following day a different trained interviewer assessed the participant's diet, and on the last pass of the 24-hour dietary recall the interviewer used the images (FP 24-hR) to confirm or modify the portion sizes and enquire about any foods that were partially obscured.

²²⁹ Compared to weighed meals the FP 24-hR under-estimated mean EI by only 4% (1456 kcal SEM 63 kcal vs. 1399 kcal SEM 62 kcal, p<0.05), and Bland-Altman plots revealed good agreement between methods with no systematic bias. ²²⁹ No specific analysis pertaining to how the images changed the initial recall was reported. ²²⁹

Table 6 Studies that have evaluated or validated an image-assisted method of dietary assessment compared to a reference method assessing dietary energy intake.

Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
Image-base	ed food records							
Lassen et al, 2010 ²²⁷	Digital Photography Method	Active Nikon Coolpix SS210 digital camera	Healthy adults / N=23 ^d Age: 37±16 BMI: 24±3	5-day weighed FR ^e (dinner only)	-11.3	<0.001	 Easy to complete High compliance (94%) Difficulties when away from home or food prepared by others Participants reported awkward to separate items Did not appear to influence eating habits to any great extent Sometimes larger servings were chosen or sauces not consumed to avoid taking extra images. 	No criterion measure, sample size, no beverages in analysis, food items separated on plate, only dinner
Kikunaga et al, 2007 ²¹¹	Wellnavi method	Active PDA ^f with camera + phone card	Healthy adults / N=75 Age: 49±10 BMI: 24±4	7-day weighed FR	-13.1	<0.001	 Most participants did not capture images at 45° angle which made image analysis difficult Low compliance using stylus made image analysis difficult without supporting text Traditional Japanese foods hard to visually identify 	No criterion measure, normal body size
Martin et al, 2009 ²²²	Remote Food Photography Method (RFPM)	Active Motorola i860 mobile phone	Healthy adults / N=50 Age: 32 (SEM 2) ^g BMI: 27 (SEM 1)	3-day weighed meals Dine-in group (n=25) (lunch & dinner in			 Most participants satisfied with the RFPM and ease of use Almost all participants rated that they prefer the RFPM over a written FR 	Food items separated on multiple plates, limited number of foods provided,

Iaboratory) -4.7 0.046 • Some participants forgot to take images of food images	Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
Study 1 Overweight & obese adults / N=40 Study 1 Study 2 Study 1 Study 1					laboratory) Take-out group (n=25) (lunch in laboratory) Take-out group (dinner in simulated free-living conditions)	-4.7 -5.5 -6.6	0.046	 Some participants forgot to take images of food Some EMA^h messages were sent at the wrong time. Suggestion to maintain a written FR in case of technology failure 	only dinner consumed in free-living conditions only 3-days of assessment
Martin et al, 2012 ²⁰⁹ Active Study 2 Overweight & obese adults / N=50 BMI:31±5 BMI:31±5 Martin et al, 2012 ²⁰⁹ Active Study 2 Overweight & obese adults / N=50 Age: 41±13 BMI:31±5 (N=49) ⁱ 1-week RFPM vs. 2-week DLW (N=42) ⁱ BMI:31±5 Customized EMA group (n=13) ⁱ -0.7 0.67 0.67 -0.7 0.67 -0.64 -0.16 -0.68				Study 1 Overweight & obese adults / N=40 Age: 43±14 BMI: 24±48	Study 1 1-week RFPM vs. 2-week DLW ⁱ Standard EMA group	-36.3 -12.4	<.0001 0.22		Study 1 No participants +65 years, small sample size in customized group, mainly females
2-week DLW (N=42) ^j	Martin et al, 2012 ²⁰⁹	Remote Food Photography	Active	Study 2 Overweight & obese adults / N=50 Age: 41±13 BMI:31±5	(n=22) ^r Customized EMA group (n=13) ^j Study 2 2 weighed buffet meals in laboratory (N=49) ^j 1-week RFPM vs.	-0.7 -6.4	0.67 0.16	 82% of participants rated their satisfaction with the method as 5 or higher on a 6 point Likert scale 93% and 89% rated the ease of use as 5 or higher 93% and 96% rated the usefulness of the run-in period and training as 5 or higher on a 6 point Likert scale 	Study 2 No participants +65 years, mainly females only 1-week of assessment compared to DLW
					2-week DLW (N=42) ⁱ				

Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
Wang et al, 2002 ²²³	Wellnavi method	Active PDA with camera + phone card	Female nutrition students / N=20 Age: NR ^k BMI: NR	1-day weighed FR	6	>0.05	 Certain foods hard to visualize separately for analysis Difficulty using the PDA stylus on small screen Battery charging was time consuming PDA was considered heavy Few (10%) indicated the method was satisfactory, 40% somewhat satisfactory, 25% somewhat unsatisfactory, and 25% participants unsure 	No criterion measure, sample size, female nutrition students, 1-day of assessment
Wang et al, 2006 ²²⁶	Wellnavi method	Active PDA with camera + phone card	Female nutrition students / N=28 Age: 20±5 BMI: 21±3	2 x 1-day weighed FR June & November	-3.8 (June) 2.3 (November)	>0.05 >0.05	 The majority considered the Wellnavi least burdensome compared to FR About half indicated they could continue using Wellnavi for one month Nearly one third indicated being self-conscious about their meals seen by strangers 	No criterion measure, sample size, female nutrition students, 2 x 1-day of assessment only
Rollo et al, 2011 ²²⁴	Nutricam	Active Sony Ericson K800i mobile phone	Adults with type 2 diabetes / N=10 Age: 65±34 BMI: 34±7	Estimated 3-day FR	-9.3	0.03	 Software was easy to use and study participants preferred Nutricam over the FR All subjects were confident they could use Nutricam to record their dietary intake for one month Some poor quality images restricted image analysis Participants commonly reported failure to use Nutricam at eating episodes 20s voice recording limit was too short 	No criterion measure sample size, no criterion measure, short duration, adults with type 2 diabetes only, only 3- days of assessment

Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
Schap et al, 2012 ²²⁸	Mobile phone Food Record (mpFR)	Active IPhone 3GS	Adults from campus community / N=12 Age: 23 ^I (median) BMI: 24±4	3 non-consecutive days of weighed meals (portions of known quantities provided)	-6.4	0.243	 Some foods were not photographed by participants 	Sample size, short duration, no criterion measure for total energy intake, foods items known by analysts, only 3-days of assessment

Image-assisted food records

			Adults with obesity /	Estimated 3-day FR				No criterion measure,
Gregory et al, 2006 ²¹⁰	Food Diary with photographs	Active Single-use camera	Addits with obesity / N=9 Age: NR BMI: >30	(non-consecutive) Dietitian 1	8	0.71	 Images revealed misreporting errors 	sample size, little detail on the changes to
				Dietitian 2	5.2	0.87		addition of photographs

Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
O'loughlin et al, 2013	SenseCam- assisted FR	Passive SenseCam wearable camera	Healthy young adults / N=47 Trainee jockeys / n=17 Age: 18±2 BMI: NR Gaelic footballers / n=15 Age: 22±1 BMI: NR Active university students / n=15 Age: 23±1 BMI: NR	Estimated 1-day FR alone Trainee jockeys (n=11) ^j Gaelic footballers (n=10) ^j Active university students (n=13) ^j	 12 ≤0.001 User and camera error resulted in 28% (n=13) of participant data excluded from analysis Camera not worn properly by all participants Poor image quality in low-light conditions 		No criterion measure, mainly physically active participants, no details on changes made due to images	
Image-assis	ted 24-hour dietary	/ recalls						
Arab et al, 2011 ²¹²	Image-DietDay 24-hour dietary recall	Semi-Passive Nokia N80 Mobile Phone	Healthy adults / N=14 Age: 35±12 BMI: 27±7	3 x Image-DietDay 24- hour dietary recall vs. 2- week DLW	7.6	NR	 No technical failures occurred Battery not always sufficient Device cumbersome to wear for the majority of participants 57% found the images helpful, and 79% were comfortable using the ImageViewer Sometimes eating behavior was affected (eating out less often and eating more rapidly) 	Sample size, predominantly females, motivated adults already participating in nutrition research

Author	Method	Method of image capture	Participants / (N) Mean Age (years±SD) ^a BMI (kg/m ² ±SD)	Reference method(s)	Difference compared to reference ^b (%)	P-Value ^c	Feedback of method/technology	Study limitations
Gemming et al, 2013 ²¹³	SenseCam- assisted 24- hour dietary recall	Passive SenseCam wearable camera	Healthy adults / N=10 Age:33±11 BMI:26±11	1 x 24-hour dietary recall alone	12.5	0.02	 Wearing SenseCam was a low burden Images helped participants remember unreported foods Posture and body shape affect lens direction Poor image quality in low-light environments Imaging frequency too low to capture all foods consumed Device fault resulted in loss of images for 2 participants 	No criterion measure, sample size, mainly males, only 1 x 24-hour dietary recall
Lazarte et al, 2012 ²²⁹	Food Photography 24h recall (FP 24-hR)	Active Samsung Digimax S760 digital camera	Adult women / N=43 Age: 35±9 BMI: 25±4	1-day weighed meals (kept by research assistant)	-3.9	<0.05	 Some foods obscured in images Memory lapses could not always identify unknown food 	Only 1-day of assessment, females only

^aSD = Standard deviation; ^bDifference in mean energy intake compared to reference method (%); ^cExact P-value if reported; ^dEnergy intake data for 19 participants; ^eFR = Food Record; ^fPDA = Personal Digital Assistant; ⁹SEM = Standard error of the mean; ^hEMA = Ecological Momentary Assessment; ⁱDoubly labelled water; ^jFewer participants due to dropouts and/or excluded data; ^kNR = not reported; ^lStandard deviation not reported.

5.7. Discussion

We conducted the first systematic review to examine studies which have evaluated or validated methods of image-assisted dietary assessment. Research to date has primarily explored the potential of image-assisted methods in pilot and feasibility studies, and few methods have been formally validated using criterion measures and adequately sized samples. ^{209, 222, 229} However, several studies have demonstrated how images can enhance self-reported dietary intake by revealing unreported foods and misreporting errors. ^{210, 213, 230} The additional dietary information obtained from the images appears to increase reported El ^{210, 213, 230} and likely reduces random errors. ⁹⁶ Furthermore when used as the primary record of dietary intake, images can be analyzed to obtain valid and reliable estimates of El with reduced measurement error (for El) compared to traditional methods. ^{26, 131, 152-154, 209, 216-219, 222} However, due to the complexity and diversity of foods, El is likely to be under-estimated if the methods procedure is not followed correctly by the user, the images are of poor quality, or if the user forgets to capture images prior to the eating episode. ^{209, 211, 224, 227}

Trade-offs during image analysis are also made as certain dietary components including hidden ingredients and cooking method, which affect energy and nutrient composition cannot be determined with image analysis alone. Consequently, it appears essential that images of foods must be supported by additional dietary information to achieve optimal accuracy. ^{211, 222, 224} However, the coding errors associated with image analysis are likely to be random,⁹⁶ which are less problematic compared to the systematic bias observed when food type and portion size are self-reported. ^{27, 28, 119, 246, 247} Thus there are both strengths and limitations of using images to assess dietary intake, but these cannot be entirely understood until further high quality studies have been conducted. Moreover, as image-assisted methods and systems continue to develop, the strengths and limitations will also evolve.

The search strategy identified three other image-assisted methods in development (yet to be evaluated among users) ^{31, 214, 231, 232} that have innovative features that may enhance image-assisted methods further. Two of these methods, The Food Intake Visual Recognizer (FIVR),²³¹ and The Diet Data Recorder System (DDRS) ²³² are image-based food records (on smartphones) that also incorporate the use of automated image analysis systems. ^{231, 232, 248} However, FIVR aims to incorporates voice recognition software to clarify details of dietary intake, and DDRS

removes the need for a reference marker by using a laser within a smartphone case to project a visual reference to assist analysis.²³²

Techniques to analyze images without a reference marker have also been described for the bespoke wearable camera eButton (worn on the chest) specifically designed for the passive assessment of dietary intake and physical activity. ^{214, 221, 249} The eButton and other future wearable devices may address some technical limitations, such as insufficient battery life, poor quality images, and insufficient imaging frequencies ^{212, 213, 230} but will need to capture useable images in all environments and during non-daylight hours to effectively record dietary intake. ^{212, 213, 230} Considering the diversity and complexity of dietary intake in free living environments,

accurate assessment of dietary intake using passive methods alone will be very challenging.

Privacy concerns also need to be addressed. Though wearable cameras capture images of third parties in a similar manner to smartphones or security cameras, their acceptance and etiquette for use in society is yet to be established. The limited release of smart-wearable eyewear by Google "Glass" has received substantial media attention regarding the devices innovative applications, but also concerns regarding privacy at a government level. ²⁵⁰ It appears wearable devices need to be paused or switched off in certain buildings or locations, and images could be captured accidently when the user or others expect privacy, especially problematic if the research involves children and the images are transmitted automatically in real-time. In order to address privacy concerns related to the use of wearable cameras in health research, Kelly et al ²⁵¹ have developed an ethical framework that provides guidelines for best practice. Other potential safeguards could also alleviate privacy concerns, such as automated face-blurring, and sensors (accelerometers) to trigger image capture or store images (in memory) only when eating is detected.^{252, 253} However, it is clear that further research is required to better understand these ethical issues and potential solutions.

Other questions regarding image-assisted dietary assessment remain. No study has validated an image-assisted method among the elderly (>70 years), children and adolescents, and the feasibility of image-assisted methods in large studies (N>100) has not been demonstrated. Older adults may fail to remember the method more often and are potentially less accustomed to using smartphones and other technologies, while usability studies among children and adolescents indicate they prefer methods using technologies over traditional methods.^{241, 254, 255} With respect to

large samples, participants will need to own or be provided with a suitable device (to reduce the potential for respondent bias), ⁹⁶ which increases study costs.

Limitations of the review need to be considered. The majority of studies reviewed were pilot and feasibility studies, thus future research may not reflect the results of these preliminary studies. Due to the heterogeneity of the technologies used, methods employed to assess dietary intake and different study designs, only a narrative review could be conducted. Furthermore limiting the search strategy to scientific journal articles may have excluded the most recent technologies or unpublished information as some commercial methods may not undergo scientific testing.

Study heterogeneity also made evaluating study quality challenging. The European Micronutrient Recommendations Aligned (EURRECA) Network of Excellence scoring system is a useful tool to objectively assess the quality of dietary intake validation studies, ²⁵⁶ but is not particulalry suited to rating the quality of studies evaluating image-based methods. The EURRECA system allocates points for the data gathered by a face-to-face interview, but image-based methods can be assessed remotely by independent analysts/researchers. ²⁵⁶ Development of a scoring system that differentiates between traditional self-report data and data obtained independently from image analysis (manual or automated) would be useful.

5.8. Conclusions

Current evidence regarding the validity of image-assisted methods of dietary assessment is limited, but studies to date have demonstrated images can enhance self-report data, and provide the primary record of dietary intake to obtain valid estimates of EI (when incorporated into a comprehensive system). Additional validation studies using criterion measures are needed. The validity among children, adolescents, and the elderly, as well as the feasibility of using imageassisted methods in large samples, warrants examination. Wearable cameras are a recent development in dietary assessment but further research is required to better understand the potential applications and pitfalls of using wearable technologies.

CHAPTER 6. FEASIBILITY OF A SENSECAM-ASSISTED 24h RECALL TO REDUCE UNDER-REPORTING OF ENERGY INTAKE

6.1. Introduction to publication

A feasibility study was required to explore if wearable camera images can be incorporated into and enhance self-report in the 24-hour dietary recall method, and establish if the devices and methods used would be suitable for a larger validation study. This chapter (Chapter 6) is a reformatted version of the manuscript entitled "Feasibility of a SenseCam-assisted 24-h recall to reduce under-reporting of energy intake" published in the European Journal of Clinical Nutrition, 2013, Volume 68, Issue 10, Pages 1095-1099. DOI 10.1038/ejcn.2013.156. The latest impact factor available was 2.95 (2013). The chapter presents the results from the feasibility study, which explored the potential of wearable cameras to enhance self-report in 24-hour dietary recalls. This study was undertaken during a three-month research exchange in 2012, at the British Heart Foundation Health Promotion Research Group, University of Oxford, where the candidate was familiarised with the wearable camera, SenseCam and software used for the analysis (under supervision by Dr Aiden Doherty who developed the software and has expertise in the field). The study was undertaken to address Objective Three, which was to explore the feasibility of using wearable cameras to assist an interviewer-administered 24-hour dietary recall.

6.2. Author contribution

Luke Gemming was involved in developing the research question, ethics applications, study design, recruitment, data collection and data analysis. He also interpreted the findings and wrote the paper for publication.

FEASIBILITY OF A SENSECAM-ASSISTED 24H RECALL TO REDUCE UNDER-REPORTING OF ENERGY INTAKE

^aLuke Gemming MSc

^bAiden Doherty PhD

^bPaul Kelly MSc

^cJennifer Utter PhD

^aCliona Ni Mhurchu PhD

^aPostal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

^bPostal Address: British Heart Foundation Health Promotion Research Group

Old Road Campus, Headington, University of Oxford, OX3 7LF, United Kingdom

^cSchool of Population Health, University of Auckland, Private Bag 92019, Auckland, 1142, New Zealand

Address for correspondence:

Luke Gemming

Postal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

l.gemming@auckland.ac.nz

6.3. Abstract

BACKGROUND/OBJECTIVES: The SenseCam is a camera worn on a lanyard around the neck that automatically captures point-of-view images in response to movement, heat, and light (every 20 to 30s). This device may enhance the accuracy of self-reported dietary intake by assisting participants' recall of food and beverage consumption. It was the objective of this study to evaluate if the wearable camera, SenseCam, can enhance the 24-hour dietary recall by providing visual prompts to improve recall of food and beverage consumption.

SUBJECT/METHODS: Thirteen volunteer adults in Oxford, United Kingdom were recruited. Participants wore the SenseCam for two days while continuing their usual daily activities. On day 3, participants' diets were assessed using an interviewer-administered 24-hour recall. SenseCam images were then shown to the participants and any additional dietary information that participants provided after viewing the images was recorded. Energy and macronutrient intakes were compared between the 24-hour recall and 24-hour recall+SenseCam.

RESULTS: Data from 10 participants were included in the final analysis (8 males and 2 females), mean age 33 ± 11 yrs, mean BMI 25.9 ±5.1 kg/m². Viewing the SenseCam images increased self-reported EI by approximately 1432 ±1564 KJ or 12.5% compared with the 24-hour recall alone (p=0.02). The increase was predominantly due to reporting of 41 additional foods (241 vs. 282 total foods) across a range of food groups. Eight changes in portion size were made which resulted in a negligible change to EI.

CONCLUSIONS: Wearable cameras are promising method to enhance the accuracy of self-reported dietary assessment methods.

6.4. Introduction

The 24-hour dietary recall is widely used in dietary assessment due to its low participant burden, ease of administration, and suitability for a wide range of populations including participants with low literacy. ² However, as a retrospective method it relies on self-report without verification, thus lapses in participant's memory, errors in portion size estimation, or intentional misreporting remain unidentified, but contribute to measurement error. ^{8, 257, 258}

Analyses of data from national nutrition surveys in many countries including Australia, New Zealand, and the United States of America which used the 24-hour recall have demonstrated substantial under-reporting of energy intake (EI), particularly in females, overweight, and obese participants. ^{70, 71, 73} Considerable under-reporting has also been revealed in doubly labelled water validation studies using the 24-hour recall administered in-person or via the telephone. ²⁵⁸⁻²⁶¹ Under-reporting leads to attenuation and misclassification error, thus impeding the ability to identify associations between diet and disease. Consequently the need for technological innovation is well recognised,^{15, 16} as technology has the potential to provide objective dietary intake data, reduced participant burden, standardised and automated methods of assessment, and new methods, previously not feasible, to assess dietary intake.

The development of structured interviewer-administered and self-administered computer based 24-hour recall systems can reduce interviewer bias and measurement error, but such systems still rely on self-report without the ability to verify dietary recall. ^{20, 26, 262} The recent development of wearable cameras with a point-of-view lens may provide a complementary objective measure to assist self-report, as these devices can record food consumption objectively and passively; which was previously only possible using a trained observer. Such devices are common place in adventure sports, but have recently been integrated directly into smart glasses (e.g. Google Glass) for everyday use, to complement and enhance smart phone features. Thus the rapid development and use of wearable cameras in society may provide new opportunities to obtain objective dietary intake data. ²⁶³

Arab et al. ²¹ first demonstrated the feasibility of wearable cameras for dietary assessment in using a customised mobile phone worn around the neck that captured images automatically every 10s. The images were sent to an internet server automatically and assisted participants self-report dietary intake using a web-based 24-hour recall (Image-DietDay) which revealed promising

results ²¹²; however, the cameras narrow field of view and insufficient battery life were noted limitations.

The SenseCam is a wearable camera worn around the neck and captures wide angle point-ofview images passively, every twenty to thirty seconds, in response to movement (tri-axil accelerometer), heat (infrared heat sensor), and light (light intensity sensor). ²⁴³ SenseCam and other similar devices can capture images automatically over entire days, weeks, or years, and are used to create digital life-logs. Life-logging is the digital capture and storage of personal data with the aim to record complete and searchable personal digital archives to assist people with everyday tasks, and/or remember past events. ²⁶⁴⁻²⁶⁶ Thus wearable cameras, such as SenseCam, are used to assist people with memory impairment, as the captured images provide a powerful cue for memory recall, ^{33, 267} but have also been used in health research, ²⁴⁴ to enhance self-reported measures of physically active travel, ^{268, 269} sedentary behaviour, ²⁷⁰ physical activities, ²⁷¹ and as a supplement to food records. ²⁷²

Wearable cameras may also enhance the 24-hour dietary recall by providing visual prompts to improve recall of food and beverage consumption. Therefore we conducted a feasibility study to evaluate the potential of the SenseCam to assist an interviewer-administered 24-hour dietary recall. The study was designed to assess the degree to which the images assisted and changed the participants' self-reported dietary intake. Specific objectives' were to determine the effect viewing the images had on self-reported energy and nutrient intakes.

6.5. Methods

A convenience sample of thirteen healthy adults between the age of 18 and 65 years were recruited through an advertisement posted on notice boards in Oxfordshire, United Kingdom. Participants were excluded if they followed strict dietary regimes (e.g. vegan) to ensure a wide range of foods were captured, or were unable to complete usual activities of daily living. Participants were provided a brief training session, and information sheet, explaining how to use and wear the SenseCam. The SenseCam is very simple to operate and only requires the participant to turn the device on for continuous operation. A privacy button can be activated to temporally stop image capture when required (SenseCam automatically starts capturing images again after 7 minutes).

V=V=List of research project topics and materials

Participants were instructed to wear the SenseCam for two full days whilst continuing their usual daily activities. The two-day period allowed participants to become familiar with using the SenseCam and participants were informed their diet would be assessed using a standard method for nutrition research. On day three, participants had their diet assessed over the previous 24-hours (day 2) by a trained dietitian (LG) using an interviewer administered multiple pass 24-hour dietary recall (MP24), based on the United States Department of Agriculture multiple pass method. ²⁶ A portion size guide and standard household measures were used to assist participants' self-reported dietary intake. ²⁷³

After the MP24, participants were given the opportunity to privately screen the captured images using a freely available SenseCam browser ²⁷⁴ and instructed to delete any photos they wished. This procedure followed ethical guidelines for SenseCam research to ensure the privacy of the participants was maintained (some images may contain private and/or sensitive content). The remaining SenseCam images were viewed by participants and dietitian together to identify all eating episodes and to confirm or modify details reported in the initial 24-hour recall (MP24+SenseCam). Dietary details recorded in the MP24 were re-stated, but the information was not scrutinised and no changes were suggested by the dietitian to reduce the possibility of interviewer-bias. However, the dietitian did query unreported foods visible in the images and participants confirmed or modified these foods accordingly.

Unreported foods, changes to portion size, and other misreporting errors (e.g. exchanging or removing foods) were grouped by the following food categories: breads and cereals; beverages (including milk; excluding water due to no energy content); fruit and vegetables; meats and dairy (e.g. cheese); biscuits, sweets, and snacks; condiments (spreads, sauces, dips and dressings). After completion of the MP24+SC (SenseCam) participants completed a brief feedback survey to explore the user experience of wearing the device. The survey used seven point Likert scales, or categorical scales with open text response sections. The study was approved by the Central University Research Ethics Committee University of Oxford (Ref: SSD/CUREC1A/12-008), and the University of Auckland Human Participants Ethics Committee (Ref:7942).

6.5.1 Analysis

Foods and beverages were analysed using nutrient analysis software WISP (Tinuviel Software, V 3.0, Warrington, United Kingdom). Data analysis was performed using SPSS Statistics (V 20.0,

IBM, Armonk, New York, USA). A paired t-test was used to compare the differences in selfreported energy and nutrient intakes between the MP24 and MP24+SC. Participant characteristics were described with summary statistics. Statistical significance was set at $\alpha \leq 0.05$.

6.6. Results

6.6.1 Participants

Thirteen participants were recruited but ten participants (8 males and 2 females), aged 33±11 years with a mean BMI 25.9±5.1kg/m² were included in the analysis (Table 7). Two participants were excluded due to a technical fault and loss of images, and one was excluded for protocol non-adherence (SenseCam infrequently worn). The participants were predominantly healthy and physically active (excluding one participant with obesity), all had completed a bachelor's degree or higher, and all were currently employed in professional positions. The excluded participants (1 Male, 2 Females) were slightly older 39±13 years but had a similar BMI and education.

Participant characteristics (n=10)								
Male (n)	8							
Female (n)	2							
Mean ± SD								
Age (yrs)	33 ± 11.3							
Height (m)	1.78 ± 0.1							
Weight (kg)	82.3 ± 21.8							
BMI (kg/m ²)	25.9 ± 5.51							

Table 7 Characteristics for the ten participants included in the analysis

6.6.2 Dietary recall

Energy and macronutrient intakes are presented in **Table 8**. The MP24+SC resulted in a significantly greater reported mean EI than the MP24 alone (12888 \pm 2977KJ vs. 11455 \pm 2099KJ, p=0.02) with a mean difference of 12.5%. Significantly higher reported intakes of protein, total fat, saturated fat, and mono-unsaturated fat were also evident with the MP24+SC compared to the MP24.

Table 8 Energy and macronutrient intakes from one interviewer-administered multiple pass 24-hour dietary recall (MP24), and the MP24 with the assistance of SenseCam images (MP24+SC).

	MP24	MP24+SC	
Nutrient	(n=10)	(n=10)	Mean Difference
Energy (KJ)	11 455 ± 2099.1	12 887.6 ± 2977.3	1431.8 ± 1563.7*
Protein (g)	106.7 ± 45.9	110.2 ± 64.0	17.1 ± 22.2*
Carb (g)	317.28 ± 63.7	355.2 ± 67.3	38.0 ± 56.7
Fat (g)	121.82 ± 32.7	135.9 ± 42.4	14.1 ± 16.9*
SAFA (g)	35.6 ± 13.27	40.8 ± 16.6	$5.2 \pm 6.4^*$
MUFA (g)	37.5 ± 15.8	43.1 ± 16.7	5.6 ± 6.1*
PUFA (g)	21.0 ± 13.1	25.6 ± 12.3	4.7 ± 8.0
Fibre (g)	20.3 ± 8.1	22.7 ± 9.4*	2.4 ± 2.8*

Means and standard deviations, difference = MP24+SC - MP24, *P<0.05

SAFA: saturated fat, MUFA: monounsaturated fat, PUFA: polyunsaturated fat.

Information on unreported foods, changes to portion size and misreported foods detected by SenseCam are presented in Table 9. A total of 41 unreported foods in the MP24 were revealed by viewing the SenseCam images (MP24+SC), 241 vs. 282 total foods respectively (additional 17% foods items). The additional foods were from all food categories and had energy contents from 0KJ to 1820KJ. The beverage category had greatest number of unreported items (n=9) but provided less energy than other groups excluding fruits and vegetables. Unreported breads and cereals, meats and dairy, and biscuits sweets, and snacks accounted for the greatest additional energy. The condiments category had the fewest unreported foods (n=4) but on average these additions each provided substantial additional energy (773 \pm 737KJ).

There were only eight changes to portion size, (five increases in portion size, three reductions in portion size) that provided little effect on EI overall (-259KJ to 189KJ). Misreporting errors i.e. foods incorrectly reported in the initial MP24 and modified after image review (e.g. salmon exchanged for chicken) had a greater effect on EI (-787KJ to165KJ) compared to changes to portion sizes and the majority of misreporting errors were in the fruits and vegetables category (n=8).

6.6.3 Feedback survey

Results from the participant feedback survey are presented in **Table 10**. The survey revealed participants found the images helped them remember foods they had forgotten about (median score 7) and were able to provide more accurate information (median score 7) but had less impact on helping them to remember extra details of how foods were prepared or purchased (median score 4). Participants indicated wearing the SenseCam and using it to assess their diet was a low burden but three sometimes felt uncomfortable in public situations, such as riding the bus or purchasing items (open text feedback). Wearing the SenseCam never (50%), seldom (10%), or sometimes (40%) affected dietary behaviours. Qualitative feedback revealed five participants were conscious of the device during food consumption and some considered making healthier food choices and may have eaten fewer snacks, but the degree of behaviour change was not clear. Participants indicated that about one week would be the maximum time they would wear a SenseCam.

Table 9 Energy provided from unreported foods, change to portion size or misreported foods revealed from one SenseCam-assisted interviewer administered MP24+SC.

r	า=10	Alterations	Breads &	Beverages	Fruit and	Biscuits,	Meats and	Spreads,
			Cereals	(excl. water)	Vegetables	Snacks, Sweets	Proteins	Sauces,
								Dressings
		(n)	7	9	7	8	6	4
ported	spo	Energy (KJ)			000 0 474 4	540 7 004 0		
Inre	Бо	Mean ± SD	585.6±255.0	277.2±192.9	323.2±171.1	510.7±294.2	688.7±535.2	773.2±736.8
)		Total	4099.5	2494.5	2262.3	4086.1	4132.1	3092.8
ze		(n)	1	3	1	1	1	1
on Si		Energy (KJ)						
ortic		Mean ± SD	-	-55.2±34.8	-	-	-	-
∆ Pc		Total	112.1	-165.6	5.64	-68.5	-258.6	189.9
σ		(n)	0	2	8	0	1	1
orte	s	Energy (KJ)						
repo	00	Mean ± SD	-	-73.9±149.9	-11.5±433.6	-	-	-
Mis	<u></u>	Total	-	147.7	-80.3	-	1087.2	1651.6

Table 10 Participant evaluation of the SenseCam-assisted 24-hour recall(MP24+SC).

Median values, n=10 Strongly disagree = 1 to strong									
The images helped me to remember s	ome foods I h	ad forgotten a	about		7				
The images helped me to remember ex		4							
prepared or purchased									
The images helped me to remember the portion size of the foods I ate 5									
The images helped me to verify the pol	rtion size of fo	ods I ate usir	ng the portion size		6				
guides									
The images allowed me to provide more accurate information about the foods I ate 7									
very low burden =1 to very high burden = 7*									
Was the camera a burden to wear?2.5									
Was reviewing the images on the computer a burden? 2									
Was the overall method to assess your	diet a burden	1?			2				
	Never	Seldom	Sometimes	Usually	Always				
Wearing the camera affected my eating) F	4	4	0	0				
behaviour (n)	5	I	4	0	0				
I felt uncomfortable wearing the	0		0	0	0				
camera (n)	3	4	3	0	U				
1 d	ay 2-3 day	s Week	Fortnight	Month	> Month				
What is the maximum period you would wear the camera for?	0 4	5	0	1	0				

6.7. Discussion

This study demonstrated that a wearable camera can be used to enhance the 24-hour dietary recall by revealing unreported foods and misreporting errors. Overall reviewing the images revealed 17% additional food items and increased reported EI by 12.5%.

As a novel study no direct comparison was possible, however, the 24-hour recall method typically provides EI data that is under-reported by approximately 8-24% compared to doubly labelled water. ⁸ Thus the increase in self-reported EI, revealed from viewing passively captured images, indicates that wearable cameras may help to reduce levels of under-reporting. Though, additional and more rigorous testing of wearable cameras is necessary.

O'Loughlin et al ²⁷² conducted a similar study (n=34) in Ireland that used SenseCam to enhance a 1-day food kept record by trainee Jockeys, Gaelic football players, and university students which revealed a 10-18% increase in EI after image review. No data regarding what accounted for the increase was reported but unreported foods and changes to portion size were indicated by the investigators. Arab et al. ²¹² also reported promising results using images captured automatically on mobile phones to assist a self-reported 24 recall (Image-DietDay) but the study did not classify how the images assisted recall.

Other image-assisted methods to assess dietary intake have are also in development. Sun et al ²⁰⁵ has described a customised wearable camera system for objective dietary assessment. Similar to the SenseCam the device is worn at chest height but aims to use automated images analysis to objectively assess intake, however, at the time of writing no dietary intake data has been presented. Other research investigating image-assisted electronic food records on smartphones (or cameras) is encouraging. ^{224, 242, 275} However, this approach is fundamentally different as participants are still required to initiate data collection, thus a similar burden to traditional self-report methods is placed on the participant to actively record food consumption. Though, it is likely future developments of technology will allow these two image-assisted approaches to complement each other in an effort to objectively assess dietary intake.

The use of images captured from a wearable camera to assist dietary recall in this study revealed a diverse range of unreported foods throughout the day (only one food item revealed at supper). Some foods, including tea, black coffee, mandarins, and peas provided little energy but other foods including avocado on toast, potato crisps, cheese, and lemon drizzle cake provided considerable additional energy. The wide variety of foods and the fact that both healthy and unhealthy foods were under-reported indicated under-reporting was likely random. The degree of under-reporting revealed may reflect the physically active sample, as increased levels of physical and varied lifestyle have been shown to reduce the accuracy of self-reported intake.^{157, 272} Conversely, this finding may simply reflect widespread under-reporting that plagues traditional dietary assessment methods.⁸

The misreported foods, as identified by participants (i.e. removed or corrected following image review), further exposed and reduced random error associated with existing dietary assessment methods. The errors were predominately single food items (e.g. orange corrected to banana) but some involved multiple foods within the same eating episode (small salmon fillet and salad corrected to chicken breast, jacket potato, with salad and caesar dressing). Fewer changes to portion size were probably attributed to poor image quality (foods in low-light, food partially obscured, or undesirable angles) and the decision not to question the participant's self-report. This reduced the potential for interviewer bias and any change to portion size may lead to a correlated error (as the true portion size remained unknown).^{119,276,277}

Participant feedback indicated viewing the images assisted with portion size estimation but also revealed wearing the camera may have affected dietary behaviours; thus may not be a true representation of usual intake. Similar participant feedback was reported by Arab et al. (24) with mobile phones worn around the neck to capture dietary intake. Nearly all participants found the images helpful during the Image-DietDay 24-hour recall, but some were self-conscious in public which may have affected normal behaviour.

In the present study the magnitude of behaviour change was not clear but social desirability and other psychosocial factors including social approval and fear of negative evaluation affect dietary behaviours ^{154, 246, 247, 278, 279} and self-reported dietary intake. ^{206, 260, 280} Wearable cameras may exacerbate behaviour change as devices are visible to others and objectively record food consumption in real-time. Moreover the ability to omit foods during subsequent self-reported dietary intake is minimized as all food are recorded (unless the device was switched off) therefore may further affect behaviours, especially in groups prone to misreporting including women, overweight and obese. ^{152, 197, 217}

It was observed that often participants' memory was prompted to recall additional foods before the specific image revealing the unreported food was displayed. The mechanism that acts to enhance

memory recall is unknown, but it's suggested images are possibly captured at the same moment memory encoding is taking place ²⁶⁶, and functional magnetic resonance imaging (fMRI) has revealed regions of the brain where memories are processed and stored are activated when viewing SenseCam images, but not activated reviewing a written diary of the same events. ³² Thus viewing the images provides both a powerful cue for memory recall and an objective record of the day's events if memory isn't prompted. In combination this provides participants a better opportunity to correctly self-report dietary intake and reduce under-reporting.

The strength and novel feature of this study was the direct comparison between traditional selfreport and self-report with the assistance of SenseCam. Often studies identify under-reporting but can only suggest why or how under-reporting has occurred. The SenseCam and other wearable cameras allow under-reporting to be explored in detail providing insight into the cause of underreporting, previously not feasible in a free living situation.

However, the SenseCam has some limitations. The imaging frequency of 20-30sec was not always sufficient to capture all foods consumed, posture and body shape can affect lens angle resulting in non-useful images (more commonly in women), and poor image quality in low-light environments were limitations of the SenseCam device with respect to assessing dietary intakes. Further, the device malfunctioned for two participants that resulted in a loss of images and one participant was non-compliant (camera worn infrequently). Similar reliability and compliance issues were noted by O'Loughlin et al when the SenseCam was to enhance written food records.²⁷² These issues would need to be resolved before wide-scale use of wearable cameras in dietary assessment.

This study also had limitations. A sole dietitian (LG) conducted the dietary recalls without a goldstandard reference method thus interviewer error cannot be ruled out, and the sample size was small and primarily consisted of well-educated active adult men. Therefore the findings of this pilot may not apply to women or other groups, and should be repeated in an adequately sized sample to confirm these findings.

6.8. Conclusion

In summary the use of wearable cameras appears to be a promising method to enhance selfreported dietary assessment. Viewing the images revealed a number of unreported foods, and

other misreporting errors, which increased self-reported EI. Overall this provided a clear indication to how wearable cameras may reduce levels under-reporting. Further research using goldstandard methods, such as doubly labelled water, is required to validate the use of wearable cameras in dietary assessment, and the degree to which wearable cameras affect dietary intake and behaviours should be explored.

Acknowledgements

We would like to thank the participants and the British Heart Foundation Heart Promotion Research Group, University of Oxford for their expertise, guidance, and technical support for SenseCam research.

Conflict of Interest

There were no conflicts of interest.



CHAPTER 7. WEARABLE CAMERAS CAN REDUCE DIETARY UNDER-REPORTING: DOUBLY LABELLED WATER VALIDATION OF A CAMERA-ASSISTED 24-HR RECALL

7.1. Introduction to publication

The feasibility study indicated wearable cameras may enhance self-report in the 24-hour dietary recall by revealing unreported foods and misreporting errors. Therefore, if similar findings could be demonstrated using a criterion reference method this would provide strong evidence supporting their use in future methods of dietary assessment and nutrition research. This chapter (Chapter Seven) is a reformatted version of the manuscript entitled "Wearable cameras can reduce dietary under-reporting: Doubly labelled water validation of a camera-assisted 24-HR recall" was published in the British Journal of Nutrition, 2015 volume 113, pages 284-291 DOI: 10.1017/s0007114514003602. The latest impact factor available was of 3.302 (2012). A few minor revisions were made to this reformatted manuscript at the request of the thesis examiners. The chapter presents the results from a validation study that examined the magnitude to which wearable cameras enhance self-report in 24-hour dietary recalls. The study was undertaken to address Objective Four, which was to validate a wearable camera image-assisted 24-hour dietary recall using the criterion measure doubly labelled water.

7.2. Author contribution

Luke Gemming was involved in developing the research question, ethics application, study design, recruitment, study management, data collection and analysis. He also interpreted the findings, and wrote the paper for publication.

WEARABLE CAMERAS CAN REDUCE DIETARY UNDER-REPORTING: DOUBLY LABELLED WATER VALIDATION OF A CAMERA-ASSISTED 24-HR RECALL

^aLuke Gemming MSc

^bElaine Rush PhD

^aRalph Maddison PhD

^cAiden Doherty PhD

^dNicholas Gant PhD

^eJennifer Utter PhD

^aCliona Ni Mhurchu PhD

^aPostal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

^bSport and Recreation, Auckland University of Technology, Private Bag 92006, Auckland 1142, New Zealand

^bBritish Heart Foundation Health Promotion Research Group, Nuffield Dept of Population Health, Old Road Campus, University of Oxford, OX3 7LF, United Kingdom

^bDepartment of Sport & Exercise Science, Faculty of Science, University of Auckland

^eSchool of Population Health, University of Auckland, Private Bag 92019, Auckland, 1142, New Zealand

Address for correspondence:

Luke Gemming

Postal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

l.gemming@auckland.ac.nz

7.3. Abstract

Preliminary research suggests that wearable cameras may reduce under-reporting of energy intake (EI) in self-reported dietary assessment. The aim of the present study was to test the validity of a wearable camera-assisted 24h dietary recall against doubly labelled water. Total energy expenditure (TEE) was assessed over a 15-day DLW protocol among 40 adults (n 20 males; age 35, SD 17 yrs; BMI 27, SD 8; and 20 females; age 28, SD 7yrs; BMI 22, SD 2). EI was assessed using three multiple-pass 24h dietary recalls between days 2-4, 8-10, and 13-15 (MP24). On the days prior to each nutrition assessment, participants wore an automated wearable camera (SenseCam) in free-living conditions. The wearable camera images were viewed by participants following completion of the dietary recall, and their changes to self-reported intakes were recorded (MP24+SenseCam). TEE and EIs assessed by MP24 and MP24+SC were compared. Among men, MP24 and MP24+SC underestimated TEE by 17% and 9%, respectively (P<0.001 and P=0.02). Among women, MP24 and MP24+SC underestimated TEE by 13% and 7%, respectively (P<0.001 and P=0.004). The assistance of the wearable camera (MP24+SC) reduced the magnitude of under-reporting by 8% for men and 6% women compared to the MP24 alone (P<0.001, and P<0.001). The increase in EI was predominantly from 265 unreported foods revealed by participants during the image review. Wearable cameras enhance the accuracy of self-report by providing passive and objective information regarding dietary intake. High definition image sensors and increased imaging frequency may improve the accuracy further

7.4. Introduction

Traditional methods of dietary assessment are subjective and rely on self-report. Therefore the accuracy of dietary data is influenced by memory recall, burden of administration, ² psychosocial factors, ^{28, 278, 281} and other behavioural characteristics ^{134, 156-158} with gender, ^{73, 150, 151} age, ^{70, 73, 145, 146} body size, ^{73, 146, 152-155} and ethnicity ^{73, 150, 151, 213} all shown to affect reporting.

To address these constraints there is interest in image-assisted dietary assessment. ^{205, 209, 214, 231,} ²⁸² Handheld devices were first used to capture images of foods manually but development of automated wearable cameras has allowed first-person point-of-view image capture to be explored. ^{212, 213, 230, 245} Wearable cameras provide a new opportunity to improve the accuracy of dietary assessment as the images/video provide a passive and objective record of an individual's eating episodes. ²⁰⁵

Initial estimates suggest wearable camera-assisted methods may have a relatively small measurement error for energy intake (EI) (7%), ²¹² and increase self-reported EI of traditional methods by 10-18%. ^{213, 230} The increased EI results from detection of unreported foods, changes to reported portion size, and other misreporting errors identified within the images. ²¹³ However, previous studies relied on the participant to manually capture images, ²¹² had small sample sizes, ^{212, 213} and none have compared wearable-camera assisted dietary assessment against a criterion measure, such as doubly labelled water. ^{213, 230}

In the present study we used doubly labelled water (DLW) to validate a wearable camera-assisted 24h recall. Analyses compared reported EI from 24h dietary recalls alone and dietary recalls plus the wearable camera (SenseCam) compared with total energy expenditure (TEE), estimated using DLW. A secondary objective was to examine the mechanism by which camera images enhance dietary recalls by quantifying alterations in self-report after viewing the images.

7.5. Subjects and methods

Forty volunteers, aged 18-64yrs, (20 male, 20 female) from the greater Auckland city metropolitan area, New Zealand were recruited by advertisements on community notice boards located at 15 supermarkets and three university campuses, and a campaign through a participant recruitment service website (www.researchstudies.co.nz). The recruitment service sent email announcements to people who had previously indicated interest in human research, and used paid advertisement

campaigns on Facebook. Potential participants who indicated interest were phoned, assessed for eligibility, and provided with written and verbal information regarding study procedures. Written informed consent was obtained before commencement of study.

Participants were informed the purpose of the research was to evaluate the use of Passive Image Capture to Record Everyday Events (PICTURE study) using wearable cameras. There was no specific reference to a validation study but participants were informed the images would be used to help assess their dietary intake and other health behaviours, such as the time spent watching television or travelling to work (not reported here).

All eligible participants were in self-reported good health, were not actively pursuing weight loss, and did not plan to conduct additional physical activity (above normal) or travel during the study period. Pregnant and lactating females or people with a recent acute illness, blood tests or intravenous fluids two weeks prior to the study scheduled study period were excluded. Recruitment efforts were targeted to all adults 18 to 65 years of age, but the participants contacted ensured an equal number of male and female participants were selected. The University of Auckland Human Participants Ethics Committee approved the research (ref 8701). An \$80 NZD gift card was given to participants at the conclusion of the testing period to compensate them for their time.

7.5.1 Study design

The study was a cross-sectional repeated measures design. Each participant took part in the study for a 15-day period in free living conditions, wore a wearable camera on four days, and data were collected at four appointments; a baseline assessment and three follow-up dietary assessments. Data collection was between March and September 2013.

Participants were contacted 1-3 days prior to their scheduled baseline assessment to reaffirm the study procedures and schedule the nutrition assessments. Participants fasted overnight (≥10h) and were instructed to refrain from any strenuous activity the day before the baseline assessment. Anthropometric characteristics, weight in light clothing with shoes and jewellery removed (±0.05kg, BWB-620, Wedderburn, Australia), height (±0.1cm, Secca, London, United Kingdom), and percentage body fat (±5 ohm, Imp DF50, ImpediMed, Pinkenba, Australia) were assessed.

Resting energy expenditure (REE) was measured before participants were given a weight-specific dose of DLW to determine TEE during the study period. Participants then wore a wearable camera (SenseCam) for four days; one familiarisation day, and each day prior to three interviewer-administered multiple pass 24-hour dietary recalls (MP24) conducted between days 2-4 (nutrition assessment 1), days 8-10 (nutrition assessment 2), and days 13-15 (nutrition assessment 3). Therefore, there was a minimum of three days and a potential maximum of eight days between nutrition assessments.

Five timed urine samples were collected for the determination of TEE at baseline, 5hr post dose, and days 3, 9 and 15. Participants were instructed to collect at least 50 mL of the second void of the day, to place and seal the sample in a specimen pot and record the time-of-void on a form provided. The samples were collected from participants at the earliest opportunity, either at the next nutrition assessment or from their home (if diet was assessed before collection). To assess any weight change during the testing period, body mass was reassessed on day 15 using the same scales. During the study period participants were told to follow their usual daily routines. A basic instruction leaflet for the SenseCam and a timeline of scheduled study assessments were provided at the conclusion of the baseline assessment. Standardised text messages were used to remind participants of the study protocol (e.g. Day 3: please remember to collect urine sample #3 (not the first void of the day) PICTURE STUDY). Messages were sent at times to align with the participant's reported usual daily schedule.

7.5.2 Use of wearable camera

The SenseCam is a wearable camera worn around the neck on a lanyard with a wide-angled lens. ^{243, 244} Sensors detect movement (accelerometer), heat (infrared), and light to approximately trigger image capture approximately every 20s (~2000-3000 images per day). Internal flash memory is sufficient for one week and battery capacity is adequate for a typical 12-16h day. Once turned on, the SenseCam operates continuously until the camera is switched off; or a privacy button can be activated to cease image capture temporally (7min). Participants were provided instructions and demonstrated they could operate and wear the device correctly at the baseline assessment. On the recording days participants were instructed to wear the device from when they woke (after bathing and dressing) until bedtime but could remove the camera anytime they felt uncomfortable or in locations where photography was inappropriate (e.g. gymnasium or public

restroom). An information sheet was provided to help participants determine when the camera should be switched off or not worn. The images were encrypted to ensure participants could not view the images before the nutrition assessment or if lost could not be viewed by third parties. After the completion of each MP24 participants were provided with a chance to screen the images privately and instructed to delete any image they did not wish to disclose before a joint image review with the researcher.²⁵¹

7.5.3 Energy intake assessment

MP24: The dietary recalls were conducted by a trained dietitian (LG) using a pen and paper based multiple pass method with a forgotten foods list to probe for unreported foods, adopted from the United States Department of Agriculture ²⁶ followed by an image review, see **Figure 13**. The assessments were conducted at the University of Auckland or participants' homes, or workplaces. Standard household measures, example crockery and glassware, and a portion size guide were used to assist participants to estimate portion sizes. The portion size guide used ²⁷³ was developed for the Australian population where the food supply is similar to New Zealand (no New Zealand specific portion size guide has been developed).

MP24+SC: After the final pass of MP24 the researcher (LG) used Doherty's wearable camera browser ²⁴⁴ to review the SenseCam images with the participant (after the images were screened privately by the participant). The participants were instructed to confirm, modify, add, or remove food items present in the images (MP24+SC). To assist the process the researcher simply restated the foods and portions self-reported in the MP24, but did not suggest changes or scrutinise self-reported intakes and queried about any unreported food items present in the images. All changes made by the participants were detailed to determine the frequency and impact on El. All unreported foods, misreporting errors, and alterations to portion size grouped by the following food categories: breads and cereals; beverages (excluding water due to no energy content); fruit and vegetables; meats & fish dairy products; snack foods (biscuits, sweets, and other snack foods); condiments (spreads, sauces, dips and dressings); alcohol; and other. The changes to self-report were both individual foods and composite foods (some foods cannot be separated into individual components).



Figure 13 Procedure to estimate energy intake with the wearable camera
7.5.4 Measurement of total energy expenditure

Daily TEE was measured using the DLW method. At the baseline assessment participants ingested a pre-mixed dose of ~0.1g 99.9% ²H₂O/kg total body water (TBW) and 2g 10% H₂¹⁸O/kg TBW. To ensure the full dose was consumed the dose bottle was rinsed three times with tap water followed by an additional mouth rinse. To ascertain background isotope levels a baseline urine sample was collected before dosing. Participants collected timed urine samples (5h post-dose, and on days 3, 9 and 15) were frozen (in duplicate in glass bottles) until study completion and analysed using an elemental analyser (Thermo, USA) coupled with a DeltaV isotope ratio mass spectrometer Thermo, USA). Daily TEE was calculated by the multipoint method by linear regression from the difference between elimination constants of ¹⁸O and ²H with individual respiratory quotient determined using the mean form the three MP24+SC dietary recalls ¹³³.

7.5.5 Resting energy expenditure

A standardised protocol was used to assess REE between 07:00 and 10:00. The TEE divided by REE was used to calculate the participants' activity factor during the study period. Participants were instructed to fast overnight (≥10h) and refrain from any strenuous activity the day before the assessment. Before commencement participants were positioned (near supine) on a folding bed for ≥10min asked to relax but remain awake. Testing was conducted in an environmental chamber maintained at 22°C with lights turned off during all assessments (participants remained awake throughout). REE was measured by indirect calorimetry using mouth piece and nose clip analysed breath by breath with a Moxus Modular system (S-3A/I Oxygen Analyser, CD-3A Carbon Dioxide Analyser, and KTC3 Turbine Volumetric System, AEI technologies, Pittsburgh, USA). A 30min measurement protocol was followed with data from the first 10min and final 2min omitted, along with periods of movement by the participant. Before each assessment the metabolic cart was calibrated with standard gas mixtures and a volumetric syringe. The activity factor was calculated by dividing TEE by REE.



7.5.6 Statistical analysis

All participants' data were included in the final analysis as device non-compliance and technical issues best reflect free-living conditions. Dietary intake was analysed using the nutrient analysis software FoodWorks 7 Professional edition (Xyris Software, Australia). Data analysis was performed using the software package for statistical analysis SPSS Statistics (V 20-0, IBM, USA). Paired t-tests were used to compare differences in self-reported EI between the MP24 and MP24+SC and EIs versus TEE. Limits of agreement between the EIs of the MP24 and MP24+SC and TEE were assessed according to the recommendations of Bland and Altman ²⁸³. Alterations in self-report (MP24+SC) and participant characteristics were described with summary statistics. Statistical significance was set at $\alpha \leq 0.05$.

7.6. Results

All participants completed the study procedures. General characteristics of the study population are presented in **Table 11**. The participants were predominantly New Zealand Europeans and the majority had tertiary education, but differences in age, body size and education between the male and female samples were apparent. Men were predominately overweight or obese (65%) and approximately eight years older than women who were generally of normal body weight (75%). Moreover a greater proportion of women had attended university or completed graduate degrees (80% versus 65%). Mean body weight did not differ significantly between day 0 and day 15 for men or women (difference = -0.4kg SD 0.2 kg, P=0.296 and 1.1 kg SD 0.7 kg, P=0.400, respectively).

		Ма	ale	Fem	ale
		Mean	SD	Mean	SD
Age (years)		34.8	12.6	27.1	7.5
Weight (kg)		86-4	14.7	61.3	9.7
Height (m)		178-0	6.1	165-3	6-9
BMI (kg/m²)		27.2	3.9	22.3	2.3
		n	%	п	%
Body Size					
Normal		7	35	17	85
Overweight		9	45	3	15
Obese		4	20	0	0
Ethnicity					
NZ European		14	70	15	75
Maori		2	10	1	5
Asian		4	20	4	20
Education					
High school or less		7	35	4	20
University diploma	or	10	50	11	55
undergraduate degree					
Postgraduate degree		3	15	5	25

Table 11 Demographic characteristics of the study sample.

TEE measures and EI data for all 40 participants were used. Three participants did not wear the camera for one of three recording days (two participants indicated they were in bed for most of the day, and one participant was noncompliant for one of the three recording days) and six device malfunctions occurred which resulted in failure to capture images. Additionally on two occasions participants forgot to bring the camera to the assessment. However, the MP24 was still conducted as usual and thus the MP24 and MP24+SC simply had the same values for these 11 instances (9% of all dietary recalls). Regarding the image screening the proportion of participants that chose not to screen the images privately increased at each successive nutrition assessment (n=13, n=18, and n=25 respectively). When images were screened the median time was 5.5 min (range = 1.0 to 17.1 min, IQR = 5.2 min).

For both male and females the 24h dietary recalls were distributed across all days of the week, but the proportion was lower on weekend days, shown in **Table 12**. Mean TEE measured by DLW and reported EIs assessed from the three 24h recalls (MP24 and MP24+SC) are presented in **Table 13**. For men the reported EIs were 17% (MP24) and 9% (MP24+SC) below the measured TEE. For women the mean reported EIs were 13% (MP24) and 7% (MP24+SC) below TEE. For men raw correlations between TEE and EI in the MP24 and MP24+SC were 0.68 and 0.61, respectively. In women correlations were 0.82 and 0.81, respectively. An assessment of agreement revealed EI does not influence the magnitude of measurement error (see online

supplementary material, **Figure 16** and **Figure 17**). The reduced magnitude of under-reporting of EI is presented in **Table 14**. The assistance of the wearable camera (MP24+SC) significantly reduced under-reporting for both men and women compared to the MP24 alone in all three dietary recalls, and there was no significant difference observed between TEE and EI of the MP24+SC in nutrition assessment two.

	Male	Female
	%	%
Monday	18	12
Tuesday	15	13
Wednesday	18	22
Thursday	12	22
Friday	15	12
Saturday	10	8
Sunday	12	12

Table 12 Distribution by day of the week for the dietary recalls collected for the total sample.

Each participant (n=40) had three multiple pass dietary recalls obtained over a two week period.

The increased EI associated with MP24+SC is presented in **Figure 15**. There was no relationship evident between the increase in EI and TEE, but viewing the images sometimes resulted in a decreased EI. Alterations in participants self-report are summarised in **Table 14**. The increase in reported EI was predominantly due to the addition of 265 unreported foods. Portion size was increased in most instances (49/51) but overall this had less impact on reported EI than misreported foods that were removed or exchanged during the image review.

One male and three females were excluded for REE data as the participants did not achieve a rested state during the REE procedures. The REE and activity factor for men and women were 7807 \pm 2125 kJ, activity factor 1.9 \pm 0.5, and 6548 \pm 2033 kJ activity factor 1.8 \pm 0.6 respectively.

			M	ale (n=20)					Fema	le (n=20)		
	El (kJ/	'day)	TEE (k	J/day)	TEE - EI		El (kJ/d	day)	TEE (kJ	/day)	TEE - EI	
	mean	SD	mean	SD	%	Р	Mean	SD	TEE	SD	%	Р
Total												
MP24	12 004	2122	44 405	0000	-17	<0.001	9420	1694	40.044	4000	-13	<0.001
MP24+SC	13 196 [*]	2529	14 400	2032	-9	0.02	10091 [*]	1672	10 641	1039	-7	0.004
Recall 1												
MP24	11 770	3564	14 405	2622	-19	0.003	9253	1501	10 0 / 1	1620	-15	<0.001
MP24+SC	12 543 [*]	3941	14 400	2032	-13	0.02	9975 [*]	1674	10 641	1039	-8	0.02
Recall 2												
MP24	12 769	3183	14 405	2622	-12	0.008	9805	2699	10 0 / 1	1620	-10	0.013
MP24+SC	14 411 [*]	3417	14 400	2032	-1	0.923	10455 [*]	2621	10 041	1039	-4	0.319
Recall 3												
MP24	11 472	3447	14 405	2622	-21	0.001	9202	2068	10 0 / 1	1620	-15	<0.001
MP24+SC	12 634 [*]	3331	14 400	2032	-13	0.025	9843	1983	10 641	1039	-9	0.218

Table 13 Daily energy intake (EI) measured in three Multiple Pass 24h dietary recalls (MP24 and MP24+SenseCam) and total energy expenditure (TEE) measured with the doubly labelled water technique.

SC = SenseCam, * MP24+SC significantly different to MP24 P≤0.05

Table 14 Alterations in energy intake for all participants viewing wearable camera images after completion of the multiple pass dietary recalls (MP24+SenseCam).

	Unreported foods					▲ Portion Size				Misreported foods			
		En€	ergy (kJ)			Energy (kJ)				Energy (kJ)			
	n	Mean	SD	Total	n	Mean	SD	Total	n	Mean	SD	Total	
Breads/ cereals	23	462	162	10 634	16	462	306	7386	3	-607	191	-1821	
Beverages	40	308	182	12 334	6	26	80	155	13	-135	149	-1754	
Fruit/ vegetables	47	153	98	7187	10	43	56	434	8	-130	107	-1037	
Meat /fish/eggs	11	590	140	6494	3	88	22	264	2	-402	211	-804	
Dairy	18	616	181	11 090	2	333	78	667	3	-113	56	-340	
Snack foods	64	571	425	36 547	3	1140	411	3419	3	-1112	331	-3335	
Condiments	50	307	270	15 343	7	181	35	1266	1	-184	-	-184	
Alcohol	7	712	130	4983	1	128	-	128	0	-	-	-	
Other	5	1191	203	5957	3	-30	105	-91	3	-2526	793	-7577	
Total	265	417	222	110 570	51	267	166	13 628	36	-468	311	-16 852	



Figure 14 Under-reported energy intake in the 24h dietary recall alone (MP24) and with the assistance of the wearable camera (MP24+SC) compared to total energy expenditure (TEE) for men and women. *Significantly different to TEE P ≤ 0.05 .



Figure 15 Change to self-reported energy intake in the 24h dietary recalls (MP24) after viewing the wearable camera images (MP24+SenseCam), compared to total energy expenditure (TEE) for men and women (n=120), three 24h dietary recalls for each participant).



Figure 16 Agreement between mean self-reported energy intake (MP24+SenseCam) and TEE for men (n=20).



Figure 17 Agreement between mean self-reported energy intake (MP24+SenseCam) and TEE for women (n=20).

7.7. Discussion

The present study validated a wearable camera-assisted dietary recall against DLW and examined the impact of wearable cameras on dietary underreporting. Overall the addition of wearable cameras significantly reduced the magnitude of dietary energy under-reporting by 9% in men and 6% in women compared to dietary recall alone. The findings confirm preliminary research which suggested wearable cameras may reduce measurement error by revealing unreported foods and misreporting errors. ^{213, 272}

The degree of under-reporting dietary EI observed in the MP24 alone (male 17%, female 13%) was similar to that reported in other 24h recall DLW validation studies, which have reported EI values approximately 8-24% below TEE. ⁸ When compared with the largest DLW study (n=524) to date, which validated the Automated Multiple Pass Method (AMPM), the level of dietary EI underreporting bias was comparable for males (14% among overweight men and 20% among obese men compared to 17% in the present study) but higher for females (6% among women with normal body weight compared to 13% in the present study). ²⁶ The lower value observed in the AMPM validation was likely attributable to the structured AMPM software (not pen and paper), and robust trial conditions used to replicate the procedures of the NHANES survey ²⁶ in conjunction with a motivated sample, which received substantial financial incentives.

There have been few DLW validations of image-assisted dietary assessment methods. A custom wearable camera "eButton" designed to objectively assess dietary intake and physical activity has been demonstrated, but is yet to be validated. ^{205, 214} Other image-assisted methods in development differ as they require participants to actively capture images using smart phones or other handheld devices. Nonetheless validation of the Remote Photography Food Method (a manually triggered image-based dietary record) against DLW among free-living adults (predominantly overweight and obese) revealed reduced measurement error compared to traditional methods (mean EI 6% below TEE compared to 12-49% among overweight and obese populations reported elsewhere). ^{26, 152-154, 209, 216, 218, 219} Similar image-based dietary records in development are yet to be validated with human participants. ^{231, 282}

Several factors may explain why under-reporting was not completely eliminated with the assistance of the wearable camera. The wearable cameras imaging frequency was insufficient (2~3 images per minute) to capture all foods consumed, and image quality was relatively poor, List of research project topics and materials

especially in low light environments. Moreover the position of the camera on the body allows the lens angle to be affected by posture, and foods in bowls or on high tables can be obscured easily. Intentional under-reporting may also go undetected if participants delete images of foods during the private screening. However, the short duration participants took to screen the images and the proportion of participants that chose not to screen their images in second and third nutrition assessment, suggested this was not a frequent issue. Reactivity could also be a factor. Previous studies have indicated using wearable cameras are a low burden but may impact on participants' usual dietary behaviours; however, the degree of behaviour change was unclear. ^{212, 213} Additional dietary recalls for non-camera days would have provided a within-person comparison to assess reactivity.

Other study limitations include the relatively small heterogeneous sample that was not representative of the general population; therefore the study may have produced different results. Moreover, due to differences in body size male and female data were treated separately. However, participants of all body sizes were deliberately recruited as the earlier feasibility study revealed camera-assisted recalls were helpful for all types of people ²¹³. Additionally, the prevalence of under-reporting increased substantially in New Zealand's most recent adult nutrition survey among people with normal body weight, as well as overweight and obese ⁹⁷. Furthermore, a sole dietitian conducted the nutrition assessments and image review process, thus interviewer bias cannot be ruled out (the interview procedures were not audited). Additionally, the dietary intake data may not be representative of usual intake, due to the short duration of the study, limited number of dietary recalls conducted, and the lower proportion of weekend days versus weekdays. ⁹⁵

A unique feature of the study was the detailed comparison between traditional self-report alone versus self-report assisted by the wearable camera. This study design allowed determinants of under-reporting to be identified, and confirmed unreported snack foods, condiments and beverages are a primary source of under-reported EI in a free-living setting. An interesting finding in both the feasibility ²¹³ and current study was the frequency that fruit and vegetables were unreported (47/265 foods). Often these were during snacks, such as bananas, apples, carrots, and raisins, and further highlights the difficulty participants have remembering snacking episodes during retrospective dietary assessments. The alterations by participants to portion size may have produced correlated errors (as the true portion size remained unknown), ^{119, 276, 277} but in most

instances the portion size was clearly incorrect (e.g. two vs. one slice of toast). The decision not to question or scrutinise the participants self-report during the image-review reduced the potential for interviewer bias ⁹⁶ but trained image analysts or use of automated image-analysis techniques could enhance the method further. ^{209, 240}

The strength of automated wearable cameras over handheld devices is their ability to capture images passively, which means they are potentially less intrusive during daily activities and may reduce participant burden. Wearable technologies can also collect physical activity data passively (using inbuilt accelerometers and GPS), a key lifestyle consideration often overlooked when collecting dietary intake data. ^{144, 185} Thus wearable technologies have potential greater than simply revealing under-reported foods and misreporting errors, which needs to be explored.

7.8. Conclusions

The wearable camera significantly reduced the magnitude of under-reporting in the 24h dietary recall by 9% in men and 6% in women, as the images revealed unreported foods and misreporting errors not captured by the traditional method alone. Wearable cameras with faster imaging frequencies, high definition image sensors, and the use automated image analysis techniques may enhance the method further. Additional research is needed in larger representative samples of the population. Future studies should explore the use of wearable cameras in different settings using a variety of image-assisted methods.

Acknowledgments

The authors responsibilities were as follows: LG, CN, ER, RM, AD, and NG: study design; LG and CNM: data collection and study management; LG, ER, RM, and NG: energy expenditure analyses; LG, CN, ER, RM, AD, NG, and JU: data interpretation; LG: statistical data analysis; and LG wrote the manuscript; CN, ER, RM, AD, NG, and JU edited the manuscript. None of the authors had a personal or financial conflict of interest.

Funding

There was no external funding source for this manuscript.

CHAPTER 8. THE USE OF A WEARABLE CAMERA TO CAPTURE AND CATEGORISE THE ENVIROMENTAL AND SOCIAL CONTEXT OF SELF-IDENTIFIED EATING EPISODES

8.1. Introduction to publication

Wearable cameras provide a unique opportunity, which was previously not feasible, to passively collect objective data on behaviours in free-living settings. Consequently, these new data may provide additional useful information regarding the populations' dietary intake and dietary behaviours compared to the data currently collected. This chapter (Chapter Eight) is a reformatted version of the manuscript entitled "The use of a wearable camera to capture and categorise the environmental and social context of self-identified eating episodes" submitted to the journal "Appetite" on the 10th September 2014. The journal specialises in behavioural nutrition and the cultural, sensory, and physiological influences on choices and intakes of foods. The manuscript has been through peer-review, revised and resubmitted, and is currently awaiting final editorial decision. The most recent impact factor available was 2.520 (2013). The chapter presents results from a manual image analysis conducted on the dataset obtained in the PICTURE study reported in Chapter Seven. The study was undertaken to address Objective Five, which was to assess the utility of wearable cameras to objectively record and reliably assess environmental and social contexts of eating episodes.

8.2. Author contribution

Luke Gemming was involved in developing the research question, ethics application, study design, data collection and analysis. He also interpreted the findings, and wrote the paper for publication.

THE USE OF A WEARABLE CAMERA TO CAPTURE AND CATEGORISE THE ENVIROMENTAL AND SOCIAL CONTEXT OF SELF-IDENTIFIED EATING EPISODES

^aLuke Gemming MSc

^bAiden Doherty PhD

^cJennifer Utter PhD

^aEmma Shields BSc

^aCliona Ni Mhurchu PhD

^aPostal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

^bBritish Heart Foundation Health Promotion Research Group, Nuffield Dept of Population Health,

Old Road Campus, University of Oxford, OX3 7LF, United Kingdom

^bSchool of Population Health, University of Auckland, Private Bag 92019, Auckland, 1142, New Zealand

Address for correspondence:

Luke Gemming

Postal Address: National Institute for Health Innovation, University of Auckland

Private Bag 92019, Auckland Mail Centre, Auckland, 1142, New Zealand

l.gemming@auckland.ac.nz

8.3. Abstract

Background: Research investigating the influence of the environmental and social factors on eating behaviours in free-living settings is limited. This study investigates the utility of using wearable camera images to assess the context of eating episodes.

Methods: Adult participants (N=40) wore a SenseCam wearable camera for 4 days (including 1 familiarisation day) over a 15-day period in free-living conditions, and had their diet assessed using three image-assisted multiple-pass 24-hour dietary recalls. The images of participants eating episodes were analysed and annotated according to their environmental and social contexts; including eating location, external environment (indoor/outdoor), physical position, social interaction, and viewing media screens.

Results: Data for 107 days were used, with a total of 742 eating episodes considered for annotation. Twenty nine percent (214/742) of the episodes could not be categorised due to absent images (12%, n=85), dark/blurry images (8%, n=58), camera not worn (7%, n=54) and for mixed reasons (2%, n=17). Most eating episodes were at home (59%) and indoors (91%). Meals at food retailers were 24.8 minutes longer (95% CI: 13.4 to 36.2) and were higher in energy (mean difference = 1196kJ 95% CI: 242, 2149) than at home. Most episodes were seated at tables (27%) or sofas (26%), but eating standing (19%) or at desks (18%) were common. Social interaction was evident for 45% of episodes and media screens were viewed during 55% of episodes. Meals at home watching television were 3.1 minutes longer (95% CI: -0.6 to 6.7) and higher in energy intake than when no screen was viewed (543kJ 95% CI: -32 to 1120).

Conclusion: The environmental and social context that surrounds eating and dietary behaviours can be assessed using wearable camera images.

Keywords: SenseCam, nutrition assessment, context, eating behaviours

8.4. Introduction

Poor quality diets, excess energy intakes, and unhealthy energy dense foods are associated with increased morbidity and mortality from diet-related non-communicable diseases. ^{4, 5, 54} Understanding the determinants of eating behaviours, including the environmental and social context in which they occur, is important to inform nutrition policies and interventions to positively change these behaviours. ⁴⁰⁻⁴⁴ Investigations on human eating and dietary behaviours have examined a range of sensory and internal physiological mechanisms relating to food intake, primarily within laboratory settings. ^{41, 42, 284-286} However, research exploring the influence of environmental and social factors on intake in free-living settings is needed, ^{35-39, 44, 286} as labbased research is not indicative of real-world eating behaviours. ²⁸⁷⁻²⁸⁹

Environmental and social factors that are associated with or affect food intake include eating location (eating at home, versus out-of-home), physical position (seated eating at dining tables versus sofa or work desk), social-interaction (eating alone or in a social situation), presence of media screens (eating viewing television and other media screens), and other ambient factors (lighting, temperature, sounds and colours). ^{41, 286, 290}. For example viewing media screens is associated with increased energy intake (EI) in laboratory settings, ²⁹¹ but this may not reflect free-living conditions. However, if true, in which real-word contexts is the associated increase in EI present? Accurate measurement of the eating episodes and contexts surrounding food consumption in free-livings settings are therefore important. ^{286, 292}

The difficulty with undertaking research in free-living settings is the absence of an unobtrusive measure to record valid data on contexts and dietary intake. ^{36, 286, 293-295} Lab-based research, though stringent in design, is not always indicative of real-world eating behaviours. ^{42, 285-289} Direct observations in free-living settings are possible, but researchers can inadvertently become part of the context they are observing, ³⁸ and direct observations are not practical for large-scale research.

Wearable cameras may provide an unobtrusive approach to passively record contexts, as they capture images of events automatically from a first-person perspective. ^{243, 244} In a recent doubly labelled water validation study we demonstrated that wearable camera images can enhance self-report in dietary assessment by revealing unreported foods and misreporting errors not captured by traditional methods alone. ²⁹⁶ Therefore, we investigate the utility of wearable cameras to

objectively record and reliably assess environmental and social contexts of eating episodes, and compare the contexts EI and density using data obtained from wearable camera-assisted 24-hour dietary recalls in the validation study.²⁹⁶

8.5. Methods

The images and dietary EI data used in this study were taken from a 15-day study using doubly labelled water to validate the wearable camera image-assisted 24h dietary recall, conducted between March and September 2013²⁹⁶. The study recruited forty healthy adult participants (20 males; age 35, SD 17 yrs; BMI 27, SD 4; and 20 females; age 28, SD 7yrs; BMI 22, SD 2) from Auckland, New Zealand using an advertisement on a research studies website and flyers placed on community and university notice boards. This study was not designed or powered to detect differences between specific contexts during eating episodes, but rather to evaluate and demonstrate the potential of wearable cameras to passively record and assess contexts objectively. For this reason we limited our analysis to five contexts similar to those currently assessed in national dietary surveys, ^{24, 85} and only compared the episodes duration, EI and density of the different contexts. Figure 18 presents a flow diagram of the study procedures and the dimensions coded for each eating episode. Ethical approval for the study was granted by the University of Auckland Human Participants Ethics Committee on the 30th November 2012 (Ref 8170). All participants provided written informed consent and were given information on appropriate use of a wearable camera in a free-living setting. Participants were compensated with an \$80 (NZD) gift card for their time.

8.5.1 Data collection

SenseCam is a wearable camera worn around the neck on a lanyard with a wide angle lens that automatically captures first person point-of-view images approximately every 20 seconds. ²⁴³ The exact imaging frequency differs as images are captured in response to sensors that detect changes in movement, heat and light. ²⁴³ The battery life is approximately 16 hours and thus sufficient to capture a participant's typical day without turning the device off. This allows both planned and spontaneous eating episodes to be captured passively.

Participants wore the SenseCam four allocated days from wake until sleep over the two week period in a free-living setting (completing their usual daily activities). Participants were familiarised with the device for 1-day before wearing the camera each day prior to three interviewer-administered multiple pass 24-hour dietary recalls (MP24). The three nutrition assessments for each participant were conducted between days 2-4 (nutrition assessment 1), days 8-10 (nutrition assessment 2), and days 13-15 (nutrition assessment 3). During the dietary assessment, the SenseCam images were downloaded from the device into freely available wearable camera software. ²⁷⁴ Participants screened the images privately, and were instructed to delete any that they did not want to be viewed by the researchers. ²⁵¹ The remaining images were used by participants to enhance their self-reported dietary intake (MP24+SenseCam). The captured images are used in the present study to classify the context of the self-identified eating episodes.

8.5.2 Data processing

The eating episodes were first defined and labelled in the software. The images were viewed chronologically by the researchers and the eating episodes were numbered and categorised into six possible meal types using the self-reported 24h-hour recall data: (1) breakfast, (2) morning snacks, (3) lunch, (4) afternoon snacks, (5) dinner, and (6) evening snacks/supper. These categories are in-line with those used in the National Health and Nutrition Examination Surveys to describe eating occasions (excluding brunch).²⁴ The starts of episodes were defined using the first image of food present, not including food preparation or cooking. The ends of episodes were defined using the image following the last image where food was present, rather than using the last image where food was present, as wearable cameras are insensitive to sub-15 second changes. For episodes where foods were only partially consumed, the image following the last to depict eating (indicated by movement of the food or participant in relation to the food) was used. Once the episodes were defined in the browser a new screen appeared showing all the images for the given episode. To confirm the duration of the episodes were defined correctly, the researcher viewed images before and after episodes to ensure no additional foods were consumed; the eating episodes were verified by comparing the time stamps and foods visible within the images to those self-reported by participants during the three SenseCam-assisted 24h recalls.

8.5.3 Annotation protocol

Figure 18 shows the eating episodes were annotated on five dimensions (location, external environment, physical position, social interaction, and media screens) and included similar contexts as those self-reported in the National Diet and Nutrition Surveys of the United Kingdom, and those explored by other work using wearable cameras and personal digital assistants. 85, 297, ²⁹⁸ A final dimension was an optional comment field for episodes which could not be confidently annotated, which included (1) foods absent in images, (2) dark/blurry images, (3) camera not worn, and (4) for mixed reasons. In a similar manner to defining the episodes duration, the images before and after the defined eating episode provided visual clues to annotate on the context dimensions. Visual clues included approaching the outside of a food retailer or work location, viewing a menu with others while seated at a table, or walking to the sofa with the television turned on. Social interaction was primarily indicated by an active conversation with third parties. Other indicators of social interaction included body language/movements and eve contact of third parties. To determine the consistency among raters for classifying the five dimensions (location, external environment, physical position, social interaction, and media screens) a random sample of \geq 25% of the eating episodes was independently annotated by two researchers (LG, ES) and an inter-rater reliability analysis using a kappa statistic was performed.

8.5.4 Data analysis

The data from the eating episodes were stored in the SenseCam software's Microsoft SQL database. The data were extracted for further analysis to SPSS (IBM Corporation, Version 20, USA). The variables recorded for each episode included participant ID, eating episode, meal type, annotatable, reason why episode was not-annotatable, start time, end time, duration, number of images, EI, and energy density. The contextual dimensions coded were location, external environment, physical position, social interaction, and media screens. Summary statistics were generated for each meal type to investigate how many eating episodes could be classified. Summary statistics were then generated for the context dimensions of all eating episodes, main meals (breakfast, lunch and dinner) and during snacking episodes (morning snacks, afternoon snacks, evening snacks). Mean differences and 95% confidence intervals were used to compare

different contexts for duration EI and density. Unpaired t-tests assessed any differences in EI and

density between episodes that could be assessed with those that could not be assessed.



Figure 18 Flowchart of study procedures to capture and categorise the environmental and social contexts of eating episodes using wearable camera images. Contexts were assessed using five dimensions with dimension-specific sub categories. ^aParticipant actively engaged in social interaction during the eating episode. ^bParticipant in a social situation e.g. café but not engaged in social interaction.



8.6. Results

The mean (SD) wear time of the wearable cameras was 13h 0min (\pm 2h 23min) per day. Due to non-compliance wearing the camera (n=3), and technical failures that resulted in incorrect time stamps (n=4) and loss of images (n=6), data from 107/120 days were used in this analysis. **Table 15** provides details of the 742 self-reported eating episodes over the 107 days with associated images. The total time to define the eating episodes within the software, input the data for all variables, and annotate the episodes was 64h, an average of approximately 36 (\pm 14.2) min per day of dietary intake. The average time for the manual annotation of context attributes associated with any given episode was 1min 21s (\pm 44s). A random subset of 144 episodes (27%) was annotated independently by two researchers. The inter-rater reliability was 0.96 for location, 0.81 for indoor/outdoor, 0.84 for sitting/standing position, 0.68 for social interaction status, and 0.79 for the presence of media screens.

Table 15 Eating episodes identified by 40 participants during three wearable camera image-assisted 24h dietary recalls.

Meal	Number episodes	Mean energy intake	Mean energy density
		(kJ ± SD)	(kJ/g ± SD)
Breakfast	115	2013 ± 1496	4.9 ± 3.5
Morning snack/s	123	752 ± 943	4.8 ± 5.7
Lunch	87	2681 ± 1654	6.1 ± 3.3
Afternoon snack/s	201	988 ± 1156	7.4 ± 7.2
Dinner	103	3506 ± 1832	5.6 ± 2.6
Evening Snack/s	113	1071 ± 1095	7.3 ± 7.1
All eating episodes	742	1668 ± 1665	6.2 ± 5.7

Effectiveness of wearable cameras to capture eating episodes for analysis

From the 742 self-reported eating episodes, 29% (214 episodes) could not have their context categorised due to absent images (12%, n=85), dark/blurry images (8%, n=58), camera not worn (7%, n=54) and for mixed reasons (2%, n=17). Of the 71% of episodes (n = 528) that could be annotated, 35,299 images were captured, with one image being captured every 13.0s \pm 3.2s.

Table 16 compares the EI and density of eating episodes that could not have their context annotated (n = 528) with those that could not be annotated (n = 214). There were no significant differences observed in EI or density for specific meals that were coded compared to those not coded. The majority of episodes that could not be annotated were snacks (169/214), which had a similar energy content overall (mean difference = 44kJ 95% CI: -167 to 262, P=0.664) but were less energy dense (-1.8kJ/g 95% CI: -3.2 to -0.4, P=0.010).

Table 16 A comparison of eating episodes that could be annotated for context using SenseCam images (n=528) compared to those that could not (n=214).

	Energy intake (kJ ± SD) Energy der			Energy density (nsity (kJ/g ± SD)			
			Mean				Mean	
			Difference				Difference	
	Not –		kJ	P-	Not –		kJ/g	
	Annotatable	Annotatable	(95% CI)	Value	Annotatable	Annotatable	(95% CI)	P-Value
Breakfast	1528 ± 1618	2141	613 (105 to	0.404	4.4 ± 4.2	5.0 ± 3.4	0.6	0.505
Breaklast	(n = 24)	± 1445 (n = 91)	(-125 to 1352)	0.101	(n = 24)	(n = 91)	(-1.2 to 2.5)	0.535
Morning	755 ± 1184	751 ± 845	-4	0.000	4.7 ± 6.4	4.9 ± 5.5	0.2	0.050
snack/s	(n = 33)	(n = 90)	(-456 to 448)	0.986	(n = 33)	(n = 90)	(-2.3 to 2.7)	0.859
Lunch	2638 ± 2694 (n = 6)	2684 ± 1577 (n=81)	46 (-2774 to 2867)	0.968	7.8 ± 5.6 (n = 6)	6.0 ± 3.1 (n = 81)	-1.8 (-7.6 to 4.1)	0.467
Afternoon	857 ± 1064	1057 ± 1200	200	0.007	9.5 ± 8.0	6.3 ± 6.5	-3.2	0.05
snack/s	(n = 70)	(n = 131)	(-125 to 525)	0.227	(n = 70)	(n = 131)	(-5.4 to -1.0	0.05
Dinner	4341 ± 2079 (n = 15)	3363 ± 1760 (n = 88)	-987 (-2174 to 219)	0.103	5.2 ± 2.6 (n = 15)	5.6 ± 2.6 (n = 88)	0.7 (-1.1 to 1.9)	0.550
Evening	1053 ± 1219	1097 ± 905	43	0.000	7.5 ± 7.4	7.0 ± 6.7	-0.5	0 747
snack/s	(n = 66)	(n = 47)	(-352 to 439)	0.829	(n = 66)	(n = 47)	(-3.1 to 2.2)	0.717
Total	1271 ± 1614 (n = 214)	1829 ± 1646 (n = 528)	558 (299 to 816)	<0.001	7.2 ± 7.1 (n = 214)	5.7 ± 4.9 (n = 528)	-1.5 (-4.3 to -2.5)	0.006

Duration, energy intake and density of eating episodes by contexts

Table 17 shows the 528 eating episodes classified for context and their associated duration, EI, and density. On average the duration of eating episodes was 14.7 (±16.8) minutes. Snacking episodes were 3.9 minutes shorter on average (95% CI: -6.8 to -1.0) than meals, and contained approximately two thirds less energy (-1762kJ, 95% CI: -2002 to -1523) and were less energy dense (-0.4kJ/g 95% CI: -1.3 to -0.5).

Location and Environment

With respect to location, meals at home were 5.3 minutes longer (95% CI: 8.2 to 9.8) than meals in occupational settings (82% were lunch), but had a similar energy content and density. Snacks

at home were 4.4 minutes shorter (95% CI: -9.2 to -0.3) than snacks in occupational settings, but were higher in energy (367kJ 95% CI: 131 to 602) and more energy dense (2.5kJ/g, 95% CI: 0.8 to 4.2). Few episodes (6%) were in food retailers (restaurants/cafés/bars/fast food), but meals were 24.8 minutes longer (95% CI: 13.4 to 36.2) than meals at home, were higher in energy (1196kJ 95% CI: 242 to 2149), but had a similar energy density. Regarding other environmental factors, eating episodes indoors were 8.2 minutes longer (95% CI -2.4 to -18.9) than episodes outdoors, but had a similar EI and density. There were few (3%) episodes in vehicles or mixed environments (e.g. walking in park, drinking a coffee before catching a bus, and continuing to drink the coffee).

Physical Position

Regarding physical positions, meals consumed while seated at tables were 7.3 minutes longer (95% CI: 2.7 to 11.8) than meals in other positions, had a higher EI (938kJ 95% CI: 512 to 1365), but were less energy dense (0.9kJ 95% CI: -1.57 to -0.19). When compared directly to meals consumed while seated on sofas, meals at tables were 7.0 minutes longer (95% CI: 2.6 to 11.5), had more energy (462kJ, 95% CI:-29.8 to 955), and a similar energy density. The majority of eating episodes while standing/active were snacks (73%), which were 9.8 minutes shorter (95% CI: -12.8 to -6.8) than when seated, with a similar EI, but were more energy dense (2.9kJ/g 95% CI: 1.7 to 4.8).

Social Interaction

With respect to social context, meals with social interaction were 11.4 minutes longer (95% CI: 7.4 to 15.5) than meals with no social interaction, had more energy (684kJ 95% CI: 279 to 1088), but a similar energy density. Snacks with social interaction were 4.2 minutes longer (95%, CI: 0.0 to 8.6) than snacks with no social interaction, but with a similar EI and density. Few episodes (11%) were in social situations with no interaction.

Media Screens

Regarding screens, both meals and snacks consumed while viewing any type of screen were 4.3 and 2.7 minutes shorter (95% CI: 0.1 to 8.4 and -6.8 to 1.38, respectively) than when no screen was viewed, with EIs similar for both types of meals. Snacks consumed while viewing screens also had a greater energy density (1.8kJ/g 95% CI: 2.9 to 3.4) compared to when no screen was

viewed. Viewing media screens during eating episodes only within the home were also examined (see **Table 18**). Meals watching television at home were 3.1 minutes longer (95% CI: -0.6 to 6.7) than meals not viewing screens and were higher in energy (543kJ 95% CI: -32 to 1120). Snacks viewing television were a similar duration compared to snacks when no television was viewed, but were higher in EI (514kJ 95% CI: 47 to 1077). Computers were the most common screen viewed during snacks at home (31%) but were similar in duration, EI and density to when no screen was viewed.

Table 17 Meal and snack eating episodes (N=528) from 40 participants by location, environmental context, positional context and social context.

Context	Moole	Snacks	All eating
Context	Media	Olideks	episodes
LOCATION			
At home	n=165 (63%)	n=147 (55%)	n=312 (59%)
Energy intake (kJ± SD)	2665 ± 1562	992 ± 1002	1877 ± 1567
Energy density (kJ/g ± SD)	5.5 ± 3.0	6.5 ± 6.5	6.0 ± 5.0
duration (min \pm SD)	11.3 ± 9.7	10.5 ± 14.5	11.7 ± 12.2
Occupation	n=51 (20%)	n=68 (25%)	n=119 (23%)
Energy intake (kJ± SD)	2550 ± 1333	625 ± 711	1450 ± 1398
Energy density (kJ/g ± SD)	5.4 ± 2.8	4.0 ± 5.5	4.6 ± 4.6
duration (min \pm SD)	18.0 ±15.1	15.0 ± 17.2	16.2 ± 16.3
Restaurant/Bar/Café	n=21 (8.1%)	n=13 (4.9%)	n=34 (6%)
Energy intake (kJ±SD)	3861 ± 2038	1714 ± 2164	3040 ± 2311
Energy density (kJ/g ± SD)	4.7 ± 1.9	3.8 ± 2.6	4.4 ± 2.2
duration (min \pm SD)	37.5 ± 24.8	31.4 ± 28.1	35.2 ± 25.9
Other	n=23 (8.8%)	n=40 (14.9%)	n=63 (12%)
Energy intake (kJ±SD)	2492 ± 2346	1175 ± 1028	1656 ± 1739
Energy density (kJ/g ± SD)	6.7 ± 4.5	8.0 ± 6.5	7.5 ± 5.8
duration (min ± SD)	23.6 ± 32.0	11.1 ± 13.4	15.7 ± 22.7
ENVIRONMENT	Meals	Snacks	All eating episodes
Indoor	n=243 (93%)	n=236 (88%)	n=479 (91%)
Energy intake (kJ±SD)	2760 ± 1683	904 ± 1001	1846 ± 1670
Energy density (kJ/g ± SD)	5.3 ± 2.8	5.9 ± 6.2	5.6 ± 4.8
duration (min \pm SD)	16.2 ± 15.1	23.2 ± 30.6	14.1 ± 15.0
Outdoor	n=12 (4.6%)	n=18 (6.7)	n=30 (5.7%)
Energy intake (kJ± SD)	2094 ± 1524	1231 ± 1040	1576 ± 1303
Energy density (kJ/g ± SD)	8.4 ± 5.5	6.3 ± 5.9	7.2 ± 5.8
duration (min ± SD)	20.9 ± 25.7	11.9 ± 14.7	22.3 ± 28.3
In Vehicle	n=3 (0.6%)	n=10 (3.7%)	n=13 (2.5%)

Contoxt	Moolo	Speeke	All eating
Context	Weals	Shacks	episodes
Energy intake (kJ±SD)	1674 ± 893	1511 ± 1508	1549 ± 1358
Energy density (kJ/g ± SD)	6.6 ± 3.1	7.8 ± 7.9	7.5 ± 7.0
duration (min ± SD)	7.7 ± 5.7	10.0 ± 10.1	10.2 ± 6.0
Mixed	n=2 (0.8%)	n=4 (1.5%)	n=6 (1.4%)
Energy intake (kJ±SD)	3655 ± 826	1754 ± 2027	2388 ± 1888
Energy density (kJ/g ± SD)	7.8 ± 0.9	3.1 ± 1.8	4.6 ± 2.8
duration (min \pm SD)	68.6 ± 79.2	26.0 ± 28.2	40.2 ± 47.0

PHYSICAL POSITION

Sitting at table	n=100 (38%)	n=41 (15%)	n=141 (27%)
Energy intake (kJ±SD)	3301 ± 1826	1146 ± 1521	2762 ± 1932
Energy density $(kJ/g \pm SD)$	5.0 ± 2.2	4.8 ± 5.2	4.9 ± 3.3
duration (min \pm SD)	21.2 ± 20.2	24.7 ± 30.2	22.2 ± 23.5
Sitting on sofa	n=81 (31%)	n=57 (21%)	n=138 (26%)
Energy intake (kJ+SD)	2839 + 1529	1089 + 965	2116 + 1578
Energy density $(kJ/q \pm SD)$	5.6 ± 2.6	5.4 ± 5.5	5.5 ± 4.0
duration (min \pm SD)	14.2 ± 9.3	11.2 ± 12.6	12.9 ± 10.8
Sitting at desk	n-24 (9 0%)	n-67 (23%)	n-95 (18%)
Energy intake (k I+ SD)	2206 + 1251	756 + 751	1244 ± 1289
	2290 £ 1231	130 ± 751	1244 ± 1209
Energy density $(KJ/g \pm SD)$	9 (24)	4.4 ± 0.1	4.7 ± 5.2
duration (min \pm SD)	5.3 ± 2.9	5.0 ± 5.7	15.5 ± 12.7
Standing/Active	n=28 (11%)	n=74 (28%)	n=102 (19%)
Energy intake (kJ±SD)	1296 ± 1093	829 ± 795	957 ± 905
Energy density (kJ/g ± SD)	7.8 ± 5.4	8.1 ± 7.1	8.0 ± 6.7
duration (min \pm SD)	10.6 ± 23.0	5.7 ± 7.0	7.2 ± 13.5
Other	n=18 (6.9%)	n=67 (25%)	n=52 (10%)
Energy intake (kJ± SD)	2004 ± 1001	958 ± 1058	1320 ± 1145
Energy density (kJ/g ± SD)	5.1 ± 2.9	6.2 ± 5.8	5.8 ± 5.0
duration (min \pm SD)	13.6 ± 11.7	12.0 ± 12.9	12.6 ± 12.4

SOCIAL	Meals	Snacks	All eating episodes
Social Interaction	n=124 (48%)	n=115 (43%)	n=239 (45%)
Energy intake (kJ±SD)	3081 ± 107	1066 ± 1210	2112 ± 1846
Energy density (kJ/g ± SD)	5.2 ± 2.7	5.6 ± 5.9	5.4 ± 4.5
duration (min ± SD)	22.7 ± 21.1	15.2 ± 20.8	19.1 ± 21.3
Social – no interaction	n=27 (10%)	n=33 (12%)	n=60 (11%)
Energy intake (kJ±SD)	2969 ± 1740	851 ± 819	1804 ± 1681

Contoxt	Moolo	Speake	All eating	
Context	Wears	Slideks	episodes	
Energy density (kJ/g ± SD)	5.5 ± 1.9	8.1 ± 7.5	6.9 ± 5.8	
duration (min ± SD)	22.7 ± 11.2	8.2 ± 9.3	9.6 ± 8.3	
Not Social	n=109 (42%)	n=120 (45%) (120)	n=229 (43%)	
Energy intake (kJ± SD)	2256 ± 1366	892 ± 934	1541 ± 1344	
Energy density (kJ/g ± SD)	5.9 ± 3.6	5.6 ± 6.1	5.7 ± 5.1	
duration (min \pm SD)	11.4 ± 9.5	11.7 ± 12.7	11.2 ± 6.6	

MEDIA

Any Screen	n=131 (50%)	n=157 (59%)	n=288 (55%)
Energy intake (kJ±SD)	2693 ± 1439	905 ± 913	1718 ± 1479
Energy density (kJ/g \pm SD)	5.4 ± 2.6	5.2 ± 6.0	5.3 ± 4.8
duration (min ± SD)	14.6 ± 11.4	13.9 ± 15.5	14.2 13.8
Television	n=57 (22%)	n=31 (12%)	n=88 (17%)
Energy intake $(kJ \pm SD)$	2977 ± 1575	1341 ± 1060	2400 ± 1613
Energy density $(kJ/g \pm SD)$	5.4 ± 2.4	7.6 ± 6.9	6.2 ± 4.6
duration (min ± SD)	14.7 ± 9.8	11.0 ± 21.6	13.4 ± 13.0
Computer	n=48 (18%)	n=92 (32%)	n=140 (27%)
Energy intake (kJ± SD)	2382 ± 1288	764 ± 872	1318 ±1285
Energy density $(kJ/g \pm SD)$	5.2 ± 2.9	4.8 ± 6.1	5.0 ± 5.2
duration (min ± SD)	14.3 ± 10.8	13.9 ± 12.4	14.0 ± 11.8
Handheld Device	n=13 (5.0%)	n=16 (6.0%)	n=29 (5.5%)
Energy intake (kJ± SD)	2325 ± 1580	916 ± 907	1547 ± 1421
Energy density $(kJ/g \pm SD)$	4.7 ± 2.2	5.5 ± 5.6	5.1 ± 4.4
duration (min ± SD)	12.6 ± 14.1	10.5 ± 8.2	11.5 ± 11.1
Multiple Devices	n=13 (5.0%)	n=18 (6.7%)	n=31 (6%)
Energy intake (kJ±SD)	2966 ± 934	869 ± 645	1748 ± 1300
Energy density $(kJ/a \pm SD)$	6.3 ± 2.8	2.4 ± 1.4	4.1 ± 2.8
duration (min ± SD)	17.4 ± 16.7	21.6 ± 20.3	19.9 ± 18.7
N		- 444 (440()	- 040 (45%)
	n=129 (50%)	n=111 (41%)	n=240 (45%)
Energy Intake $(kJ \pm SD)$	2771 ± 1900	1041 ± 1218	1962 ± 1821
Energy density $(kJ/g \pm SD)$	5.7 ± 3.4	7.0 ± 6.4	6.3 ± 5.1
duration (min \pm SD)	18.7 ± 21.0	11.2 ± 17.8	15.1 ± 20.4
TOTAL (n)	n=260	n=268	n=528
Energy intake (kJ± SD)	2742 ± 1671	962 ± 1050	1829 ± 1646
Energy density (kJ/g ± SD)	5.5 ± 3.0	5.9 ± 6.2	5.7 ± 4.9
duration (min ± SD)	16.7 ± 17.0	12.8 ± 16.5	14.7 ± 16.8

Table 18 Duration, energy intake and density of meals and snacks at home (N=312)

viewing media screens from 40 participants.

Context	Meals	Snacks	All meals
MEDIA AT HOME			
ANY SCREEN	n=93 (56%)	n=91 (62%)	n=184 (59%)
Energy intake (kJ±SD)	2751 ± 1685	1055 ± 992	1913 ± 1510
Energy density (kJ/g \pm SD)	5.3 ± 2.6	5.8 ± 6.1	5.6 ± 4.7
duration (min \pm SD)	13.3 ± 9.4	11.9 ± 12.5	12.6 ± 11.0
Television	n=53 (32%)	n=22 (15%)	n=75 (24%)
Energy intake (kJ±SD)	3095 ± 1584	1405 ± 1132	2600 ± 1627
Energy density (kJ/g ± SD)	5.6 ± 2.4	8.6 ± 7.3	6.5 ± 4.6
duration (min \pm SD)	14.9 ± 10.1	6.9 ± 6.0	12.6 ± 9.8
Computer	n=21 (13%)	n=45 (31%)	n=66 (21%)
Energy intake (kJ±SD)	2136 ± 1337	925 ± 956	1311 ± 1222
Energy density (kJ/g ± SD)	4.8 ± 3.0	5.5 ± 6.1	5.2 ± 5.3
duration (min \pm SD)	11.6 ± 7.4	12.2 ± 9.7	12.0 ± 9.0
OTHER SCREENS	n= 19 (12%)	n=24 (16%)	n=43 (14%)
Energy intake (kJ±SD)	2484 ± 1072	977 ± 870	1643 ± 1217
Energy density (kJ/g ± SD)	5.3 ± 2.7	4.0 ± 3.7	4.6 ± 3.3
duration (min \pm SD)	10.7 ± 8.7	15.7 ± 19.0	13.5 ± 15.4
NO Screen	n=72 (44%)	n=56 (39%)	n=128 (41%)
Energy intake (kJ±SD)	2551 ± 1685	890 ± 1019	1825 ± 1649
Energy density (kJ/g ± SD)	5.7 ± 3.4	7.5 ± 7.0	6.5 ± 5.3
duration (min \pm SD)	11.8 ± 10.1	8.5 ± 17.2	10.5 ± 13.8
TOTAL	n=165	n=147	n=312
Energy intake (kJ± SD)	2665 ± 1562	992 ± 1001	1877 ± 1567
Energy density (kJ/g \pm SD)	5.5 ± 3.0	6.5 ± 6.5	6.0 ± 5.0
duration (min ± SD)	12.7 ± 9.7	10.5 ± 14.5	11.7 ± 12.2

8.8. Discussion

In this study we sought to investigate whether wearable camera images could enhance the data obtained in dietary assessment, by objectively assessing the context surrounding eating episodes. Where data available for analysis, we found 71% of the eating episodes reported in the image-assisted 24-hour dietary recalls could be identified and annotated for context using the wearable camera images. Of those identified, it was possible to objectively determine the location, external environment, physical position, and whether the participant was engaged in social interaction or viewed media screens during the episode.

When combined with data obtained using an image-assisted 24h dietary recall ²⁹⁶ EI values associated with different contexts could be evaluated. For example, 8% of main meals were consumed out-of-home at food retailers, and were over twice as long and approximately 1196 kJ higher in energy (95% CI: 242 to 2149) than meals at home. The annotation was possible not only for the episode itself, but also using the images proceeding and succeeding the episode. An advantage of this was that it enabled the researchers to determine if the participant was in a social situation, where there appeared to be no direct social interaction e.g. eating alone at a work cafeteria. The images also allow greater specificity, such as the differences observed in duration and EI and density for eating episodes between different positional contexts (seated at tables versus sofas and standing/active).

Other studies exploring the use of technologies to assess eating contexts have recently been reported, but these studies have not recorded dietary intake. ²⁹⁷⁻²⁹⁹ Chen et al ²⁹⁷ explored eating patterns of forty university students wearing SenseCam for 1-5 days and used a similar procedure of manual annotation. In agreement with our data, they found that eating episodes with social interaction were significantly longer compared to eating alone (11.2 minutes versus 7.4), and episodes watching television were longer compared to episodes without television, but the difference was not statistically significant (8.8 minutes versus 7.8 minutes). These data support lab-based research that has indicated meal duration, ³⁰⁰ social interaction, ^{87, 301} and viewing screens are associated with increased EI. ²⁹¹

This study was not designed or powered to detect differences between specific contexts during eating episodes. For this reason we limited our analysis to five contexts similar to those currently assessed in national dietary surveys, ^{24, 85} and only compared the episode duration and EI data;

however, many other contexts relevant to behavioural nutrition and human eating research could also be assessed, such as food advertising, purchasing habits, meal frequency, plate size, social modelling, ambient lighting and total screen time. ^{41, 42, 302}

Wearable cameras, unlike direct observation, also offer the opportunity for secondary analysis, thus maximising the utility of data obtained. Additionally, raters can be trained before reanalysing the dataset, as we have shown the procedure to annotate the episodes for a category of interest can be easily tested for inter-reliability. This is in contrast to direct observation techniques where inter-rater reliability has only been established in a few studies. ^{301, 303, 304} However, noncompliance wearing the camera and technical failures resulted in 11% of data loss which must be considered. Where data available twenty-eight percent (214 episodes) of eating episodes captured could not be classified. The most common reason (85/214) was due to no associated images, mainly during snacking episodes. Some snack foods are easily consumed between the automated image capture (e.g. biscuits and nuts), thus faster imaging frequencies should allow a greater proportion of short eating episodes to be annotated. Dark and blurry images were also difficult to classify. Often breakfast and evening meals are consumed in low-light environments, especially when eating out at a restaurant or bar. Cameras better enabled to capture images in low light environments could reduce the magnitude of this limitation. ^{213, 296, 305}

Wear time was also an issue, with 7% of episodes occurring when the camera was not worn. These episodes were mainly at night when participants took the camera off, but remained awake for prolonged periods (self-reported eating episodes after the camera was switched off). Better instructions for participants to keep the camera on during all wake hours, and less obtrusive devices may increase wear time in future research. ^{213, 230, 296} Our data also reveals that future research investigating specific contexts will need to take into account the proportion of contexts that occur in free-living settings. For example, only 25% of the total snacking episodes were in occupational settings, thus if snacking habits at work were of research interest, the relatively low proportion of these events would need to be considered in study designs and power calculations.

This work relied on manual input and annotation of the data, but future work could develop automated computer vision techniques to process the data. $^{306, 307}$ The manual process took a substantial amount of time to input the dietary intake data, and the subsequent annotation took on average an additional 1 minute 21 seconds (± 44 s) for every eating episode. However, this work focused on the feasibility of using wearable cameras to record context attributes surrounding

eating episodes but could provide a gold standard reference for automated machine learning systems to detect behaviours and features within images. Automated image-analysis has potential to increase research efficiencies and allow image-based research on a larger scale. There are already considerable efforts being undertaken to passively assess dietary intake using wearing cameras, which aims to remove the need for the initial dietary assessment (used in this study) entirely, though extensive work is still required. ^{214, 221, 306} encouragingly, Jia et al ³⁰⁶ have demonstrated that a semi-automated system can estimate portion size with less bias and variability than human raters. However, only limited foods were assessed (n=100), and limitations exist in free-living settings that automated systems will need to address, such as poor lighting, dishes/meals that may obscure some foods on the plate, and current limitations of nutrient databases, as density data are not yet available for all foods. ³⁰⁶

Our procedure to manually annotate eating episodes demonstrated strong inter-rater reliability. For example, using the guidelines of Landis and Koch, the inter-rater reliability could be considered almost perfect for location (0.96), for indoor/outdoor/in vehicle (0.81), and sitting/standing position (0.84), while social interaction status (0.68), and the presence of media screens (0.79) were classified with substantial agreement. ³⁰⁸ Social interaction was challenging to annotate as interaction may only be captured in a few images scattered throughout the episode. Likewise, media screens were challenging to annotate due to their frequent but sometimes very brief use (i.e. not captured). Thus, cameras with faster imaging frequencies at higher resolution may allow easier classification of these contexts. Furthermore, future studies could incorporate the manual annotations procedures into the initial dietary assessments to reduce annotation time. In this study the manual annotation was conducted retrospectively, as the dataset was obtained from a completed study and therefore was more arduous than would be the case if completed during the dietary assessments.

Due to the exploratory nature and primary aims of this study all eating episodes from participants were treated as discrete data. Thus, one participant could contribute disproportionately for contexts with only a few episodes assessed. Furthermore the analysis grouped some distinctly different contexts together, such as fast food retailers, restaurants, bars and cafes. These different eating locations warrant their own subgroups, but were grouped together due to the small number of episodes observed. Additionally, our annotation procedure did not account for fast foods purchased for take-away consumption. In total, only 3 of 528 episodes were consumed at

fast food retailers, as participants usually consumed take-away meals in other locations. To address this, the annotation procedure would need to include where the food was purchased prior to the episode or the food type (if purchased by someone else). Additionally, we only explored the EI and energy density of meals and snacks, whereas examining the food groups and macronutrient content of the episodes would have provided an additional level of information. However, in future research such information could easily be examined for these types of datasets.

Our findings should be used to develop guidelines and protocols for visual research on nutrition behaviours and eating in free-living conditions to allow future studies to be uniformly compared. An ethical framework for visual research using wearable cameras has already been developed, ²⁵¹ together these would help to ensure future research is of a high quality with ethical considerations appropriately addressed.

8.9. Conclusion

In summary, this is the first study to assess the context of eating episodes using wearable camera images to enhance energy and nutrient data obtained during image-assisted dietary assessment. We show that the technique is feasible and provides a new approach to collect objective data on nutrition behaviours in a free-living setting that is currently lacking. ^{42, 285, 286} At this stage wearable cameras should not be viewed as a replacement device for traditional methods of dietary assessment, but instead as a complementary tool to both enhance traditional self-report and provide contextual data surrounding eating behaviours in free-living settings. Our findings should be used to develop guidelines and protocols for visual research on nutrition behaviours and eating in free-living conditions to allow future studies to be uniformly compared. Objective assessments will provide more detailed and valid information on the contexts of eating episodes in free-living settings.

Acknowledgements

We would like to thank all the participants in the study. We have no funding to declare.

CHAPTER 9. THESIS DISCUSSION

This thesis investigated the use of wearable cameras to (1) reduce the reporting bias associated with traditional self-reported dietary assessment, and (2) passively record and assess the context of dietary behaviours. Due to the recognised limitations of self-reported dietary intake, multiple efforts are underway to enhance dietary assessment through technology. ^{15, 31, 215, 309} However, there has been little research that has evaluated image-assisted methods of dietary assessment, or explored the potential of wearable cameras to enhance self-reported intake. In order to achieve the main aim of the thesis, a statistical analysis of the ANS08/9 datasets was first undertaken to determine the prevalence of low energy reporters (LERs), and highlight the need for technological innovation in dietary assessment. A systematic review was conducted to establish what is already known in the field of image-assisted dietary assessment, and what is still required. A feasibility study and a validation study were conducted to trial the use of wearable cameras in dietary assessment, and aimed to identify mechanisms and the magnitude to which wearable camera images could reduce measurement bias in 24-hour dietary recalls. It was hypothesised that images captured by wearable cameras would reduce the self-report bias for dietary energy intake (EI) compared to self-report alone by providing additional information to augment memory recall, and enhance self-report. To achieve the secondary aim of the thesis a manual image analysis of the dataset obtained in the validation study was undertaken to evaluate if the images captured using wearable cameras could be used to objectively and reliably assess the environmental and social context of eating episodes.

Primary findings

The primary findings of this thesis were: (1) 21% of men and 25% of women were classified as LERs in the ANS08/9, and a systematic bias was observed with LERs more prevalent among women, people aged >65 years, and Maori and Pacific people, (2) previous research up to 2013 suggests image-assisted methods of dietary assessment can provide objective information to independently verify and assess self-reported dietary intake, but the limited existing evidence highlighted the need for further research, (3) wearable cameras can reveal unreported or misreported errors in the 24h-hour dietary recall not captured by traditional methods alone, which

increases self-reported dietary EI, (4) wearable cameras reduce the magnitude of under-reporting bias for dietary EI in 24-hour dietary recalls by 9% in men and 6% in women compared to dietary recalls alone, and (5) wearable camera images can be analysed to objectively and reliably assess environmental and social contexts of dietary behaviours, therefore providing a novel method to passively record objective data on dietary behaviours in free-living settings.

Analysis of the ANS 08/9

Although the limitations of self-reported dietary intake were well established,^{2, 8, 9, 84} It had been over a decade since EI data obtained using the 24-hour dietary recall method had been evaluated using a national dataset. ⁷³ Moreover, substantial changes to physical characteristics of the NZ population were evident (Chapter Two), which are known to affect self-report. ^{73, 146, 152-155} Additionally, there was a reported paradox of decreased EI in parallel with increased body weight, therefore we undertook analysis of the ANS 08/9 data using the Goldberg cut-offs, ^{144, 184, 185} and estimated the prevalence of LERs by gender, body size (normal, overweight, obese), age, and ethnicity (Chapter Four).

After exclusions were made, a total of 3919 participants (NZ European n = 2286, Maori n = 976, and Pacific people n = 657), were included in the ANS 08/9 analysis. The main finding was that overall 21% of men and 25% of women were classified as LERs, and a systematic bias was observed, with LERs more prevalent among women, people aged >65 years, and in Maori and Pacific peoples. When compared to a similar analysis undertaken on the previous 1997 National Nutrition Survey (NNS97) data, ⁷³ the prevalence of LERs had increased substantially in nearly all subgroups over the decade, and notably amongst people of normal body size and in young adults. The findings revealed that caution must be exercised when interpreting EI data in the ANS 08/9, and indicated that under-reporting remains a key limitation of self-reported intake.

These primary findings provided justification to investigate the use of wearable cameras to enhance the 24-hour dietary recall. Other efforts trialling wearable cameras in dietary assessment had predominantly used the food record (FR) method. ^{205, 310} However, national nutrition surveys in N.Z., and in other countries, such as the U.S. and Australia use the 24-hour dietary recall. ^{25, 86, 93} Thus, efforts to improve the existing method may be more easily accepted and implemented into current dietary surveillance programmes. Moreover, using a similar (augmented/improved)

method of dietary assessment should allow data from previous surveys to be more easily compared than if a new method is used.

The ANS 08/9 analysis also demonstrated the apparent paradox of reduced EI with increased body weights was most certainly an artefact caused by self-report bias. The increase in the prevalence of LERs among people of normal body weight, and young adults, was potentially due to an increased influence of psychosocial factors (e.g. social desirability) and other behavioural characteristics (e.g. body dissatisfaction). ^{8, 10, 28} Observed differences over time may be explained by sociocultural changes and events within NZ society between the NNS97 and ANS 08/9 surveys; there has been an increase in screen-based activities, a substantial shift in the genre and quantity of television shows and media advertising, many of which portray a slim-body image or health-related content (e.g. reality television focussed on weight loss, cosmetic surgery, makeover, modelling etc.), and widely publicised nation-wide government-led health campaigns. ^{28, 199-203} Together, with an increased public awareness of nutrition-related health and dieting practises, this may have influenced people's self-perceptions, ²⁰⁰⁻²⁰³ and therefore increased the likelihood of dietary under-reporting. ^{28, 199} The increased prevalence of LERs amongst people of normal body size was considered when designing the validation study.

Systematic review

The systematic review aimed to examine the evidence of image-assisted methods for assessing dietary EI. There had been five previous reviews examining the use of technologies in dietary assessment, ^{16, 19, 29-31} but no review had specifically investigated the evidence for image-assisted dietary assessment. Therefore, it was deemed necessary and timely (due to rapid advances in computer handheld and wearable technologies) to define the field of image-assisted dietary assessment, group and categorise similar image-assisted methods developed. It was also required to inform and guide the use of wearable cameras, in the feasibility (review initially undertaken in 2011), and validation study conducted for this thesis. We aimed to examine all studies that have evaluated or validated an image-assisted method of dietary assessment compared to a reference method for assessing dietary EI.

A total of 13 eligible studies were included in the review, ten studies used active image capture using handheld devices ^{209-211, 222-224, 226-229} and three used passive image capture using wearable cameras.^{212, 213, 230} Eight studies evaluated five different image-based food records. ^{209, 211, 222-224, 226-229}

Chapter 9. Thesis discussion

²²⁶⁻²²⁸ Two studies evaluated the use of image-assisted written food records,^{210, 230} and three studies evaluated the use of image-assisted 24-hour dietary recalls.^{212, 213, 229} The main findings from the review were: (1) images captured during eating episodes can enhance self-report in both prospective and retrospective methods by revealing unreported foods and identify misreporting errors not captured by traditional methods alone, and (2) when images are used as the primary record of dietary intake they can be used to provide valid estimates of EI. However, the review also revealed how image-assisted methods that rely on independent image analysis can easily underestimate EI due to: (1) the diversity and complexity of foods that make it hard to identify certain ingredients and quantities of foods, (2) poor quality images that cannot be analysed, and (3) the need for users to capture images before all foods are consumed.

However, these limitations can be minimised. For example, EMA appears to be an effective way to remind participants to record intake (capture images), and development of comprehensive procedures for image analysis and standardised training regimes reduces the potential error and improves reliability. ^{209, 222} The review also revealed that image-assisted methods are preferred over their traditional FR's and 24-hour recalls, with participant feedback indicating that image-assisted methods are easier to use and have lower burden.

Given that image-assisted dietary assessment is an emering field, it was not surpising that the review primarily consisted of studies with either small sample sizes, ^{210, 212, 213, 224, 227, 228} or studies that only assessed relative validity using traditional methods of dietary assessment as reference methods. ^{15, 16, 18, 23, 24, 28, 29, 32} Morevoer, no studies had been conducted among children, adolescents, or in the elderly (aged >70 years). Specifically, the evidence regarding the use of wearable cameras were limited to three studies, ^{212, 213, 230} two of which only assessed relative validity for 1-day dietary intake. ^{213, 230} Thus, high-quality evidence was needed.

Feasibility study

The feasability study was designed to assess the degree to which the images assisted and changed the participants' self-reported dietary intake. Specific objectives' were to determine the effect viewing the images had on self-reported energy and nutrient intakes. The feasibility study included thirteen British adult participants (10 male, 3 female; mean age 33 ± 11 years), and assessed dietary intake for one-day (SenseCam-assisted 24-hour dietary recall). Overall, the assistance of the images increased self-reported EI by 12.5% (P=0.02). The increase in EI was
predominantly due to unreported food revealed by the images. Misreporting errors (incorrect food type reported or foods reported that were not consumed) and changes to portion size were also common, but together these resulted in only modest changes to EI.

The feasibility study also revealed limitations of the SenseCam, and this guided the design of the validation study. First it was found that the imaging frequency of the SenseCam (20-30 seconds) was insufficient to capture all foods consumed. This finding supported a similar observation made by Arab et al ²⁴⁵ using mobile phones worn around the neck with a 10-second imaging frequency. Other notable limitations were the position of the device, which allowed the lens to be affected easily by posture and body shape, and poor image quality, especially in low-light environments, which resulted in non-useful images. ²¹³ Technical issues were also experienced (data loss for two participants), and reports by other researchers using SenseCam ²³⁰ indicated wear days should be limited to the minimum required for the validation study aims. On the other hand, participant feedback, which indicated that wearing the camera and the method used to assess dietary intake was a low burden, was encouraging. Participants also indicated approximately one week would be an acceptable time to wear the SenseCam. Thus, it was evident the SenseCam was not an optimal device for image-assisted dietary assessment, but the promising results combined with favourable participant feedback provided the rationale, evidence, and experience required to validate the image-assisted 24-hour dietary recall.

Validation study

A total of 40 adults (20 male, 20 female) were recruited for the 15-day DLW validation study (PICTURE study) reported in Chapter Seven. Males were older than females (35 ± 17 years versus 28 ± 7 years, respectively) and predominantly overweight (BMI = 27 ± 8 kg/m²), while females were predominantly of normal body size (BMI = 22 ± 2 kg/m²). Given the technical issues experienced in the feasibility study, participants were only required to wear SenseCam on the familiarisation day, and each day prior to the three nutrition assessments (to provide the best possible chance the camera would operate normally and reduce data loss). Overall, use of the wearable cameras reduced the magnitude of under-reporting bias for dietary EI by 9% in men and 6% in women compared to dietary recalls alone. However, the use of DLW demonstrated images alone will not eliminate the under-estimation of dietary EI (men -9%, women -7%), unless satisfactory images are captured for all foods consumed to assist self-report (the various reasons

why images are not always captured or of satisfactory quality to assist self-report were discussed above in the findings for the feasibility study).

The alterations to self-report dietary intake data further supported results from the feasibility study, as unreported foods provided the biggest impact on EI, and other misreporting errors and changes to portion size were present but together contributed only 28% of the change to EI (unreported foods (n = 265) = 110,570 kJ vs. misreported foods (n=36) = -16852 kJ, and change to portion size (n = 51) = 13628 kJ). These results demonstrate that wearable cameras enhance the 24-hour dietary recall method, as the images reveal unreported foods and misreporting errors, and this helped to reduce under-reporting reporting bias for dietary EI.

Image analysis to assess context of eating episodes

The image dataset obtained in the validation study was used to objectively assess the context of self-reported eating episodes, and was reported in Chapter Eight. A manual procedure to review the images was used and each eating episode assessed for eating location, external environment (indoor/outdoor), physical position, social context, and viewing of media screens. This was the first study to combine dietary intake data with an objective assessment of context using wearable camera images in a free-living setting. Previous research investigating dietary behaviours has been primarily limited to laboratory studies, which are not always indicative of real-world eating behaviours, ^{42, 285-289} and self-reported context has limited potential scope and untested validity. Moreover, the recent inclusion of self-reported eating context recorded in large-scale dietary surveillance programmes ^{85, 86, 91} provided further rationale to explore and evaluate the potential of obtaining objective data through wearable camera images.

After exclusions image data were available for 107 days with a total of 742 eating episodes. The main finding was that 71% of the eating episodes could be classified for context through manual annotation, and the reliability of the assessment was considered almost perfect for location for indoor/outdoor/in vehicle and sitting/standing position, while social interaction status and the presence of media screens were classified with substantial agreement. The remaining 29% of episodes that could not be analysed suggested the need for faster imaging frequencies, and better image quality in low-light environments, as quick eating episodes were often missed (particularly snack foods), and dark/blurry images could not be analysed.

Overall strengths and limitations

Strengths and limitations of specific studies were discussed within their respective chapters. However, this thesis has its own strengths and weaknesses. Firstly, this was one of the first pieces of research that aimed to address a key limitation in the 24-hour dietary recall method by assisting memory recall through wearable camera images. Previous work developing software and systematic procedures has led to noteworthy enhancements of the 24-hour dietary recall method, but these improvements have limited ability to augment participants' memory recall, a pivotal requirement to enhance a retrospective assessment. Alternatively, this research was designed specifically to assist memory recall, and demonstrated that images could assist participants to report more accurate information, and allow researchers to verify all foods consumed reported during the interview (all foods captured by the camera).

Secondly, within the field of dietary assessment, this is the first research to thoroughly examine the mechanisms whereby wearable cameras assist self-report. Prior to this research, the specific factors that result in the underestimation of dietary EI had not been quantified. Thirdly, this was the first research to validate a wearable camera-assisted method using DLW in an adequately sized sample, and the DLW is the gold-standard methodology for validating dietary EI in free-living settings. Finally, this research examined two separate benefits of wearable cameras: the reduction of measurement errors, and the ability of wearable cameras to capture additional objective information on dietary behaviours. Together, these provide stronger justification and evidence for the use of wearable cameras in dietary assessment than when considered alone.

There are also a number of limitations that need to be considered. Firstly, both the feasibility study (Chapter Six) and PICTURE validation study (Chapter Seven) had relatively small sample sizes, and there were apparent demographic differences of the male and female samples in the validation study; However, the validation study (N = 40) is the second largest DLW study ever conducted evaluating a handheld/wearable technology-assisted method. This was partly due to the lack of financial resources available to conduct a larger study using DLW. Certain populations were therefore underrepresented, and additional studies are needed to ensure wearable cameras assist other population groups to the same degree.

Secondly, the feasibility and PICTURE study, only assessed dietary intake for one (Chapter Six) and three days (Chapter Seven) respectively. Therefore, due to both inter- and intra-individual

Chapter 9. Thesis discussion

daily and seasonal variability in peoples' diets, the results may not reflect habitual dietary intake.^{95, 166} Thirdly, the research did not assess whether wearing the SenseCam, despite early reports ^{212, 230, 245} and indications from the feasibility study ²¹³ affected dietary behaviours. The decision not to assess such reactivity was due to a number of reasons: (1) assessing reactivity would require different study designs that did not answer the primary aims of the thesis, and (2) due to the emerging nature and continuing rapid development of wearable technologies, it was judged that reactivity would likely differ as the technologies progressed. Therefore, reactivity was outside the scope of this research, but should be a consideration for future research.

Fourthly, by designing the studies to assess the mechanisms whereby wearable cameras assist self-report (alterations to self-report), the studies did not incorporate the image reviews as part of the usual 24-hour dietary recall procedures. It would be more efficient for future studies to combine the image review directly into the 24-hour dietary recall procedures. Moreover, only participants made changes to their self-report, but to obtain optimal accuracy it would be advantageous for trained image analysts conducting the assessments to verify the food types and portion size. ^{209, 222, 233, 234} However, this approach ensured interviewer bias was minimised.

Lastly, the SenseCam wearable camera used in this thesis is not optimal for dietary assessment. Limitations of the SenseCam were evident in the findings of the feasibility study, validation study, and secondary image-analysis to assess contexts. As discussed above the SenseCams size, poor image quality, and insufficient imaging frequency to capture quality images of all foods consumed, likely limited the full potential that wearable camera images may enhance self-report in the image-assisted 24-hour dietary recall (MP24+SC). Very recent improvements in wearable camera technology during the latter stages of this thesis have addressed some of these limitations identified, and such technologies continue to develop and improve rapidly.

9.1.1 Contributions of this thesis

The research conducted for this thesis provided the following contributions to the field of nutrition research and dietary assessment: (1) provided recent evidence highlighting reporting bias for self-reported El in a national nutrition survey dataset, and demonstrated under-reporting of El will remain a key limitation of self-reported dietary intake until technological innovations are employed, (2) undertook the first systematic review of image-assisted methods of dietary assessment, which revealed image-assisted methods can improve traditional dietary assessment methods, and highlighted the technical challenges that need to be addressed in future methods/research, (3) established that wearable cameras can enhance self-report in the 24-hour dietary recall by revealing unreported foods and misreporting errors, thereby increasing self-reported El and reducing reporting bias for dietary El, and (4) further demonstrated the potential utility of wearable cameras in nutrition research by evaluating their ability to passively record and objectively assess dietary behaviours in free-living settings.

9.2. Future implications

In the period of time since this research began, wearable technologies have become a hot topic in the media as both large and start-up companies have released consumer friendly wrist-worn fitness trackers, smart watches, and wearable cameras. Some wearable technologies were commercially available prior to 2011 (e.g. heart rate monitors), but the rapid increase in their capabilities and range of products available, in conjunction with the associated media attention, has changed the landscape significantly. Consequently, it is evident wearable technologies are rapidly becoming ubiquitous in society, but understanding of how these technologies could be applied in public health research is still in its infancy.²⁴⁴ To date, excluding the research undertaken in this thesis, there have only been three articles published in scientific journals evaluating the use of wearable cameras for dietary assessment among users.^{212, 230, 245} Therefore, although the findings of this thesis have implications for future research, they need to be considered in light of the early stages of this emerging field. This thesis has raised several questions regarding image-assisted dietary assessment, which warrant further investigation.

Firstly, it is evident that under-reporting of dietary EI will remain a key limitation of self-report unless technological innovation is employed (Chapter Three and Chapter Four). This is of concern

Umute List of research project topics and materials

Chapter 9. Thesis discussion

as epidemiological data reveals nutrition-related diseases, such as cancers, cardiovascular diseases, obesity, and diabetes have reached epidemic proportions. ⁶ Further, trends in body size indicate NZ and other countries are in positive energy balance. ^{58, 65, 78, 311, 312} Moreover, statistical modelling indicates body size is likely to rise even further. ¹⁹⁸

The high prevalence of LERs among people with overweight and obesity, and a notable increase in LERs among people with normal body weight (Chapter Four) dictates that new methods of dietary assessment need to focus on addressing the limitations of self-report itself, rather than simply improving procedures to collect self-report data. By design, image-assisted methods of dietary assessment independently verify self-reported intake, thereby minimising a key limitation of self-report alone (Chapter Five). Wearable cameras also remove the need for participants to manually record dietary intake for the period assessed, and further reduce the chance foods will be unreported compared to handheld devices, which still rely on participants' memory as images are captured manually (Chapter Five, Chapter Six, and Chapter Seven).

Secondly, assistance of the images helps to lessen both random and systematic errors. Random errors are reduced as the images can be used to identify the correct food type, provide independent estimates of portion size, and identify other misreporting errors (Chapter Five, Chapter Six, and Chapter Seven). Although the magnitude of the underreporting identified and reduced by the wearable camera images were relatively modest (Chapters Six, and Chapters Seven), they are important, as reducing error in measurement for any branch of science is essential for the incremental improvements in our knowledge. ³¹³ Moreover, reduced random error is especially beneficial for smaller studies, which cannot minimise the negative impact of random errors by increasing the number of observations and/or days diet that are assessed.⁸⁴

Systematic errors are also theoretically reduced, as the images help to ensure the data collected is of similar quality for all participants, compared to traditional methods which see systematic differences in reporting bias with respect to age, ^{70, 73, 145, 146} gender, ^{70, 145, 147-149} body size, ^{73, 146, 152-155} ethnicity, ^{73, 150, 151} and lifestyle.^{134, 156-158} Reducing systematic error is essential to enhance our understanding of nutrition-related diseases, as systematic differences between groups in the measurement errors are not usually amenable to correction, and can have serious consequences when interpreting data (erroneous findings). ^{84, 164} However, it must be noted that some participants using image-assisted methods may intentionally misreport their intake by choosing

not to capture images of foods during certain meals, and such error will go undetected in any future method.

Thirdly, the validation study demonstrated use of the wearable cameras reduced the magnitude of under-reporting bias for dietary EI by 9% in men and 6% in women compared to dietary recalls alone (Chapter Seven). This finding provides the first evidence that retrospective self-report can be improved beyond enhancements to the traditional methodological procedures. Wearable cameras should be trialled for use in large-scale dietary surveillance programmes and studies which incorporate the 24-hour dietary recall method. A reduced reporting bias for EI will better reflect the true dietary intake for populations compared to current data. This will better inform nutrition policies and public health interventions, and improve the chance of detecting significant associations in studies of diet and disease relationships.^{18, 34, 127, 159, 160}

Fourthly, the findings from the systematic review (Chapter Five) and the two studies undertaken (Chapters Six and Chapters Seven) indicate that, to some degree, under-estimation of dietary EI is likely to occur in image-assisted methods, unless images of satisfactory quality are captured for all foods consumed. Therefore, future image-assisted methods will need to minimise this common limitation. Methods which employ active image capture using handheld devices will require reminders for participants to record diet, and these should be customised (for participants) to ensure the cues are timely with respect to eating episodes. ²⁰⁹ Methods which use passive image capture from wearable cameras will be limited by imaging frequency, image quality, and wear-time. Therefore, technological improvements of these features compared to SenseCam used in this thesis (mean imaging frequency = $13.0s \pm 3.2s$, image quality = poor, mean wear time = 13 h, 0 min $\pm 2 h$, 23 min) should enhance the image-assisted methods further, while slower imaging frequencies, reduced image quality and wear time will likely see inferior results.

Lastly, wearable cameras provide additional objective information on dietary behaviours compared with traditional methods alone (Chapter Eight); the proposed inclusion of wearable cameras in dietary surveillance programmes would provide valid and reliable population-level data on dietary behaviours, and over time, trends in dietary behaviours could be assessed. Consequently, these new data may provide additional useful information regarding the populations' dietary intake and dietary behaviours compared to the data currently collected. Additionally, research investigating human eating and nutrition behaviours can use wearable cameras to conduct more research in free-living settings, and considering the numerous

contextual factors shown to influence dietary intake and food choice, ^{41, 42} a new tool to passively collect objective data in free-living settings would be useful. ^{286, 293} Furthermore, in addition to dietary intake and dietary behaviours, other health behaviours can be assessed using wearable cameras, such as exposure to food advertising, ³⁰² time spent in different modes of exercise, ²⁷¹ durations of active travel and routes, ²⁶⁸ sedentary behaviours, ^{270, 314} and the built environment. ³¹⁵ Therefore, future research that employ wearable cameras could potentially have a wider scope, which may allow resources to be pooled, and would better utilise the data collected and time given by participants.

9.3. Future research recommendations

This research established that wearable cameras can enhance self-report by reducing underreporting of dietary EI and can be used to assess the context of eating episodes through manual annotation. However, before it would be feasible to implement wearable cameras in large-scale research, further developments and testing of the image-assisted 24-hour recall are required. Moreover, other questions relating to the potential of wearable cameras remain, which warrant examination.

Development of software for cohesive image-assisted 24-hour dietary recall

To feasibly incorporate the image-assisted 24-hour recall into large-scale dietary studies, customised software is required to integrate the image-review and context assessment within a cohesive and efficient method.

Customised software should:

- Enable a simple or automatic (wireless) procedure to upload the images.
- Guide participants to screen the images efficiently and effectively by providing on-screen prompts and tips.
- Combine the input of dietary intake data (using an appropriate nutrient database) with the captured images to assist participants and trained researchers in image-analysis to attain optimal accuracy.

- Allow researchers to define the duration of eating episodes and assess contextual attributes for each episode easily as the interview progresses.
- Allow researchers to *easily link the* dietary intake data obtained during the nutrition assessment (foods, energy and nutrients) with the associated eating episode attributes e.g. time, duration, and contextual assessment.
- Achieve the most efficient and optimal method to complete the 24-hour recall procedures with the assistance of wearable camera images e.g. the method may only require a single pass over the 24-hour period, compared to the multiple passes currently used.

The following automated image-analysis features, if feasible to develop, would enhance the system and efficiency of the method further

- Filter all images that are dark/blurry. ^{306, 316}
- Identify and tag all potential eating episodes to guide researchers and recognise potential unreported foods.
- Develop onscreen portion size tools to assist portion size estimation.
- Enable face blurring techniques to reduce privacy concerns.
- Allow wireless upload of images with an online image viewer to enable telephone based image-assisted dietary recall, or web-based automated selfreported dietary recall, to be conducted as demonstrated by Arab et al (Image-DietDay).²¹²

Recommended studies in free-living settings

As indicated in Chapter Seven and in the limitations above, this research was conducted with relatively small sample sizes of healthy adults, and did not assess reactivity (behaviour change) while wearing the camera. Moreover, there are many other questions regarding the potential of wearable cameras in nutrition research that have arisen and need investigation. Therefore, a series of studies is required to address these gaps in the literature.

- Evaluation of image-assisted 24-hour dietary recalls in children, elderly (>65 years) and specific ethnic and socioeconomic population groups
 - There is a high prevalence of LERs among adults aged ≥65 years in dietary surveys, likely due to age-related decline in memory and cognitive function. Thus, wearable cameras may be ideally suited to enhance dietary recalls in this population group, and therefore should be evaluated.
 - Wearable cameras may allow dietary recalls to be completed among younger children than usual, and enhance the information collected from all children, and therefore should be evaluated.
 - Population groups prone to under-reporting, such as Maori, Pacific people people with obesity, or people on restricted diets, may provide more accurate information if their dietary intake is monitored by the wearable camera, and therefore should be explored.
- The feasibility and acceptability of image-assisted methods in large-scale research and monitoring
 - A study should evaluate if image-assisted 24-hour dietary recalls are feasible and cost effective compared to traditional recalls in a sample N≥100.
 - A study should evaluate the acceptability and privacy concerns related to largescale research using wearable cameras
- Assessment of reactivity while wearing wearable cameras
 - Reactivity (behaviour change) should be examined to ensure the dietary recalls assessed by wearable cameras are indicative of participants' usual intake. A within-person comparison between participants' wear and non-wear days could be used.
- Unannounced image-assisted 24-hour dietary recalls
 - Often unannounced 24-hour dietary recalls are conducted in nutrition research to reduce the potential of behaviour change during the assessment period.

Therefore, the feasibility of implementing unannounced image-assisted 24-hour dietary recalls should be evaluated. This would require participants to wear cameras for sustained periods e.g. seven days or more.

Wearable cameras in nutrition interventions

 Non-compliance reduces interventions effectiveness. Currently non-compliance is assessed using biomarkers (for some nutrients) or relies on self-reported dietary intake alone. Therefore, wearable cameras may provide a useful tool to monitor and verify compliance in dietary interventions, which should be explored.

Wearable cameras for private practice, clinical and community settings

 Reviewing the cameras images provides valuable insights into participants/patients/clients daily lives and the environments they live in. This information is potentially valuable for dietitians/nutritionists in private practice, and in clinical and community settings to make informed decisions and provide the best possible care.

Recommended laboratory studies

A series of laboratory studies can answer additional questions that may enhance image-assisted methods using wearable cameras further.

- Assessing the potential of image-assisted dietary recalls for periods longer than 24-hours
 - A key limitation of the 24-hour dietary recall, not addressed in this thesis, is that only one-day of dietary intake is assessed. Experience gained working with wearable cameras and reviewing images of eating episodes suggests wearable cameras may allow valid 48-hour or potentially 72-hour recalls to be conducted, which would greatly reduce the burden and costs of nutrition research compared to conducting multiple separate 24-hour dietary recalls.

The potential for dietary recalls longer than 24-hours could first be examined using weighed meals in the laboratory. For example, participants would consume two buffets meals (breakfast and lunch) in the laboratory over three successive days while wearing a wearable camera. On day four, an independent researcher (without knowledge of the meals) would assess participants' dietary intake for the six meals consumed over the previous 72, 48, and 24-hours using an imageassisted dietary recall.

Determining the optimal imaging frequency

 The optimal imaging frequency could be assessed in the laboratory to determine the ideal rate to capture images of all foods consumed, while maximising battery life and memory capacity.

Detecting eating episodes using wrist-worn accelerometers

• Amft et al ²⁵³ have demonstrated that eating can be detected using wrist worn accelerometers (to detect movement on three axis). Therefore, it might be possible to develop an algorithm to initiate image capture, or store images in the devices memory, only when or around the time (e.g. ± 5 seconds) eating is detected. This would greatly reduce the number of images captured, maximise battery life, and alleviate some privacy concerns. Moreover, the range of devices available, such as wrist-worn fitness trackers and smart watches (which contain accelerometers and allow customised software), are increasing rapidly, and thus may provide a suitable and practical device for participants to wear.

9.4. Conclusion

Wearable cameras reduce the magnitude of under-reporting bias for dietary EI in 24-hour dietary recalls by 9% in men and 6% in women compared to dietary recalls alone. The images reveal unreported foods and misreporting errors not captured by traditional self-report. Used in nutrition research, wearable cameras can verify and enhance self-reported dietary intake, and compared to self-report alone, allow additional information on dietary behaviours to be objectively assessed. Further benefits from using wearable cameras may be achieved using faster imaging frequencies (<10 seconds), high resolution images/video, and independent image-analysis to verify and/or assess food type and portion size reported. However, additional research and development of software is likely required to feasibly incorporate wearable cameras and the image-assisted 24-hour recall into large-scale dietary studies.

APPENDICES



APPENDIX 1

Systematic review data extraction form

Image-assisted dietary assessment: a systematic review of the evidence

Data extraction form

Title:		
Author:	Year:	Endnote Number #
Extracted by:	·	Source Database:

1.Settin	g							
2.Object	tive							
3.Metho	od of							
recruitm	nent							
4.Partici	pant Cha	aracterist	ics (e	e.g. male or fema	le pop	ulation,	age range)	
N=	Male	Fem	Female Age Range		Σ,	x Age x Body mass		x̄ ΒΜΙ
Health S	tatus: d at spec	ial popul	atior	n/setting:				
Population group Ssocioeconomic statu			tus	Edu	cation Level	Other		
5. Inclus	ion Crite	ria			Exc	clusion C	riteria	

6. Device:

7.Image-assisted	method:
------------------	---------

8.Study duration & protocol:

9. Statistical analysis

10.Reference methods used

11. Outcomes (mean energy intake/energy expenditure)

12.Feedback regarding technology/method:

13. Study limitations:

Included	Excluded
Reason:	

APPENDIX 2

Ethics approval letter for the feasibility study from the University

of Auckland Human Participants Ethics Committee

Office of the Vice-Chancellor Research Integrity Unit



The University of Auckland Private Bag 92019 Auckland, New Zealand

Level 10, 49 Symonds Street Telephone: 64 9 373 7599 Extension: 87830 / 83761 Facsimile: 64 9 373 7432

UNIVERSITY OF AUCKLAND HUMAN PARTICIPANTS ETHICS COMMITTEE

11-Apr-2012

MEMORANDUM TO:

Dr Cliona Ni Mhurchu SOPH General Admin

Re: Application for Ethics Approval (Our Ref. 7942)

The Committee considered your application for ethics approval for your project titled **Self-reported dietary intake using passive digital photography** on 11-Apr-2012.

Ethics approval was given for a period of three years with the following comment(s).

• The Committee notes the care with which its concerns have been addressed and acknowledges the changes made to try and address them. The Committee remains concerned about the reliance on participants to respect the privacy of others, particularly in private settings, and asks that in undertaking the pilot, you take particular care in instructing participants about this. In receiving approval for the pilot, you should also not assume that approval of an application for a main study will automatically follow. The Committee reserves the right to withhold or vary its approval for a main study, if required.

• It is made clear to participants that they will have an opportunity to review all their images in private before the researcher views them, which implies that if there are any inappropriately recorded images, these can be deleted/removed. It would be pertinent to state in the PIS and CF that this is the case.

• Please consider destroying the images once the researcher has reviewed them, rather than storing them for the same 15-year duration as other data.

The expiry date for this approval is 11-Apr-2015.

If the project changes significantly you are required to resubmit a new application to the Committee for further consideration.

In order that an up-to-date record can be maintained, you are requested to notify the Committee once your project is completed.

The Chair and the members of the Committee would be happy to discuss general matters relating to ethics approvals if you wish to do so. Contact should be made through the UAHPEC secretary at humanethics@auckland.ac.nz in the first instance.

All communication with the UAHPEC regarding this application should include this reference number: 7942.

(This is a computer generated letter. No signature required.)

Secretary University of Auckland Human Participants Ethics Committee

c.c. Head of Department / School, SOPH General Admin Ms Jennifer Utter Mr Luke Gemming Dr Christopher Bullen

Additional information:

1. Should you need to make any changes to the project, write to the Committee giving full details including revised documentation.

2. Should you require an extension, write to the Committee before the expiry date giving full details along with revised documentation. An extension can be granted for up to three years, after which time you must make a new application.

3. At the end of three years, or if the project is completed before the expiry, you are requested to advise the Committee of its completion.

4. Do not forget to fill in the 'approval wording' on the Participant Information Sheets and Consent Forms, giving the dates of approval and the reference number, before you send them out to your participants.

5. Send a copy of this approval letter to the Manager - Funding Processes, Research Office if you have obtained funding other than from UniServices. For UniServices contract, send a copy of the approval letter to: Contract Manager, UniServices.

6. Please note that the Committee may from time to time conduct audits of approved projects to ensure that the research has been carried out according to the approval that was given

APPENDIX 3

Ethics approval letter for the feasibility study from the Inter-

divisional research ethics committee, University of Oxford



SOCIAL SCIENCES & HUMANITIES INTER-DIVISIONAL RESEARCH ETHICS COMMITTEE

Hayes House, 75 George Street, Oxford. OX1 2BQ Tel: +44(0)1865 614871 Fax: +44(0)1865 614855 <u>ethics@socsci.ox.ac.uk</u> www.socsci.ox.ac.uk

Co-ordinator of the IDREC Social Sciences Divisional Office



Wednesday, 15 February 2012

Dr Charlie Foster Department of Public Health

Dear Dr Foster,

Research Ethics Approval

Ref No.: SSD/CUREC1A/12-008

Self-reported dietary intake using passive digital photography

The above application has been considered on behalf of the Social Sciences and Humanities Inter-divisional Research Ethics Committee (IDREC) in accordance with the procedures laid down by the University for ethical approval of all research involving human participants.

I am pleased to inform you that, on the basis of the information provided to the IDREC, the proposed research has been judged as meeting appropriate ethical standards, and accordingly approval has been granted.

Should there be any subsequent changes to the project, which raise ethical issues not covered in the original application, you should submit details to the IDREC for consideration.

Yours sincerely,

Kerry Vernon

cc: Paul Kelly, Department of Public Health

KV/EB

APPENDIX 4

Participant information sheet and consent form for the feasibility

study





British Heart Foundation Health Promotion Research Group, Department of Public Health, University of Oxford

Rosemary Rue Building, Old Road Campus, Headington, Oxford OX3 7LF, United Kingdom

Telephone: +44 (0)1865 289200 Email: I.gemming@ctu.auckland.ac.nz

Participant Information Sheet

Project title: Self-reported dietary intake using passive digital photography (PILOT STUDY)

Names of Researchers: Charlie Foster, Luke Gemming, Cliona Ni Mhurchu, Jennifer Utter, Paul Kelly, and Aiden Doherty

You are invited to take part in a study to evaluate a new method of nutrition assessment, using automated photography.

To help you make a decision about participating in the study, we ask that you read this information sheet.

Who is coordinating this study?

The study is coordinated by Dr Charlie Foster, and this research will be conducted by a PhD student Luke Gemming. This research is collaboration between the British Heart Foundation Health Promotion Research Group, The University of Oxford and the Clinical Trials Research Unit, The University of Auckland, New Zealand.

What is the aim of this study?

The aim of the study is to evaluate a new method of nutrition assessment using passive digital photography. A secondary aim is to assess which dietary habits, such as breakfast eating habits and location of meals the camera can capture effectively.

Why am I suitable for the study?

You are suitable for the study because you:

- Are 18 years of age or older
- You are healthy and do not have any dietary restrictions

You cannot take part in this study if you have any of the following conditions:

- You have a medical condition which severely restricts the foods that you can eat
- You have a medical condition that restricts your ability to freely complete usual activities of daily living

RAD PIS OXFORD V1.1 13/03/2012 UAHPEC REF NO: 7942

Where will the study take place?

Participants for this study will contact the researcher in response to advertisements on community and university notice boards. The assessments will take place at the Department of Public Health, at The University of Oxford.

How long will the study take?

Overall, the entire study will run for approximately 12 months, from April 2012 to April 2013. However, each person will only be involved for approximately two days (48hrs).

How many people will be recruited into the study? We are looking to recruit approximately 12 people.

What is involved if I take part?

If after reading this information sheet, you decide that you would like to take part in the study, you will meet with the researcher on two occasions: (1) instruction and training session (30mins) (2) nutrition assessment and feedback session (1-1.5 hours).

During the first session will you have the opportunity to ask any questions about the study and be asked to sign a consent form agreeing to take part. You will also receive a digital camera (Vicon Revue, see figure 1.) and provided training on its use. We anticipate the instruction and training session will take approximately 30 minutes.

This digital camera is different from other cameras as it takes photos automatically (every 20-30 seconds) and is worn around the neck on a lanyard. The camera has a special lens which allows it to capture almost everything in front of the person wearing it and is very simple to operate. A privacy button stops photos being captured when confidentiality is required.

You will be asked to wear the camera for two days (48hrs) (from wake to sleep) while continuing your usual daily activities. After this period you will meet with the researcher to have your diet assessed.



Figure 1. Vicon Revue digital camera

When you are wearing the camera you will be required to

use the privacy button or turn off the camera in certain locations, such as private residences (or public locations that have signage indicating the photography is prohibited). You will be provided with an information sheet detailing the locations in which photography is permitted or prohibited. After the 48h period you will meet with the researcher for the second session and have your diet assessed. You will be provided sufficient time to review all your images privately before they are viewed by the student researcher, and you can request more time if required. You will also be asked to complete a questionnaire to provide feedback on wearing the camera, and the use of images in nutrition assessment. Personal characteristics will also be recorded such as body weight, height, ethnicity, and education level. We anticipate this session will take approximately 1 - 1.5 hours.

RAD PIS OXFORD V1.1 13/03/2012

UAHPEC REF NO: 7942

Will there be any costs involved?

There are no costs involved in taking part in this study.

Withdrawal from study after data collection

You may withdraw from the study at any time and withdraw any data you have provided up to two months after your final visit.

What are the risks and benefits of this study?

Possible benefits

You will receive an assessment of your diet from a registered dietitian. This may provide valuable information to you about for the foods you eat.

Possible risks

Wearing the camera may make you feel uncomfortable in certain situations. A counsellors contact details can be provided if necessary. There are no other anticipated physical risks.

Will the information about me be kept confidential?

The study files and all information that you provide will remain strictly confidential. Questionnaires and forms you complete will be numbered 10-XXX and a list will be maintained linking the codes to your personal details. This list will be kept securely by the principal investigator. No material that could personally identify you will be used in any reports on this study and only the student researcher will have access to view the images you capture. The information including the images you capture will be kept securely at the British Heart Foundation Health Promotion Research Group, The University of Oxford, and The Clinical Trials Research Unit, The University of Auckland and destroyed after 15 years according to national research guidelines. All computer records will be password protected. Any information collected, as part of this study will not be used for any other purpose, without my permission and ethical approval, nor given to any other third party outside of the research team. However, if the images you capture indicate a serious crime or illegal activity the researchers will be obliged to report this to the relevant authorities. These authorities may request access to the image files as part of a criminal investigation.

During the study, ethics committee representatives, the student researcher and coinvestigators may check your records but will not be able to identify you (only the student researcher will have access to view the images you capture). This will only be done to check the accuracy of the information collected for the study and the information will remain confidential.

When will the results be available?

This study will take up to 12 months to conduct, so results will not be available until after April 2013. If you would like to be sent a copy of the overall results you can indicate this on the consent form.

Use of Data

The data obtained from this study will be used as part of a PhD thesis, and used for future conference presentations, and scientific publications.

Has the study received ethical approval?

Yes, this study has received ethical approval from the Central University Ethics Committee (CUREC) University of Oxford, and the University of Auckland Human Participants Ethics Committee (UAHPEC) New Zealand.

RAD PIS OXFORD V1.1 13/03/2012

UAHPEC REF NO: 7942

What is funding this research?

No funding is required for this research. The student conducting this research is supported by a New Zealand Health Research Council PhD stipend 10/077.

What are my legal rights?

Your participation in this study is entirely voluntary (your choice). You do not have to take part. If you choose not to take part in this study you will not be affected in any way. You may withdraw from the study at any time, without having to give a reason. Your withdrawal from the study will not affect your future health care. You are encouraged to ask questions at any time during the study. If you have any questions please contact:

Dr Charlie Foster Primary Investigator British Heart Foundation Health Promotion Research Group, Department of Public Health University of Oxford Rosemary Rue Building, Old Road Campus, Headington, Oxford OX3 7LF, United Kingdom Ph: + 44 (0) 18 65289241 Email: Charlie.foster@dhp.ox.ac.uk

Luke Gemming PhD Student Department of Public Health University of Oxford Rosemary Rue Building, Old Road Campus, Headington, Oxford OX3 7LF, United Kingdom Ph: +44 (0)1865 289200 Email: I.gemming@ctru.auckland.ac.nz

APPROVED BY CENTRAL UNIVERSITY ETHICS COMMITTEE (CUREC) UNIVERSITY OF OXFORD ON 15th February 2012. Reference Number: SSD/CUREC1A/12-008

Study Investigators

- Charlie Foster, Paul Kelly, and Aiden Doherty, British Heart Foundation Health Research Promotion Group, The University of Oxford
- Luke Gemming, and Cliona Ni Mhurchu Clinical Trials Research Unit, Faculty of Medical and Health Sciences, The University of Auckland
- Jennifer Utter, Faculty of Medical and Health Sciences, The University of Auckland

Thank you for taking the time to read about this study. Please keep this sheet for your information.

RAD PIS OXFORD V1.1 13/03/2012 UAHPEC REF NO: 7942





British Heart Foundation Health Promotion Research Group, Department of Public Health, University of Oxford

Rosemary Rue Building, Old Road Campus, Headington, Oxford OX3 7LF, United Kingdom

Telephone: +44 (0)1865 289200 Email: I.gemming@ctu.auckland.ac.nz

CONSENT FORM

This form will be held for a period of 15 years

Project title: Self-reported dietary intake using passive digital photography (PILOT STUDY)

Names of Researchers: Charlie Foster, Luke Gemming, Cliona Ni Mhurchu, Jennifer Utter, Aiden Doherty, Paul Kelly,

I have read and I understand the information sheet dated 13/03/2012 for volunteers taking part in this study. This study is designed to evaluate the potential of a new method of nutrition assessment using passive digital photography. I have had the opportunity to discuss this study with study researchers and I am satisfied with the answers I have been given.

- I understand that I will be passively (automatically) capturing digital photographs from a first person perspective
- I understand that I will meet with the researcher on two separate occasions taking approximately 2.5 hours of my time, and will be required to wear the digital camera for a 48h period.
- I understand that I will have sufficient time to review all my images privately before they are viewed by the student researcher, and I can ask for more time if I require it.
- I understand that taking part in this study is voluntary (my choice) and that I may withdraw at any time
 and withdraw any data that I have provided up to two months after my final visit.
- I understand that I will be required to be responsible and turn off the camera or use the privacy button in certain locations, such as private residences and anywhere photography is prohibited.
- I understand that my participation in this study is confidential and that no material that could identify me will be used in any reports on this study.
- I know whom to contact if I have any questions about the study.
- I understand that any data collected including the images I capture as part of this study will be stored securely for 15 years at British Heart Foundation Health Promotion Research Group, University of Oxford, and the Clinical Trials Research Unit, University of Auckland, New Zealand in accordance with the Privacy Act, 1993 (NZ). After this time the information will be safely destroyed.
- I understand that any information collected, as part of this study will not be used for any other purpose, without my permission and ethical approval, nor given to any other third party outside of the research team.
- I understand that there may be a significant delay between data collection and publication of the results.
- I understand that the results of the study will be published in scientific journals but none of these
 publications will contain information about me personally.

RAD CF OXFORD V1.1 13/03/2012

UAHPEC REF NO: 7942



• I wish to receive a copy of the results.

Participant to complete:

I	(print full name)
Agree to take part in this research.	
Date:// day/month/year	

One signed consent form is to be given to the participant and the second signed consent form is to be placed in the study record file.

APPROVED BY THE CENTRAL UNIVERSITY RESEARCH ETHICS (CUREC) COMMITTEE ON 15TH FEBRUARY 2012. REFERENCE NUMBER: SSD/CUREC1A/12-008

APPROVED BY THE UNIVERSITY OF AUCKLAND HUMAN PARTICIPANTS ETHICS COMMITTEE ON for (3) years, Reference Number/.....

RAD CF OXFORD V1.1 13/03/2012

UAHPEC REF NO: 7942
Registration form for the feasibility study

Page 1 of 2



This form is designed to assess the potential participant's eligibility for and interest in participating in the study, and to document assent, consent and attendance at the initial instruction and training session.

- Answer all questions. DO NOT LEAVE BLANK SPACES
- Refer to the Manual of Procedures for complete instructions

Please give the participant an explanation about what the study involves, using the PIS provided.

Explain to the participant that there are certain criteria for taking part in the study, which you need to check.

PART 1: Initial Assessment

1. Participant Details

1.01	Screening date	Date DD/MM/20YY
1.02	DOB	Date DD/MM/20YY
1.03	Sex	Male/Female (circle one)
1.04	height (m)	m
1.05	weight (kg)	kg
1.06	BMI (=weight/height ² ; kg/m ²)	kg/m²

RAD FROM A - Participant Registration 12/01/2012

2. Inclusion Criteria

2.01	Aged between 18 and 65 years	Yes/No
2.02	Healthy without strict dietary restrictions	Yes/No
2.03	Able to give written informed consent/assent	Yes/No
2.04	Speaks and understands English	Yes/No

3. Exclusion Criteria

3.01	Have a medical condition which severely restricts	Yes/No
	the food they can eat	
3.02	Have a medical condition that restricts their ability	Yes/No
	to freely complete usual activities of daily living	

4. Assent and Consent

Participant consent

4.07	consent obtained	Yes/No
	If NO to Q 4.07, participant is not eligible.	
4.08	If YES to Q 4.07, date of consent	Date DD/MM/20YY

5. CRF sign off - Signature of Researcher (PI)

Form Submitted: Eligibility Checked and if OK Participant.

5.01	Signature	
	Printed Name	
	Date	Date DD/MM/20YY

RAD FROM A - Participant Registration 12/01/2012

Brief information sheet for participants in feasibility study on

photography in public places

Photography in public and private locations

As a participant in the RAD pilot study you will be required to wear to a camera which captures photographs automatically for two days. Use this information sheet as guide and reference for locations in which photography is legally permitted or prohibited.

Where is photography permitted?

Members of the public do not need a permit to film or photograph in public places (legal advice from the metropolitan police, United Kingdom)

Photography is permitted in all public places (except when signage indicates otherwise)

Where should use the privacy button or turn the camera off?

Photography is prohibited and may be a criminal offence in the following locations

- · Private residences without explicit permission from the landlord or property owner
- Commercial and governmental premises with signage that indicates photography is prohibited

o e.g. banks, airports, embassies, customs and certain shops

Within a court building or its surroundings

If you're not certain if photography is permitted use the privacy button or the OFF button!

RAD Photography Info sheet V1.0 13/03/2012

UAHPEC REF: 7942

Participant contact details for feasibility study







Participant Initials

Participants Date of Birth

Registration Number

Please fill in your contact details, and preferred method of contact.

		_					
1.0	Participant Co	ontact Deta	ils				
1.01	Title						
1.02	First Name	I		Middle N	ames		
1.03	Surname						
1.03	Preferred meth	nod of contact					
2.00	Home O	Work	0	Mobile	0	Email	0
							_
1.04	Home phone						
1.05	Work phone						
1.06	Mobile phone						
1.07	Email						
	Postal Address						
1.08	Flat number, st and name	reet number					
1.09	Suburb						
1.10	City, district or	town					
1.11	Post Code						
		L					
2.0/	Alternative Co	ontact Deta	ils				

2.01 2.02 2.03	Title First Name Surname	Middle Name	15	
2.04 2.06 2.07	Home phone Mobile phone Email			

RAD FORM Z - PARTICIPANT Contact Details V 1.0 24/01/2012 UAHPEC REF NO: 7942

2.08	Postal Address Flat number, street number and name	
2.09	Suburb	
2.10	City, district or town	
2.11	Post Code	

3.0 Signature of Study Researcher

3.01	Name	
3.02	Signature	
3.03	Date	

RAD FORM Z - PARTICIPANT Contact Details V 1.0 24/01/2012 UAHPEC REF NO: 7942

Baseline demographics for feasibility study

Participant Initials	Participant DOB	Registration Number
Self-reported dieta	ry intake using passi	ive digital photography
Form B: RAD PILO	I STUDY (OXFORD):	Baseline Demographics

Survey Information Sheet

The purpose of this questionnaire is to collect some background information about you. This survey will take about 5 minutes to complete. If you have any questions then you can ask one of the researchers.

Please answer ALL questions. DO NOT LEAVE BLANK SPACES. Tick circles and write numbers in boxes.

The answers that you write are private and will not be shared with anyone other than the researchers. You do not need to tell anyone else what you write on this questionnaire.

If you are worried about anything, or have any questions let the researcher know.

1	Assessment Date					
1.01	1 Date			/		(dd/mm/yyyy)
2	About you					
2.01	Sex			Male	Female	
Which Please	e ethnic group or groups do yo e indicate Yes or No to every	ou be optio	long to? n.			
	White			Asian or Asi	an British	
2.02	British		2.08		Indian	
2.03	Irish		2.09		Pakistani	
2.04	Other white background		2.10		Bangladeshi	
	Mixed		2.11	Othe	r Asian background	
2.04	White & Black Caribbean			Black or Bla	ck British	
2.05	White & Black African		2.12		Caribbean	
2.06	White & Asian		2.13		African	
2.06	Chinese		2.14	Othe	er Black background	
			2.15	Any other gr	oup	

RAD – FORM B baseline demographics PILOT STUDY V 1.0 12/01/2012 UAHPEC REF NO: 7942

3 Work and Education

3.01 At present are you? (select one only)

- Self-employed
- Full time salary or wage earner
- Part-time salary or wage earner (less than 30 hrs/week)
- Retired
- Full-time home-maker
- Student
- Unemployed
- Other beneficiary
- Refuse to answer

If you answered 'full-time home-maker', 'student', 'unemployed', 'other beneficiary', or 'refuse to answer', go to Q 3.04. For all other answers, go to Q 3.02

3.02 What is your current occupation? Or if retired, what was your previous occupation? (select one only)

	Clerical or sales employee
	Semi-skilled worker
	Technical or skilled worker
	Business manager or executive
	Business owner or self-employed
	Teacher, nurse, police, other trained service worker
	Professional or senior government official
	Labourer, manual, agricultural or domestic worker
	Farmer owner or manager
	Other
	Refuse to answer
3.03	If Other, please specify

RAD – FORM B baseline demographics PILOT STUDY V 1.0 12/01/2012 UAHPEC REF NO: 7942

3.04	What is your highest educational qualification? (select one only) If None, select 'No school examinations'
3.05	 National Curriculum tests Key Stage 3 National Curriculum tests Key Stage 4 / GCSE, or GNVQ A-levels, International Baccalaureate or Cambridge Pre-U Other school qualification (e.g. overseas school) National Vocational Certificate / Trade Certificate Polytechnic/university course below Bachelors degree Bachelors degree Degree higher than Bachelor (Bachelors with honours, Masters, PhD) Other tertiary Refuse to answer No school examinations
0.00	
3.06	Which of these categories best describes you? (select one only) Married/living with partner
	Separated, divorced, widowed
	Refuse to answer
3.07	Please indicate which of these categories best matches your household's income (select one only)
RAD –	 Under £5,000 £5001 to £15,000 £25,001 to £35,000 £35,001 to £45,000 £60,001 to £70,000 Over £70,000 Don't know Refuse to answer

4 Nutrition

4.01 Are you currently on any specific or special diet? If yes please specify

4.02 Are you restricting yourself from eating any specific foods? If yes please specify

5 Signature of Study Researcher

5.01 Signature

Printed Name

Date

|__|__| /|__|_/|_2|_0|__|_|

RAD – FORM B baseline demographics PILOT STUDY V 1.0 12/01/2012 UAHPEC REF NO: 7942

Feedback survey form for the feasibility study



FORM F: RAD feedback survey

Self-reported dietary intake using passive digital photography

Initials |_|_|_| DOB |_|_|_|1|9|_|_|

Registration Number |_|_|-|_|_|

This questionnaire asks for your views on passive digital photography assisting dietary assessment. Your responses will provide important information regarding the use of photos for nutrition assessment and practicalities of wearing the camera. *Thank you for completing this questionnaire*!

Rate on the scale your response to the following statements. Tick one circle per row.

		Strongly disagree				Strongly		
		1	2	3	4	5	6	7
1.1	The images helped me to remember some foods I had forgotten about	0	0	0	0	0	0	0
1.2	The images helped me to remember extra details of how my foods were cooked, prepared or purchased	0	0	0	0	0	0	0
1.3	The images helped me to remember the portion size of the foods I ate	0	0	0	0	0	0	0
1.4	The images helped me to verify the portion size of foods I ate using the portion size guides	0	0	0	0	0	0	0
1.5	The images allowed me to provide more accurate information about the foods I ate	0	0	0	0	0	0	0
1.6	The images helped me to remember where I ate foods	0	0	0	0	0	0	0
1.7	The images helped me to remember who I ate my foods with	0	0	0	0	0	0	0
1.8	The images helped me remember the time that I ate foods	0	0	0	0	0	0	0

		Never	Seldom	Sometimes	Usually	Always
2.1	Wearing the camera affected my eating behaviour	Ò	0	0	0	0
	cuting schartour	If never	skip to que	estion 3.1		
2.2	How was eating behaviour affected (P	lease wri	te below)			

		Never	Seldom	Sor	netim	es U	Jsually	Alv	ways
3.1	I felt uncomfortable wearing the camera	Ò	0		0		0	(С
		↓ If never	skip to qu	estior	n 5.1				
3.2	2 In what situations did you feel uncomfortable? (Please write below)								
		strongly Stro disagree a						ongly Igree	
			1	2	3	4	5	6	7
3.3	I would feel more comfortable with a s camera that was hidden	maller	0	0	0	0	0	0	0

		Very						Very
		little						high
		burden				burden		
		1	2	3	4	5	6	7
5.1	Was the camera a burden to wear?	0	0	0	0	0	0	0
5.2	Was reviewing the images on the computer a burden?	0	0	0	0	0	0	0
5.3	Was the overall method to assess your diet a burden?	0	0	0	0	0	0	0

	1 day or less	2-3 days	A week	A fortnight	One month	Longer than a month
6 What is the maximum period you would wear the camera for?	0	0	0	0	0	0

If you have any other comments in regards to wearing the camera or the assessment of your diet please comment in the box below

Thank you for completing this questionnaire!

RAD Questionnaire V 1.0 12/01/2011

16

Ethical approval letter for the PICTURE study, University of

Auckland Human Participants Ethics Committee

Office of the Vice-Chancellor

Research Integrity Unit



The University of Auckland Private Bag 92019 Auckland, New Zealand Level 10, 49 Symonds Street Telephone: 64 9 373 7599 Extension: 87830 / 83761 Facsimile: 64 9 373 7432

UNIVERSITY OF AUCKLAND HUMAN PARTICIPANTS ETHICS COMMITTEE

30-Nov-2012

MEMORANDUM TO:

Dr Cliona Ni Mhurchu

SOPH General Admin

Re: Application for Ethics Approval (Our Ref. 8701)

The Committee considered your application for ethics approval for your project entitled Passive Image Capture to record everyday events (PICTurRE).

Ethics approval was given for a period of three years.

The expiry date for this approval is 30-Nov-2015.

If the project changes significantly, you are required to submit a new application to UAHPEC for further consideration.

In order that an up-to-date record can be maintained, you are requested to notify UAHPEC once your project is completed.

The Chair and the members of UAHPEC would be happy to discuss general matters relating to ethics approvals if you wish to do so. Contact should be made

through the UAHPEC Ethics Administrators at humanethics@auckland.ac.nz in the first instance.

All communication with the UAHPEC regarding this application should include this reference number: 8701.

(This is a computer generated letter. No signature required.)

UAHPEC Administrators

University of Auckland Human Participants Ethics Committee

c.c. Head of Department / School, SOPH General Admin

Ms Jennifer Utter

Mr Luke Gemming

Additional information:

1. Do not forget to fill in the 'approval wording' on the Participant Information Sheets and Consent Forms, giving the dates of approval and the reference number, before you send them out to your participants.

2. Should you need to make any changes to the project, write to the UAHPEC Administrators by email (humanethics@auckland.ac.nz) giving full details of the proposed changes including revised documentation.

3. At the end of three years, or if the project is completed before the expiry, please advise UAHPEC of its completion.

4. Should you require an extension, write to UAHPEC by email before the expiry date, giving full details along with revised documentation. An extension can be granted for up to three years, after which a new application must be submitted.

5. If you have obtained funding other than from UniServices, send a copy of this approval letter to the Manager - Funding Processes, UoA Research Office. For

198

UniServices contracts, send a copy of the approval letter to the Contract Manager, UniServices.

6. Please note that UAHPEC may from time to time conduct audits of approved projects to ensure that the research has been carried out according to the approval that was given.

Invitation letter for the PICTURE study





THE UNIVERSITY OF AUCKLAND FACULTY OF MEDICAL AND

National Institute for Health Innovation The University of Auckland

Level 4, School of Population Health Bldg Tamaki Campus Morrin Road, Glen Innes, Auckland Private Bag 92019 Auckland NEW ZEALAND Telephone: 64 9 373 7599, ext 84612 Facsimile: 64 9 373 1710 Email: I.gemming@nihi.auckland.ac.nz www.nihi.auckland.ac.nz

Dear NAME

DATE

Thank you for expressing interest in the PICTuRE study (Passive Image Capture to Record Everyday Events)

This letter contains the following items:

- 1. Participant Information Sheet
- 2. A map of the University of Auckland Tamaki Campus (overleaf)

As discussed in our earlier phone call, I would like you to come to the Auckland University Tamaki Campus, 261 Morrin Rd, Glen Innes (see attached map overleaf) on the Day, Date and Time for the baseline assessment. Please come to Room 100, Building 731. Enter the Tamaki Campus through Gate 1. Free guest parking is available to your left but alternatively you can park in any of the far rows as there are always spare parks.

Building **731** is the Sport and Exercise Building. This is the second building on your right as you enter through Gate 1. Enter into the building on the ground floor. If you veer right you will see some double doors opposite the elevator (**SES LABORATORIES Room 100**). These doors are locked but please press the doorbell. However, I should already be waiting outside for your arrival. If you have arranged an early time or are coming in on the weekend I may need to let you in at the gate. If so please call and I'll come to the GATE ASAP.

This session will take approximately 1-2 hours in duration.

- Participants: Please do not eat anything on the day of the assessment, and drink only water. i.e. overnight fast (10-12hrs). Please reschedule your appointment if you have a fever or feel unwell.
- Please make sure that you have passed urine in the morning before you come. Then, drink plenty of water so that you can give another urine sample when you arrive at the assessment.
- Please wear light clothing and comfortable shoes. (shoes required for the LAB)

Please read through the Participant Information Sheet again prior to this assessment to give you time to consider the information. I will contact you 1-2 days before the assessment to remind you of the appointment. Alternatively, you are welcome to phone me on 021 434317 (anytime)

If you have trouble finding the location of the room on the day of the assessment, please contact me on 021 434317.

I appreciate the time you have taken to consider your involvement in this study and I look forward to meeting with you.



Yours sincerely,

Luke Gemming PICTuRE Study Manager

CAMPUS MAP



Participant information sheet and consent form for PICTURE

study



National Institute for Health Innovation School of Population Health University of Auckland Private Bag 92019 Auckland Mail Centre, Auckland, 1142

Telephone: +64 21 434317 Email: l.gemming@nihi.auckland.ac.nz

Participant Information Sheet

Project title: Passive Image Capture to Record Everyday Events (PICTuRE)

Names of Researchers: Luke Gemming, Cliona Ni Mhurchu, Jennifer Utter, Elaine Rush, Aiden Doherty.

You are invited to take part in a study using a new digital method to capture everyday events of people's lives with a special camera called SenseCam. To help you make a decision about participating in the study, we ask that you read this information sheet.

Who is coordinating this study?

The study is coordinated by Associate Professor Cliona Ni Mhurchu, and this research will be conducted by a PhD student Luke Gemming at the National Institute for Health Innovation, The University of Auckland, New Zealand.

What is the aim of this study?

The aim of this study is to record people's usual daily activities and eating episodes using passive image capture. The images captured will be used to evaluate a new method to assess dietary intake, explore the context surrounding eating episodes, and may be used to assess the duration of typical daily activities, such as watching television or traveling to work.

Why am I suitable for the study?

You are suitable for the study because you:

- Are 18 65 years of age
- You are healthy and do not have any strict dietary restrictions
- You can operate basic functions of a computer
- Can read and understand the English

You cannot take part in this study if:

- You plan on travelling during the study period
- · You plan to conduct extra (above normal) vigorous exercise during the study
- · You have had an recent acute illness
- You who have had blood tests or intravenous fluids two weeks prior or scheduled during the testing period
- · You plan to have an increased physical activity level during the testing period
- · You have a medical condition which severely restricts the foods that you can eat
- You have a medical condition that restricts your ability to freely complete usual activities of daily living

PICTURE PIS DLW VALIDTION STUDY V1.1 14/11 /2012

Where will the study take place?

Participants for this study will contact the researcher in response to advertisements and the assessments will take place at The School of Population Health, University of Auckland and within your own home if requested.

How long will the study take?

Each participant will only be involved for 15 days but overall, the entire study will run for approximately 12 months, from February 2013 to February 2014.

How many people will be recruited into the study?

We are looking to recruit approximately 40 people.

What is involved if I take part?

If you decide that you would like to take part in the study you will meet with the researcher on four occasions over a 15 day period. You will be required to attend a baseline assessment session, and have your diet assessed three times. The baseline assessment will be at the School of Population Health, University of Auckland in Glen Innes but the other three appointments can be in your own home if requested.

Day 0: Baseline assessments - 2hr

Days 3, 9, 15: Three nutrition assessments - 1.5h each

Study Timeline



Day 0 – Baseline Assessments

During the first session will you have the opportunity to ask any further questions about the study before providing consent to participate.

You will arrive at the first session after fasting (not eating) over-night to complete baseline assessments including:

- A urine sample
- Assessment of height, body weight, resting energy expenditure, and percentage body fat.
- Background and demographics questionnaire

PICTURE PIS DLW VALIDTION STUDY V1.1 14/11 /2012
You will be shown the SenseCam digital camera that is used in the study. SenseCam is different from other cameras as it takes photos automatically every 20-30 seconds and is worn around the neck on a lanyard/strap. The camera has a wide-angle lens which allows it to capture almost everything in front of the person wearing it and is very simple to operate. A privacy button stops photos being captured when confidentiality is required, and you will have the opportunity to screen the photos privately and delete any photo you wish before the researcher views the images.



As a participant you will be required to wear the SenseCam for four days from wake to sleep while continuing your usual daily activities. The images will be used to help assess your diet, and the context

SenseCam Digital Camera

around which foods are consumed. The images will also be used to assess the time spent in usual activities of daily living, such as time spent watching television or travelling to work.

Screening photos

After each nutrition assessment you will have a chance to screen the images you have captured privately before they are viewed by the researchers. You will be provided adequate time to screen the images and you can request more time if required. This process usually takes about 10-15 minutes and is usually enjoyed by participants as they watch their own recording of their life similar to a short movie.

Participant obligations

While wearing the SenseCam you will be required to gain permission to capture images in private locations, such as other people's homes, private clubs, or certain workplaces. If you do not have permission to capture images you must use the privacy button or turn the camera off. An information sheet detailing the locations which photography is permitted or prohibited will be provided and you will be given verbal instructions by the researcher to assist your understanding. You will also be given basic information cards to provide to third parties. This card will help third parties make an informed decision regarding their privacy.

Doubly Labelled Water

After completion of the baseline assessments you will be provided a small dose (about 150ml) of special water called doubly labeled water to drink. This water tastes like normal water and contains an invisible marker called an isotope that can be traced and measured to assess your energy expenditure. This marker is completely safe and is commonly used to assess body composition and energy expenditure in research. The doubly labeled water will be eliminated naturally from your body in urine, sweat, and the moisture in your breath.

Five hours after consuming the water you will provide a second urine sample. This can be collected by you at your home. You will be provided instructions on how this is done correctly.

Days 3, 9, 15: Nutrition Assessments – 1.5 hours

On days 3, 9 and 15 you will meet with the researcher to have your diet assessed and provide urine samples which takes about 1.5 hours. On day 15 you will also have your weight remeasured, and complete a feedback survey. On completion of the study you will receive an \$80 Grocery voucher.

Will there be any costs involved?

There are no direct costs involved in taking part in this study, but there may be some transport costs to attend the baseline session 1.

PICTURE PIS DLW VALIDTION STUDY V1.1 14/11 /2012

Withdrawal from study after data collection

You may withdraw from the study at any time and withdraw any data you have provided up to two months after your final visit.

What are the risks and benefits of this study?

Possible benefits

You will receive three assessments of your diet from a registered dietitian. This may provide valuable information to you about for the foods you eat. You will also have your energy expenditure assessed using a gold standard method. This will give you an estimate of the energy you expend each day over a two week period which may be useful for future weight management.

Possible risks

Wearing the camera may make you feel uncomfortable in certain situations but you can remove the SenseCam at any time. There are no other anticipated physical risks.

Will the information about me be kept confidential?

The study files and all information that you provide will remain strictly confidential. Questionnaires and forms you complete will be numbered 20-XXX and a list will be maintained linking the codes to your personal details. This list will be kept securely by the principal investigator. No material that could personally identify you will be used in any reports on this study and only the researcher directly involved in this study will have access to view the images you capture. All data will be stored securely at The National Institute for Health Innovation, University of Auckland and destroyed after 15 years according to national research guidelines. The images you capture will be destroyed once the analysis has been completed. All computer records will be password protected. Any information collected, as part of this study will not be used for any other purpose, without your permission and ethical approval, nor given to any other third party outside of the research ream. However, if the images you capture indicate a serious crime or illegal activity the researchers will be obliged to report this to the relevant authorities. These authorities may request access to the image files as part of a criminal investigation.

During the study, ethics committee representatives, the student researcher and coinvestigators may check your records but will not be able to identify you (only the researchers will have access to view the images you capture). This will only be done to check the accuracy of the information collected for the study and the information will remain confidential.

When will the results be available?

This study may take up to 12 months to conduct, so results will not be available until mid-2014. If you would like to be sent a copy of the overall results you can indicate this on the consent form.

Use of Data

The data obtained from this study will be used as part to form part of a PhD thesis, and used for future conference presentations, and scientific publications.

Has the study received ethical approval?

Yes, this study has received ethical approval from The University of Auckland Human Participants Ethics Committee (UAHPEC) New Zealand.

Who is funding this research?

PICTuRE PIS DLW VALIDTION STUDY V1.1 14/11 /2012

The funding is provided by the National Institute for Health Innovation. The student conducting this research is supported by a New Zealand Health Research Council of New Zealand PhD stipend (10/077).

What are my legal rights?

Your participation in this study is entirely voluntary (your choice). You do not have to take part. If you choose not to take part in this study you will not be affected in any way. You may withdraw from the study at any time, without having to give a reason. Your withdrawal from the study will not affect your future health care. You are encouraged to ask questions at any time during the study. If you have any questions please contact Luke or other investigators:

Cliona Ni Mhurchu National Institute for Health Innovation Primary Investigator Ph: +64 (9) 923 84494 Email: <u>c.nimhurchu@nihi.auckland.ac.nz</u>

Luke Gemming PhD Student National Institute for Health Innovation Ph: +64 021 43 43 17 Email: I.gemming@nihi.auckland.ac.nz

Chris Bullen Director National Institute for Health Innovation Ph: +64 (9) 923 84730 Email: c.bullen@nihi.auckland.ac.nz

Address: National Institute for Health Innovation, School of Population Health University of Auckland, Tamaki Campus, The University of Auckland 261 Morrin Road, Glen Innes, Auckland 1072

For any queries regarding ethical concerns you may contact the Chair, The University of Auckland Human Participants Ethics Committee, The University of Auckland, Research Office, Private Bag 92019, Auckland 1142. Telephone 09 373-7599 extn. 87830/83761. Email: humanethics@auckland.ac.nz

APPROVED BY THE UNIVERSITY OF AUCKLAND HUMAN PARTICIPANTS ETHICS COMMITTEE ON 30th November for (3) years, Reference Number 8701.

Study Investigators

Cliona Ni Mhurchu, Luke Gemming, National Institute for Health Innovation, Faculty
of Medical and Health Sciences, The University of Auckland

Jennifer Utter, Faculty of Medical and Health Sciences, The University of Auckland

 Elaine Rush, Faculty of Health and Environmental Sciences, Auckland University of Technology

Aiden Doherty, British Heart Foundation Health Research Promotion Group, The University of Oxford

Thank you for taking the time to read about this study. Please keep this sheet for your information.

PICTURE PIS DLW VALIDTION STUDY V1.1 14/11 /2012



Brief information sheet on photography for the PICTURE study

Photography in public and private locations

As a participant in the PICTuRE study you will be required to wear to a camera which captures photographs automatically for four days. Use this information sheet as a reference and guide for locations in which photography is legally permitted or prohibited.

Where is photography permitted?

Members of the public do not need a permit to film or photograph in public places.

i.e. Photography is permitted in all public places (except when signage indicates otherwise)

You can capture images in private locations (e.g. homes, workplaces, clubs) BUT only
if you have permission first! Ask the home owner/tenant or senior member of the
premises for permission. (use the study information cards to help explain)

Where should use the privacy button or turn the camera off?

Photography is prohibited and may be a criminal offence in the following locations

- · Private residences without explicit permission from the landlord or property owner
- Commercial and governmental premises with signage that indicates photography is prohibited

o e.g. banks, airports, embassies, customs and certain shops

Within a court building or its surroundings

Other locations we recommend using privacy button or turning the camera off!

- Close proximity to public bathrooms or changing rooms
- Schools or childcare centres
- Public swimming pools or the beach
- Any other location that you feel photos should not be captured

If you're not certain if photography is permitted ask for permission first otherwise use the privacy button or the OFF button!

PICTURE_Photography_Sheet V1.1 10/11/2012 REF 8701

SenseCam user instructions for PICTURE study

Using the SenseCam

Turning on/off – you can turn the SenseCam on or off by pressing the button in the top of the device for a few seconds. When it is turning on it will make a beep sound at an orange light will appear beside the button.

Turn the SenseCam off when you are not wearing it to save the battery. You will most likely only need to turn it off when you are going to bed or if you decide you do not want it to record anything for an extended period.

- Privacy button press this button to temporarily stop the device from taking pictures. It will reactivate automatically after 7 minutes.
- Activate button this button allows you to take a picture manually or to reactivate the device if you had previously pressed the privacy button.
- Status lights an orange flashing light will indicate every time an image is captured. A red light indicates that the privacy button has been pressed and the device is not taking any images at this time.

Charging

You will be given a charger lead with a plug on one end and a small square plug on the other end. It is recommended that you charge SenseCam at night when you are sleeping so that the battery will be full for the next day.

To charge the SenseCam, put the small end into the SenseCam in the slot on its side (see picture below) and plug the other end into your domestic plug socket.



Participant registration form for PICTURE study



PICTURE Study Form A: Participant Registration, Consent and Attendance at the Instruction and Training Session Participant Initials Participants Date of Birth Participant Initials

This form is designed to assess the potential participant's eligibility for and interest in participating in the study, and to document assent, consent and attendance at the initial instruction and training session.

- Answer all questions. DO NOT LEAVE BLANK SPACES
- Refer to the Manual of Procedures for complete instructions

Please give the participant an explanation about what the study involves, using the PIS provided.

Explain to the participant that there are certain criteria for taking part in the study, which you need to check.

PART 1: Initial Assessment

1. Participant Details

1.01	Screening date	Date DD/MM/20YY
1.02	DOB	Date DD/MM/20YY
1.03	Sex	Male/Female (circle one)
1.04	height (m)	m
1.05	weight (kg)	kg
1.06	BMI (=weight/height ² ; kg/m²)	kg/m²

2. Inclusion Criteria

2.01	Aged between 18 and 65 years	Yes/No
2.02	Healthy without strict dietary restrictions	Yes/No
2.03	Able to give written informed consent/assent	Yes/No
2.04	Speaks and understands English	Yes/No

PICTuRE FORM A - Participant Registration V 1.0 25/09/2012 Ref:8701

3. Exclusion Criteria

3.01	Have a medical condition which severely restricts the food they can eat	Yes/No
3.02	Have a medical condition that restricts their ability to freely complete usual activities of daily living	Yes/No
3.03	Have had an recent acute illness	Yes/No
3.04	Intend to travel during the testing period	Yes/No
3.05	Have had blood tests or intravenous fluids two weeks prior or scheduled during the testing period	Yes/No
3.06	Plan to have an increased physical activity level during the testing period	Yes/No

4. Assent and Consent

Participant consent

4.07	consent obtained	Yes/No
	If NO to Q 4.07, participant is not eligible.	
4.08	If YES to Q 4.07, date of consent	Date DD/MM/20YY

5. CRF sign off - Signature of Researcher (PI)

Form Submitted: Eligibility Checked and if OK Participant.

5.01	Signature
	Printed Name
	Date Date DD/MM/20YY

Participant contact details form for PICTURE study

						PICTuRE
FOR	M Z: CONT	ACT DETAILS				
P	articipant Initials		Participants D	ate of Birth	Registration N	lumber
	_ _ _		_ _ - _ _ -	1 9 _ _	_ _ - _ _	
Plea	Please fill in your contact details, and preferred method of contact.					
1.0 F	Participant	Contact Deta	ails			
1.01	Title]	_		
1.02	First Name			Middle Names		
1.03	Surname			J		

1.04	Home phone	
1.05	Work phone	
1.06	Mobile phone	
1.07	Email	
	-	
	Postal Address	
1.08	Flat number, street number and name	
1.09	Suburb	
1.10	City, district or town	
1.11	Post Code	

2.0 Alternative Contact Details

2.01	Title	
2.02	First Name	
2.05	Sumane	
2.04	Home phone	
2.06	Mobile phone	
2.07	Email	
	Postal Address	
2 08	Flat number, street number	
2.00	and name	
2.09	Suburb	
2.10	City, district or town	
2.11	Post Code	

PICTURE FORM Z - PARTICIPANT Contact Details V 1.0 18/09/2012 UAHPEC REF NO: 8701

3.0 Signature of Study Researcher

3.01 Name

3.02 Signature 3.03 Date

PICTURE FORM Z - PARTICIPANT Contact Details V 1.0 18/09/2012 UAHPEC REF NO: 8701

Participant characteristics form for PICTURE study

Participan	Sinetie Bata	
	t Initials	
Date of bir	rth (dd/mm/yyyy)	
Registratio	on Number	
Section '	1	
Anthropo	, ometric data baseline	
1.01	Date of baseline assessment	dd/mm/yyyy
1.02	Height (cm)	
1.03	Weight (kg)	
Section 2		
2.01	ng, resting metabolic rate	
	Baseline urine sample time	
		hh:mm
2.10		
	Impedance 1 (Z)	
2.11	Phase 1 (P)	
2.12		
2.02	Resistance 1 (R)	
	Reactance 1 (Xc)	·
2.02	Impedance 2 (7)	
2.02		
2.02	Phase 2 (P)	'
	Resistance 2 (R)	
2.02	Reactance 2 (Xc)	
2.02	If difference between resistance 1 and	
2.02	resistance 2 is >1 ohm, take a third measure:	
	Impedance 3 (Z)	
2.02	Phase 3 (P)	
2.02		
2.02	Resistance 3 (R)	·
	Reactance 3 (Xc)	·
2.02	% body fat (Using Excel Spreadsheet)	
TURN OVER FOR DLW DOSING		

PICTuRE Form P Version 2.0 26/03/2013



DLW DOSING			
2.03	Fat mass (kg)		
		`	
2.04	Fat free mass (kg)		
		·	
2.05	Body water (kg) (Dosing calculator)		
2.06	Dose calculated (g) (Dosing calculator)		
2.07	Dose given (g)		
2.08	Time of dose		
		hh:mm	
2.09	Any dose lost	Yes/No	
	MOXUS: VO2		
		`	
	MOXUS: VCO2		
		`	
2.10	Resting metabolic rate		
		·	

Section 3

Anthropometric data 15 days			
4.01	Weight (kg)		
		`_	

PICTuRE Form P Version 2.0 26/03/2013

Urine log form for PICTURE study

PICTURE

Form D: Urine Sample Log

Page 1 of 3

	Participant date of birth	
Participant initials	day month year	Registration number

We will be asking you to collect urine samples over the following days:

	Date	Time		
Sample	day month y	ear	нн	ММ
Baseline	2 0			
5 hours	2 0	at		
Day 3	2 0			
Day 9	2 0			
Day 15	<u> 2 0 </u>			

Instructions

With this form, we will give you four labeled yellow jars for collecting urine.

The first time you empty your bladder in the morning (into the toilet) record the time and date. Do not collect this sample.

The next time you go to the toilet, collect your urine in the white container provided. Transfer your urine from the white container into the yellow jar (filling it to two thirds). Then pour the left over urine into the toilet and record the time and date. Place the lid on the jar firmly, place the jar in the plastic bag provided and refrigerate, or if possible freeze the urine. It is important that no water from the toilet is mixed with this urine sample.



PICTURE Study • Form D: Urine Sample Log © The National Institute for Health Innovation, The University of Auckland, 2013

Form D vDF01

PI	CTURE Form D: Urine Sample Log Page 2 of 3
	Participant initials Participant date of birth ay month year
1.	Baseline Sample
1.01	day month year Date
Instru you to	ctions: Please record the time you first passed urine in the morning. Please then record the time ok the urine sample.
1.02	Today, I first passed urine at:
	HH O AM O PM (Tick one only)
1.03	Today, I took my urine sample at:
	HH MM O AM O PM (Tick one only)
2.	Five Hour Sample
2.01	day month year Date
Instru time y	ctions: Please record the time you took the dose of doubly labeled water. Please then record the ou took the urine sample.
2.02	Today, I took the doubly labeled water at :
	HH MM O AM O PM (Tick one only)
2.03	Today, I took my urine sample at:
	HH MM O AM O PM (Tick one only)
3.	Day 3 Sample
3.01	day month year Date
Instru you to	ctions: Please record the time you first passed urine in the morning. Please then record the time ok the urine sample.
3.02	Today, I first passed urine at:
	HH MM O AM O PM (Tick one only)
3.03	Today, I took my urine sample at:
	HH MM O AM O PM (Tick one only)
_	

PICTURE Study * Form D: Urine Sample Log © The National Institute for Health Innovation, The University of Auckland, 2013

Form D vDF01

PI	CTURF Form D: Urine Sample Log Page 3 of 3
•••	Participant initials Participant date of birth day month year
4.	Day 9 Sample
4.01	day month year Date
Instru you to	ctions: Please record the time you first passed urine in the morning. Please then record the time pok the urine sample.
4.02	Today, I first passed urine at:
	HH MM O AM O PM (Tick one only)
4.03	Today, I took my urine sample at:
	HH MM O AM O PM (Tick one only)
5.	Day 15 Sample
5.01	day month year Date
Instru you to	ctions: Please record the time you first passed urine in the morning. Please then record the time ook the urine sample.
5. <mark>0</mark> 2	Today, I first passed urine at:
	HH MM O AM O PM (Tick one only)
5.03	Today, I took my urine sample at:
	HH MM O AM O PM (Tick one only)
6.	Signature of Study Manager
6.01	signature printed name day month year
	signature prince name and the prince

PICTURE Study • Form D: Urine Sample Log © The National Institute for Health Innovation, The University of Auckland, 2013

Form D vDF01

24-hour dietary recall data sheet for PICTURE study

			-		
		Dictore			
-URIVIL: 74		Therary			
01111 01 24	i necu		THE REAL PROPERTY AND ADDRESS OF ADDRESS OF ADDRESS ADDRES	21 marci	

Passive Image Capture to Record Everyday Events (PICTuRE)

Participant Initials: |_|_|_| DOB: |_|-|-|_|-|_| Registration Number: |_|_|-|_|_|

The purpose of this form is to collect all dietary intake data during the nutrition assessments. The 24h recall is collected once for the pilot and twice for the main study.

Nutrition Assessment No. 1 / 2/3 Time of Image Review:_____ DAY 15 Body Weight:____

Day of week MON TUE WED THUR FRI SAT SUN

Give the participant a brief explanation of what the recall involves (quick list, detailed description and review):

"I will be asking you to tell me about all the food and drink that you had yesterday, from the time that you got up to the time that you went to bed. First I will ask you to try and remember all the food and drink that you had yesterday as a list. Once we have all of the foods recorded we will go through the list and discuss further details about the foods, such as what time you ate the foods, how much of the food you ate, how these foods were prepared, and the location of the meals. Then we will go through the list again to check we haven't missed anything. After this assessment you will review the images that you have captured on the digital camera and remove any image you do not wish to disclose. Once this is complete we will view the images together that involve the consumption of food. I will ask you if there are any changes to the information you have already provided for each eating occasion".

No Change = - or NC Methods: MPR = 24h recall, RAD = Revue-assisted Time of Meal = 24hr Meal Code 1-6 (1 = breakfast, 2 = mid meal, 3 = Lunch, 4 = mid meal, 5 = dinner, 6 = Supper) Food or dish =food or dish eaten Brand= Brand or type of purchased food Quantity = quantity of food prepared or purchased % consumed = percentage of quantity consumed Cooking method = method used to prepare food or dish Portion Size Codes = X-X Additions = Additional items added such as sugar or sauces

PICTuRE FORM C – 24h Recall V1.1 24/04/2013

Method	Time of	Location	Meal Code	Food or dish	Brand/Type	Quantity	Wastage?	Portion size code	Additions
	Meal								
MPR									
RAD									

PICTuRE FORM C – 24h Recall V1.1 24/04/2013

Method	Time of	Location	Meal Code	Food or dish	Brand/Type	Quantity	Wastage?	Portion size code	Additions
	Meal								
MPR									
RAD									
PICTuRE I	FORM C-	24h Recall	V1.1 2	4/04/2013	UAHPEC REF NO: 8701				

Method	Time of Meal	Location	Meal Code	Food or dish	Brand/Type	Quantity	Wastage?	Portion size code	Additions
MPR									
RAD									
	CONTINUE								
	CONTINUE	D							

PICTuRE FORM C – 24h Recall V1.1 24/04/2013
REFERENCES

1. Thompson FE, Byers T. Dietary assessment resource manual. *J Nutr.* 1994;124(11):2245S-317S.

2. Thompson FE, Subar A. Dietary assessment methodology. In: Coulston A, Boushe C, eds. *Nutrition in the Prevention and Treatment of Disease*. 3rd ed. San Diego: Academic Press; 2013:3-30.

3. World Health Organization. 2008-2013 action plan for the global strategy for the pevention and control of noncommunicable diseases. Geneva: World Health Organization; 2008.

4. Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, et al. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. *The Lancet.* 2013;380(9859):2224-60.

5. Murray CJL, Vos T, Lozano R, Naghavi M, Flaxman AD, et al. Disability-adjusted life years (DALYs) for 291 diseases and injuries in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. *The Lancet*. 2013;380(9859):2197-223.

6. World Health Organization. Global status report on noncommunicable diseases 2010. Geneva: World Health Organization; 2011.

7. Margetts B, Gibney M, Margetts B, Kearney J, Arab L. *An overview of public health nutrition.* Oxford: Blackwell Publishing; 2004.

8. Livingstone MB, Black AE. Markers of the validity of reported energy intake. *J Nutr*. 2003;133 Suppl 3:895S-920S.

9. Poslusna K, Ruprich J, De Vries JHM, Jakubikova M, Van't Veer P. Misreporting of energy and micronutrient intake estimated by food records and 24hour recalls, control and adjustment methods in practice. *Br J Nutr.* 2009;101(SUPPL. 2):S73-S85.

10. Hill R, Davies P. The validity of self-reported energy intake as determined using the doubly labelled water technique. *Br J Nutr.* 2001;85(4):415-30.

11. Abegunde DO, Mathers CD, Adam T, Ortegon M, Strong K. The burden and costs of chronic diseases in low-income and middle-income countries. *Lancet*. 2007;370(9603):1929-38.

12. World Health Organization. The global burden of disease 2004 update. Geneva: World Health Organization; 2004.

13. World Health Organization. Action plan for the global strategy for the prevention and control of noncommunicable diseases. Geneva: World Health Organization; 2008.

14. World Health Organization. Diet, nurtition and the prevention of chronic diseases. Geneva: World Health Organization; 2003.

15. Thompson FE, Subar AF, Loria CM, Reedy JL, Baranowski T. Need for technological innovation in dietary assessment. *J Am Diet Assoc.* 2010;110(1):48-51.

16. Illner A, Freisling H, Boeing H, Huybrechts I, Crispim S, et al. Review and evaluation of innovative technologies for measuring diet in nutritional epidemiology. *Int J Epidemiol.* 2012;41(4):1187-203.

17. Willett W. Overview of Nutritional Epidemiology. *Nutritional Epidemiology*. Vol 30. 2nd ed: Oxford University Press, USA; 1998.

18. Kipnis V, Midthune D, Freedman LS, Bingham S, Schatzkin A, et al. Empirical evidence of correlated biases in dietary assessment instruments and its implications. *Am J Epidemiol.* 2001;153(4):394-403.

19. Probst YC, Tapsell LC. Overview of computerized dietary assessment programs for research and practice in nutrition education. *J Nutr Educ Behav*. 2005;37(1):20-6.

20. Subar AF, Kirkpatrick SI, Mittl B, Zimmerman TP, Thompson FE, et al. The automated self-administered 24-hour dietary recall (ASA24): a resource for researchers, clinicians, and educators from the national cancer institute. *Journal of the Academy of Nutrition and Dietetics*. 2012;112(8):1134-7.

21. Arab L, Wesseling-Perry K, Jardack P, Henry J, Winter A. Eight self-administered 24-hour dietary recalls using the Internet are feasible in African Americans and Whites: the energetics study. *J Am Diet Assoc.* 2010;110(6):857-64.

22. McBride J. Was is a slab, a slice, or a sliver? Agricultural Research Magazine. 2001 21/04/2011:4-7.

23. University of Otago and Ministry of Health. Methodology Report for the 2008/09 New Zealand Adult Nutrition Survey. Wellington: Ministry of Health; 2011.

24. Raper N, Perloff B, Ingwersen L, Steinfeldt L, Anand J. An overview of USDA's dietary intake data system. *J Food Compost Anal.* 2004;17(3-4):545-55.

 Australian Bureau of Statistics. 4364.0.55.007 - Australian Health Survey: Nutrition First Results - Foods and Nutrients, 2011-12

2014; http://www.abs.gov.au/AUSSTATS/abs@.nsf/Lookup/4364.0.55.007Main+Features12011-12?OpenDocument. Accessed 1st July, 2014.

26. Moshfegh AJ, Rhodes DG, Baer DJ, Murayi T, Clemens JC, et al. The US department of agriculture automated multiple-pass method reduces bias in the collection of energy intakes. *Am J Clin Nutr.* 2008;88(2):324-32.

27. Subar AF, Crafts J, Zimmerman T, Wilson M, Mittl B, et al. Assessment of the accuracy of portion size reports using computer-based food photographs aids in the development of an automated self-administered 24-hour recall. *J Am Diet Assoc.* 2010;110(1):55-64.

28. Maurer J, Taren DL, Teixeira PJ, Thomson CA, Lohman TG, et al. The psychosocial and behavioral characteristics related to energy misreporting. *Nutr Rev.* 2006;64(2):53-66.

29. Long JD, Littlefield LA, Estep G, Martin H, Rogers TJ, et al. Evidence review of technology and dietary assessment. *Worldviews on Evidence-Based Nursing*. 2010;7(4):191-204.

30. Ngo J, Engelen A, Molag M, Roesle J, Garcia-Segovia P, et al. A review of the use of information and communication technologies for dietary assessment. *Br J Nutr.* 2009;101 Suppl 2:S102-12.

31. Stumbo PJ. New technology in dietary assessment: a review of digital methods in improving food record accuracy. *Proc Nutr Soc.* 2013;72(1):70-6.

32. Berry E, Hampshire A, Rowe J, Hodges S, Kapur N, et al. The neural basis of effective memory therapy in a patient with limbic encephalitis. *J Neurol Neurosurg Psychiatry*. 2009;80(11):1202-5.

33. Berry E, Kapur N, Williams L, Hodges S, Watson P, et al. The use of a wearable camera, SenseCam, as a pictorial diary to improve autobiographical memory in a patient with limbic encephalitis: a preliminary report. *Neuropsychol Rehabi*. 2007;17(4-5):582-601.

34. Kipnis V, Subar AF, Midthune D, Freedman LS, Ballard-Barbash R, et al. Structure of dietary measurement error: results of the OPEN biomarker study. *Am J Epidemiol.* 2003;158(1):14-21.

35. Stellar E. Real eating and the measurement of real physiological and behavioral variables. *Appetite*. 1992;19(1):78-9.

36. Kissileff HR. Where should human eating be studied and what should be measured? *Appetite*. 1992;19(1):61-8.

37. Pliner P. Let's not throw out the barley with the dishwater: Comments on Meiselman's "Methodology and theory in human eating research". *Appetite*. 1992;19(1):74-5.

38. Tuorila H, Lähteenmäki L. When is eating "real"? response to meiselman. *Appetite*. 1992;19(1):80-3.

39. Rolls BJ, Shide DJ. Both naturalistic and laboratory-based studies contribute to the understanding of human eating behavior. *Appetite*. 1992;19(1):76-7.

40. Sallis JF, Owen N, Fotheringham MJ. Behavioral epidemiology: a systematic framework to classify phases of research on health promotion and disease prevention. *Ann Behav Med.* 2000;22(4):294-8.

41. Stroebele N, De Castro JM. Effect of ambience on food intake and food choice. *Nutrition*. 2004;20(9):821-38.

42. Meiselman H. The role of context in food choice, food acceptance and food comsumption In: Shepherd R, Raats M, eds. *The psychology of food choice*. Cambridge, MA: Cabi; 2006.

43. Beaglehole R, Bonita R, Horton R, Adams C, Alleyne G, et al. Priority actions for the noncommunicable disease crisis. *Lancet*. 2011;377(9775):1438-47.

44. Meiselman HL. Obstacles to studying real people eating real meals in real situations. *Appetite*. 1992;19(1):84-6.

45. Samet JM. Epidemiology and policy: the pump handle meets the new millennium. *Epidemiol Rev.* 2000;22(1):145-54.

46. World Health Organization. Health topics: Epidemiology. 2013; http://www.who.int/topics/epidemiology/en/. Accessed 23 March, 2013.

47. McNaughton S, Ball K, Crawford D. Research Skills. In: Lawrence MD, Worsley T, eds. *Public health nutrition : from principles to practice*. NSW, Australia: Allen & Unwin; 2007.

48. Lozano R, Naghavi M, Foreman K, Lim S, Shibuya K, et al. Global and regional mortality from 235 causes of death for 20 age groups in 1990 and 2010: a systematic analysis for the Global Burden of Disease Study 2010. *The Lancet*. 2013;380(9859):2095-128.

49. Statistics New Zealand. New Zealand Population Clock: Estimated Resident Population of New Zealand. 2014; http://www.stats.govt.nz/tools_and_services/population_clock.aspx. Accessed 3rd July, 2014.

50. Statistics New Zealand. Census of the population dwellings Wellington: Statistics New Zealand; 2006.

51. Statistics New Zealand. Statistics for New Zealand home; browse for stats;profile-andsummary-reports. http://www.stats.govt.nz/Census/2013-census/profile-and-summaryreports/quickstats-about-a-place.aspx?request_value=13067&tabname=Ageandsex. Accessed 3rd July, , 2014.

52. Organisation for Economic Co-operation and Development. OECD better life index. 2014; http://www.oecdbetterlifeindex.org/. Accessed 5th August, 2014.

53. World Health Organization. Noncommunicable diseases country profiles 2011. Geneva: World Health Organization; 2011.

54. Ministry of Health. Health loss in New Zealand: A report from the New Zealand Burden of Diseases, Injuries and Risk Factors Study, 2006–2016. Wellington: Ministry of Health; 2013.

55. Ministry of Health. Cancer: new registrations and deaths 2009. Wellington: Dept. of Health, Health Statistical Services; 2012. Report No.: 1171-6509

0548-9415

56. Chan W, Wright C, Riddell T, Wells S, Kerr AJ, et al. Ethnic and socioeconomic disparities in the prevalence of cardiovascular disease in New Zealand. *N Z Med J*. 2008;121(1285).

57. Wells L, Broad J, Jackson R. Estimated prevalence of cardiovascular disease and distribution of cardiovascular risk in New Zealanders: data for health care planners, funders and providers. *NZMJ*. 2006;119(1232):1-8.

58. Ministry of Health. The health of New Zealand adults 2011/12: key findings of the New Zealand Health Survey. Wellington: Ministry of Health; 2012.

59. Ng M, Fleming T, Robinson M, Thomson B, Graetz N, et al. Global, regional, and national prevalence of overweight and obesity in children and adults during 1980–2013: a systematic analysis for the Global Burden of Disease Study 2013. *The Lancet*. 2014.

60. Organisation for Economic Co-operation and Development. OECD Factbook 2014: Economic Environmental and Social Statistics; 2014.

Publishing O. http://dx.doi.org/10.1787/factbook-2014-en. 30th June, 2014.

61. Ministry of Health. A potrait of health. key results of the 2002/3 New Zealand Health Survey. Welllington: Ministry of Health; 2004.

62. Gentles D, Metcalf P, Dyall L, Scragg R, Black P, et al. Blood pressure prevalences and levels for a multicultural population in Auckland, New Zealand: results from the Diabetes, Heart and Health Survey 2002/2003. *NZ Med J*. 2006;119.

63. Gentles D, Metcalf P, Dyall L, Scragg R, Sundborn G, et al. Serum lipid levels for a multicultural population in Auckland, New Zealand: results from the Diabetes, Heart and Health Survey (DHAH) 2002-2003. *NZ Med J*. 2007;120.

64. Townsend N, Wickramasinghe K, Bhatnagar P, Smolina K, Nichols M, et al. Coronary heart disease statistics 2012 edition. London: British Heart Foundation; 2012.

65. University of Otago and Ministry of Health. A Focus on Nutrition: Key findings from the 2008/09 NZ Adult Nutrition Survey. Wellington: Ministry of Health; 2011.

66. Australian Government and the New Zealand Ministry of Health. Nutirent reference values for New Zealand and Australia: including recommended dietary intakes Wellington: Ministry of Health; 2006.

67. Willett WC, Howe GR, Kushi LH. Adjustment for total energy intake in epidemiologic studies. *The American journal of clinical nutrition*. 1997;65(4):1220S-8S.

68. Food and Agriculture Organization of the United Nations. Energy and Protein Requirements: Report of a Joint FAO/WHO/UNU Expert Consultation. Geneva: World Health Organization; 1985. Report No.: 9241207248: 724

69. Klesges RC, Eck LH, Ray JW. Who underreports dietary intake in a dietary recall? Evidence from the Second National Health and Nutrition Examination Survey. *J Consult Clin Psychol.* 1995;63(3):438-44.

70. Briefel R, Sempos CT, McDowell M, Chien S, Alaimo K. Dietary methods research in the third National Health and Nutrition Examination Survey: underreporting of energy intake. *Am J Clin Nutr.* 1997;65(4):1203S-9S.

71. Heywood P, Harvey P, Marks GC. An evaluation of energy intakes in the 1983 Australian National Dietary Survey of Adults. *Eur J Clin Nutr*. 1993;47:604-6.

72. Rennie KL, Coward A, Jebb SA. Estimating under-reporting of energy intake in dietary surveys using an individualised method. *Br J Nutr*. 2007;97(06):1169-76.

73. Pikholz C, Swinburn B, Metcalf P. Under-reporting of energy intake in the 1997 National Nutrition Survey. *N Z Med J*. 2004;117(1202):1-11.

74. Russell DG, Parnell WR, Wilson NC, et al. NZ Food: NZ People. Key Results of the 1997 National Nutrition Survey: Ministry of Health; 1999.

75. Thomson CD, Colls AJ. Twenty-four hour urinary sodium excretion in seven hundred residents of Otago and Waikato Report prepared for the Ministry of Health Dunedin: Department of Human Nutrition at the University of Otago; 1998.

76. Ministry of Health. A potrait of health. key results of the 2006/07 New Zealand health survey. Welllington: Ministry of Health; 2008.

77. Ministry of Health. Food and nutrition guidelines for healthy adults: a background paper. Wellington: Ministry of Health; 2003.

78. Wang YC, McPherson K, Marsh T, Gortmaker SL, Brown M. Health and economic burden of the projected obesity trends in the USA and the UK. *The Lancet.* 2011;378(9793):815-25.

79. Ministry of Health. Report on New Zealand cost-of-illness studies on long term conditions. Wellington: Ministry of Health; 2009.

80. Russell LB, Gold MR, Siegel JE, Daniels N, Weinstein MC. The role of cost-effectiveness analysis in health and medicine. *JAMA*. 1996;276(14):1172-7.

81. Lal A, Moodie M, Ashton T, Siahpush M, Swinburn B. Health care and lost productivity costs of overweight and obesity in New Zealand. *Aust N Z J Public Health*. 2012;36(6):550-6.

82. New Zealand Government. Government Response to the Inquiry into Obesity and Type 2 Diabetes 2008. Presented to the House of Representatives in accordance with Standing Order 253; 2008.

83.Diabetes New Zealand and PriceWaterhouseCoopers. Type 2 Diabetes – OutcomesModelUpdate;2008.September2008.https://www.diabetes.org.nz/resources/research_and_reports. Accessed 11th September 2014.

84. Gibson RS. *Principles of nutritional assessment*. Oxford University Press; 2005.

85. Department of Health. Home; Statistics; National Diet and Nutrition; National Diet and Nutrition; Appendix E: Dairy documents 2009;

http://media.dh.gov.uk/network/261/files/2012/07/Appendix-E_All-docs-combined-v2.pdf. Accessed 8th April, 2013.

86. United States Department of Agriculture. United States Department of Agriculture Research Service; Food Surveys: USDA Automated Multiple-Pass Method. 2013. Accessed 2nd April, 2013.

87. De Castro JM, Brewer EM. The amount eaten in meals by humans is a power function of the number of people present. *Physiol Behav*. 1992;51(1):121-5.

88. De Castro JM, de Castro ES. Spontaneous meal patterns of humans: influence of the presence of other people. *The American journal of clinical nutrition*. 1989;50(2):237-47.

89. Oxford Dictionaries. 2013; http://www.xforddictionaries.com. Accessed 8th April, 2013.

90. Mak TN, Prynne CJ, Cole D, Fitt E, Roberts C, et al. Assessing eating context and fruit and vegetable consumption in children: new methods using food diaries in the UK National Diet and Nutrition Survey Rolling Programme. *Int J Behav Nutr Phys Act.* 2012;9(1):1-15.

91. Stephen AM, Mak TN, Fitt E, Nicholson S, Roberts C, et al. Innovations in national nutrition surveys. *Proc Nutr Soc*. 2013;72(01):77-88.

92. Slimani N, Deharveng G, Charrondiere RU, van Kappel AL, Ocke MC, et al. Structure of the standardized computerized 24-h diet recall interview used as reference method in the 22 centers participating in the EPIC project. European Prospective Investigation into Cancer and Nutrition. *Comput Methods Programs Biomed.* 1999;58(3):251-66.

93. Department for Health and Ageing. User Guide 2007 Australian National Children's Nutrition and Physical Activity Survey: Department for Health and Ageing; 2007.

94. Ministry of Health. Food Comes First: Methodologies for the National Nurtition Survey of New Zealand. Wellington, New Zealand; 1997.

95. Basiotis PP, Welsh SO, Cronin FJ, Kelsay JL, Mertz W. Number of days of food intake records required to estimate individual and group nutrient intakes with defined confidence. *J Nutr*. 1987;117(9):1638-41.

96. Gibson RS. Measurement Errors in Dietay Assessment. *Principles of Nutrition Assessment* 2nd ed. United States of America: Oxford University Press; 2005.

97. Gemming L, Jiang Y, Swinburn B, Utter J, Mhurchu CN. Under-reporting remains a key limitation of self-reported dietary intake: an analysis of the 2008/09 New Zealand Adult Nutrition Survey. *Eur J Clin Nutr.* 2014;68(2):259-64.

98. Department of Health. National Diet and Nutrition Survey Headline results from Years 1,2 and 3 (combined) of the Rolling Programme (2008/2009 – 2010/11): Department of Health; 2012.

99. Beasley J, Riley WT, Jean-Mary J. Accuracy of a PDA-based dietary assessment program. *Nutrition*. 2005;21(6):672-7.

100. Rusin M, Årsand E, Hartvigsen G. Functionalities and input methods for recording food intake: A systematic review. *Int J Med Inf.* 2013.

101. Beasley JM, Riley WT, Davis A, Singh J. Evaluation of a PDA-based dietary assessment and intervention program: A randomized controlled trial. *J Am Coll Nutr.* 2008;27(2):280-6.

102. Gersovitz M, Madden JP, Smiciklas-Wright H. Validity of the 24-hr. dietary recall and seven-day record for group comparisons. *J Am Diet Assoc.* 1978;73(1):48-55.

103. Fukuo W, Yoshiuchi K, Ohashi K, Togashi H, Sekine R, et al. Development of a handheld personal digital assistant-based food diary with food photographs for japanese subjects. *J Am Diet Assoc.* 2009;109(7):1232-6.

104. Rebro SM, Patterson RE, Kristal AR, Cheney CL. The effect of keeping food records on eating patterns. *J Am Diet Assoc.* 1998;98(10):1163. List of research project topics and materials

105. Harnack LJ, Lytle LA, Story M, Galuska DA, Schmitz K, et al. Reliability and validity of a brief questionnaire to assess calcium intake of middle-school-aged children. *J Am Diet Assoc.* 2006;106(11):1790-5.

106. Block G, Gillespie C, Rosenbaum EH, Jenson C. A rapid food screener to assess fat and fruit and vegetable intake. *Am J Prev Med.* 2000;18(4):284-8.

107. Subar AF, Thompson FE, Kipnis V, Midthune D, Hurwitz P, et al. Comparative validation of the Block, Willett, and National Cancer Institute food frequency questionnaires. *Am J Epidemiol.* 2001;154(12):1089.

108. McDonald A, Van Horn L, Slattery M, Hilner J, Bragg C, et al. The CARDIA dietary history: development, implementation, and evaluation. *J Am Diet Assoc*. 1991;91(9):1104.

109. Van Staveren W, De Boer J, Burema J. Validity and reproducibility of a dietary history method estimating the usual food intake during one month. *The American Journal of Clinical Nutrition*. 1985;42(3):554.

110. Hartman AM, Block G, Chan W, Williams J, McAdams M, et al. Reproducibility of a selfadministered diet history questionnaire administered three times over three different seasons. *Nutr Cancer.* 1996;25(3):305-15.

111. Probst YC, De Agnoli K, Batterham M, Tapsell L. Video-recorded participant behaviours: the association between food choices and observed behaviours from a web-based diet history interview. *J Hum Nutr Diet*. 2009;22(1):21-8.

112. Foster E, Hawkins A, Adamson A. Development and validation of the Young Person's Food Atlas. *Proc Nutr Soc.* 2012;71(OCE2).

113. Nelson M, Haraldsdóttir J. Food photographs: practical guidelines II. Development and use of photographic atlases for assessing food portion size. *Public Health Nutr.* 1998;1(04):231-7.

114. Block G. A review of validations of dietary assessment methods. *Am J Epidemiol*. 1982;115(4):492-505.

115. Kuskowska-Wolk A, Holte S, Ohlander EM, Bruce Å, Holmberg L, et al. Effects of different designs and extension of a food frequency questionnaire on response rate, completeness of data and food frequency responses. *Int J Epidemiol.* 1992;21(6):1144-50.

116. Subar AF, Ziegler RG, Thompson FE, Johnson CC, Weissfeld JL, et al. Is shorter always better? Relative importance of questionnaire length and cognitive ease on response rates and data quality for two dietary questionnaires. *Am J Epidemiol*. 2001;153(4):404-9.

117. Centres for Disease and Control Prevention. Natioanl Health and Nutrition Examination Survey, MEC In-Person dietary interviewers procedures manual; 2014.

118. Young LR, Nestle M. Portion sizes in dietary assessment: issues and policy implications. *Nutr Rev.* 1995;53(6):149-58.

119. Nelson M, Atkinson M, Darbyshire S. Food photography. I: The perception of food portion size from photographs. *Br J Nutr*. 1994;72(5):649-63.

120. Greenfield H, Southgate DA. *Food Composition Data: Production, Management, and Use.* Food & Agriculture Organization of the UN (FAO); 2003.

121. Atwater WO, Woods CD. The chemical composition of American food materials. *Experimental Stations Bulletin* 28. 1896;(28).

122. New Zealand Food Composition Database [database on the Internet]. 2011 [cited 14/03/2013]. Available from: http://www.foodcomposition.co.nz/foodfiles.

123. Composition of Foods Integrated Dataset (CoFIDS) [database on the Internet]. 2002 [cited 14/03/2013]. Available from: http://www.ifr.ac.uk/fooddatabanks/nutrients.htm#CoFIDS.

124. Davis DR, Epp MD, Riordan HD. Changes in USDA food composition data for 43 garden crops, 1950 to 1999. *J Am Coll Nutr.* 2004;23(6):669-82.

125. Williamson C. The different uses of food composition databases. Synthesis report No 2. Nowich, Norfolk: European Food Information Resource Consortium (EuroFIR); 2006.

126. Block G, Hartman AM. Issues in reproducibility and validity of dietary studies. *The American journal of clinical nutrition*. 1989;50(5):1133-8.

127. Gleason PM, Harris J, Sheean PM, Boushey CJ, Bruemmer B. Publishing nutrition research: validity, reliability, and diagnostic test assessment in nutrition-related research. *J Am Diet Assoc.* 2010;110(3):409-19.

128. Bingham SA. Biomarkers in nutritional epidemiology. *Public Health Nutr.* 2002;5(6A):821-8.

129. Bingham SA. Urine nitrogen as a biomarker for the validation of dietary protein intake. *J Nutr.* 2003;133(3):921S.

130. Bingham SA, Cummings JH. Urine nitrogen as an independent validatory measure of dietary intake: a study of nitrogen balance in individuals consuming their normal diet. *The American journal of clinical nutrition*. 1985;42(6):1276-89.

131. Subar AF, Kipnis V, Troiano RP, Midthune D, Schoeller DA, et al. Using intake biomarkers to evaluate the extent of dietary misreporting in a large sample of adults: the OPEN study. *Am J Epidemiol*. 2003;158(1):1-13.

132. Willett W. Biochemical Indicators of Dietary Intake. *Nutritional Epidemiology*. 2nd ed: Oxford University Press, USA.

133. Prentice A. Doubly labeled water method for measuring energy expenditure: technical recommendations for use in humans. Vienna, Austria: International Atomic Energy Agency; 1990.

134. Westerterp K, Saris W, Van Es M, Ten Hoor F. Use of the doubly labeled water technique in humans during heavy sustained exercise. *J Appl Physiol*. 1986;61(6):2162-7.

135. Schoeller DA. Recent advances from application of doubly labeled water to measurement of human energy expenditure. *J Nutr*. 1999;129(10):1765-8.

136. Schoeller DA, Ravussin E, Schutz Y, Acheson KJ, Baertschi P, et al. Energy expenditure by doubly labeled water: validation in humans and proposed calculation. *Am J Physiol Regul Integr Comp Physiol*. 1986;250(5):R823-R30.

137. Schoeller DA, Bandini LG, Dietz WH. Inaccuracies in self-reported intake identified by comparison with the doubly labelled water method. *Can J Physiol Pharmacol.* 1990;68(7):941-9.

138. Lifson LA, McCintock R. Theory of use of the turnover rates of body water for measuring energy and material balance. *J Theor Biol.* 1966;12(1):46-74.

139. Schoeller D, Van Santen E. Measurement of energy expenditure in humans by doubly labeled water method. *J Appl Physiol.* 1982;53(4):955-9.

140. Schoeller DA. Measurement of energy expenditure in free-living humans by using doubly labeled water. *J Nutr.* 1988;118(11):1278-89.

141. Seale JL, Conway JM, Canary JJ. Seven-day validation of doubly labeled water method using indirect room calorimetry. *J Appl Physiol*. 1993;74(1):402-9.

142. Seale JL. Energy expenditure measurements in relation to energy requirements. *Am J Clin Nutr.* 1995;62(5):1042S-6S.

143. Trabulsi J, Schoeller DA. Evaluation of dietary assessment instruments against doubly labeled water, a biomarker of habitual energy intake. *Am J Physiol Endocrinol Metab.* 2001;281(5):E891-9.

144. Black AE. The sensitivity and specificity of the Goldberg cut-off for EI:BMR for identifying diet reports of poor validity. *Eur J Clin Nutr*. 2000;54(5):395-404.

145. Johansson L, Solvoll K, Bjørneboe G-E, Drevon CA. Under-and overreporting of energy intake related to weight status and lifestyle in a nationwide sample. *Am J Clin Nutr*. 1998;68(2):266-74.

146. Tomoyasu NJ, Toth MJ, Poehlman ET. Misreporting of total energy intake in older men and women. *J Am Geriatr Soc*. 1999;47(6):710-5.

147. Price G, Paul A, Cole T, Wadsworth M. Characteristics of the low-energy reporters in a longitudinal national dietary survey. *Br J Nutr*. 1997;77(06):833-51.

148. Asbeck I, Mast M, Bierwag A, Westenhofer J, Acheson K, et al. Severe underreporting of energy intake in normal weight subjects: use of an appropriate standard and relation to restrained eating. *Public Health Nutr.* 2002;5(5):683-90.

149. Hebert JR, Ebbeling CB, Matthews CE, Hurley TG, Ma Y, et al. Systematic errors in middle-aged women's estimates of energy intake: comparing three self-report measures to total energy expenditure from doubly labeled water. *Ann Epidemiol.* 2002;12(8):577-86.

150. Tomoyasu NJ, Toth MJ, Poehlman ET. Misreporting of total energy intake in older African Americans. *Int J Obes Relat Metab Disord*. 2000;24(1):20.

151. Harrison GG, Galal OM, Ibrahim N, Khorshid A, Stormer A, et al. Underreporting of food intake by dietary recall is not universal: a comparison of data from egyptian and american women. *J Nutr.* 2000;130(8):2049-54.

152. Bandini LG, Schoeller DA, Cyr HN, Dietz WH. Validity of reported energy intake in obese and nonobese adolescents. *Am J Clin Nutr.* 1990;52(3):421-5.

153. Bandini LG, Vu D, Must A, Cyr H, Goldberg A, et al. Comparison of high-calorie, lownutrient-dense food consumption among obese and non-obese adolescents. *Obes Res.* 1999;7(5):438-43.

154. Lichtman SW, Pisarska K, Berman ER, Pestone M, Dowling H, et al. Discrepancy between self-reported and actual caloric intake and exercise in obese subjects. *N Engl J Med.* 1992;327(27):1893-8.

155. Heerstrass D, Ocke M, Bueno-de-Mesquita H, Peeters P, Seidall J. Underreporting of energy, protein and potassium intake in relation to body mass index. *Int J Epidemiol.* 1998;27(2):186-93.

156. Edwards JE, Lindeman AK, Mikesky AE, Stager JM. Energy balance in highly trained female endurance runners. *Med Sci Sports Exerc*. 1993;25(12):1398.

157. Barnard J, Tapsell L, Davies P, Brenninger V, Storlien L. Relationship of high energy expenditure and variation in dietary intake with reporting accuracy on 7 day food records and diet histories in a group of healthy adult volunteers. *Eur J Clin Nutr.* 2002;56(4):358-67.

158. Hassapidou M, Manstrantoni A. Dietary intakes of elite female athletes in Greece. *J Hum Nutr Diet.* 2001;14(5):391-6.

159. Rothman KJ, Greenland S, Lash TL. *Modern epidemiology.* 3rd ed: Lippincott Williams & Wilkins; 2008.

160. Kipnis V, Carroll RJ, Freedman LS, Li L. Implications of a new dietary measurement error model for estimation of relative risk: application to four calibration studies. *Am J Epidemiol.* 1999;150(6):642-51.

161. Paeratakul S, Popkin BM, Kohlmeier L, Hertz-Picciotto I, Guo X, et al. Measurement error in dietary data: implications for the epidemiologic study of the diet-disease relationship. *Eur J Clin Nutr.* 1998;52(10):722-7.

162. Smith WT, Webb KL, Heywood PF. The implications of underreporting in dietary studies. *Aust J Public Health*. 1994;18(3):311-4.

163. Becker W, Welten D. Under-reporting in dietary surveys–implications for development of food-based dietary guidelines. *Public Health Nutr.* 2001;4(2b):683-7.

164. Willett W. Correction for the Effects of Measurement Error. *Nutritional epidemiology.* Vol 30. 2nd ed: Oxford University Press, USA; 1998.

165. Willett W. Implications of Total Energy Intake for Epidemilogical Analysis. *Nutritional Epidemiology*. Vol 30. 2nd ed: Oxford University Press, USA; 1998.

166. Beaton GH, Milner J, Corey P, McGuire V, Cousins M, et al. Sources of variance in 24hour dietary recall data: implications for nutrition study design and interpretation. *Am J Clin Nutr*. 1979;32(12):2546-59.

167. Beaton GH, Milner J, McGuire V, Feather TE, Little JA. Source of variance in 24-hour dietary recall data: implications for nutrition study design and interpretation. Carbohydrate sources, vitamins, and minerals. *Am J Clin Nutr.* 1983;37(6):986-95.

168. Bhargava A, Forthofer R, McPherson S, Nichaman M. Estimating the variations and autocorrelations in dietary intakes on weekdays and weekends. *Stat Med.* 1994;13(2):113-26.

169. Borrelli R, Cole TJ, Di Biase G, Contaldo F. Some statistical considerations on dietary assessment methods. *Eur J Clin Nutr.* 1989;43(7):453-63.

170. Kaaks R, Riboli E, Estève J, Van Kappel AL, Van Staveren WA. Estimating the accuracy of dietary questionnaire assessments: validation in terms of structural equation models. *Stat Med.* 1994;13(2):127-42.

171. Kim WW, Kelsay JL, Judd JT, Marshall MW, Mertz W, et al. Evaluation of long-term dietary intakes of adults consuming self-selected diets. *Am J Clin Nutr.* 1984;40(6):1327-32.

172. Liu K, Stamler J, Dyer A, McKeever J, McKeever P. Statistical methods to assess and minimize the role of intra-individual variability in obscuring the relationship between dietary lipids and serum cholesterol. *J Chronic Dis.* 1978;31(6):399-418.

173. Marr JW, Heady JA. Within-and between-person variation in dietary surveys: number of days needed to classify individuals. *Hum Nutr Appl Nutr*. 1986;40(5):347-64.

174. Plummer M, Clayton D. Measurement error in dietary assessment: an investigation using covariance structure models. Part I. *Stat Med.* 1993;12(10):925-35.

175. Rosner B, Willett WC. Interval estimates for correlation coefficients corrected for withinperson variation: implications for study design and hypothesis testing. *Am J Epidemiol*. 1988;127(2):377-86.

176. Sempos CT, Johnson NE, Smith EL, Gilligan C. Effects of intraindividual and interindividual variation in repeated dietary records. *Am J Epidemiol*. 1985;121(1):120-30.

177. Stallone DD, Brunner EJ, Bingham SA, Marmot MG. Dietary assessment in Whitehall II: the influence of reporting bias on apparent socioeconomic variation in nutrient intakes. *Eur J Clin Nutr.* 1997;51(12):815-25.

178. Howe GR, Miller AB, Jain M. Re:"Total energy intake: implications for epidemiologic analyses". *Am J Epidemiol*. 1986;124(1):157-9.

179. Pike MC, Bernstein L, Peters RK. Re:"Total energy intake: implications for epidemiologic analyses". *Am J Epidemiol*. 1989;129(6):1312-3.

180. Kipnis V, Freedman LS, Brown CC, Hartman A, Schatzkin A, et al. Interpretation of energy adjustment models for nutritional epidemiology. *Am J Epidemiol*. 1993;137(12):1376-80.

181. Brown CC, Kipnis V, Freedman LS, Hartman AM, Schatzkm A, et al. Energy adjustment methods for nutritional epidemiology: the effect of categorization. *Am J Epidemiol*. 1994;139(3):323-38.

182. Wacholder S, Schatzkin A, Freedman LS, Kipnis V, Hartman A, et al. Can energy adjustment separate the effects of energy from those of specific macronutrients? *Am J Epidemiol.* 1994;140(9):848-55.

183. Mackerras D. Energy adjustment: the concepts underlying the debate. *J Clin Epidemiol*. 1996;49(9):957-62.

184. Goldberg G, Black A, Jebb S, Cole T, Murgatroyd P, et al. Critical evaluation of energy intake data using fundamental principles of energy physiology: 1. Derivation of cut-off limits to identify under-recording. *Eur J Clin Nutr.* 1991;45(12):569.

185. Black AE. Critical evaluation of energy intake using the Goldberg cut-off for energy intake: basal metabolic rate. A practical guide to its calculation, use and limitations. *Int J Obes Relat Metab Disord*. 2000;24(9):1119.

186. Bogardus C, Lillioja S, Ravussin E, Abbott W, Zawadzki JK, et al. Familial dependence of the resting metabolic rate. *N Engl J Med.* 1986;315(2):96-100.

187. Ravussin E, Lillioja S, Anderson TE, Christin L, Bogardus C. Determinants of 24-hour energy expenditure in man. Methods and results using a respiratory chamber. *J Clin Invest*. 1986;78(6):1568.

188. Diaz EO, Prentice AM, Goldberg GR, Murgatroyd PR, Coward WA. Metabolic response to experimental overfeeding in lean and overweight healthy volunteers. *Am J Clin Nutr*. 1992;56(4):641-55.

189. Hall KD. Predicting metabolic adaptation, body weight change, and energy intake in humans. *Am J Physiol Endocrinol Metab.* 2010;298(3):E449-E66.

190. World Health Organization. Global Database on Body Mass Index. Geneva: World Health Organization; 2007.

191. Cole TJ, Bellizzi MC, Flegal KM, Dietz WH. Establishing a standard definition for child overweight and obesity worldwide: international survey. *BMJ*. 2000;320(7244):1240.

192. Cole TJ, Flegal KM, Nicholls D, Jackson AA. Body mass index cut offs to define thinness in children and adolescents: international survey. *BMJ*. 2007;335(7612):194.

193. Swinburn B. Using the body mass index: weight then weigh up. *N Z Med J*. 1998;111:377-9.

194. Swinburn B, Ley S, Carmichael H, Plank L. Body size and composition in Polynesians. *Int J Obes Relat Metab Disord*. 1999;23(11):1178.

195. World Health Organization. Obesity: prevention and managing the global epidemic. Report of a WHO Consultation. Geneva: World Health Organization; 2000.

196. Schofield W, Schofield C, James WPT. Basal metabolic rate: review and prediction, together with an annotated bibliography of source material. *Hum Nutr: Clin Nutr.* 1985;39C:(Suppl 1) 5–96.

197. Johnson RK, Soultanakis RP, Matthews DE. Literacy and body fatness are associated with underreporting of energy intake in US low-income women using the multiple-pass 24-hour recall: a doubly labeled water study. *J Am Diet Assoc*. 1998;98(10):1136-40.

198. Hall KD, Sacks G, Chandramohan D, Chow CC, Wang YC, et al. Quantification of the effect of energy imbalance on bodyweight. *The Lancet*. 2011;378(9793):826-37.

199. Grabe S, Ward LM, Hyde JS. The role of the media in body image concerns among women: A meta-analysis of experimental and correlational studies. *Psychol Bull*. 2008;134(3):460.

200. Kubic KN, Chory RM. Exposure to television makeover programs and perceptions of self. *Communication Research Reports*. 2007;24(4):283-91.

201. Ministry of Health. Healthy Eating – Healthy Action Oranga Kai – Oranga Pumau: A strategic framework. Wellington: Ministry of Health; 2003.

202. The official website of the New Zealand Government. Home; Releases; Mission-On: healthy lifestyles for young Kiwi. 2007; http://www.beehive.govt.nz/?q=node/27181. Accessed 18th March, 2013.

203. Bauman A, McLean G, Hurdle D, Walker S, Boyd J, et al. Evaluation of the national 'Push Play'campaign in New Zealand–creating population awareness of physical activity. *N Z Med J*. 2003;116(1179):1-11.

204. O'Loughlin G, Cullen S, McGoldrick A, Warrington G, editors. A comparison and validation of dietary analysis using a standard one day food diary and the sensecam in trainee jockeys. SenseCam 2012 Third Symposium; April 3-4, 2012; Oxford, United Kingdom.

205. Sun M, Fernstrom JD, Jia W, Hackworth SA, Yao N, et al. A wearable electronic system for objective dietary assessment. *J Am Diet Assoc.* 2010;110(1):45-7.

206. Black A, Goldberg G, Jebb S, Livingstone M, Cole T, et al. Critical evaluation of energy intake data using fundamental principles of energy physiology: 2. Evaluating the results of published surveys. *Eur J Clin Nutr.* 1991;45(12):583.

207. Ravussin E, Bogardus C. A brief overview of human energy metabolism and its relationship to essential obesity. *Am J Clin Nutr.* 1992;55(1):242S-5S.

208. Weyer C, Snitker S, Rising R, Bogardus C, Ravussin E. Determinants of energy expenditure and fuel utilization in man: effects of body composition, age, sex, ethnicity and glucose tolerance in 916 subjects. *Int J Obes Relat Metab Disord*. 1999;23(7):715.

209. Martin CK, Correa JB, Han H, Allen HR, Rood JC, et al. Validity of the remote food photography method (RFPM) for estimating energy and nutrient intake in near real-time. *Obesity* (*Silver Spring*). 2012;20(4):891-9.

210. Gregory R, Walwyn L, Bloor S, Amin S. A feasibility study of the use of photographic food diaries in the management of obesity. *Practical Diabetes International*. 2006;23(2):66-8.

211. Kikunaga S, Tin T, Ishibashi G, Wang DH, Kira S. The application of a handheld personal digital assistant with camera and mobile phone card (wellnavi) to the general population in a dietary survey. *J Nutr Sci Vitaminol (Tokyo)*. 2007;53(2):109-16.

212. Arab L, Estrin D, Kim DH, Burke J, Goldman J. Feasibility testing of an automated imagecapture method to aid dietary recall. *Eur J Clin Nutr.* 2011;65(10):1156-62.

213. Gemming L, Doherty A, Kelly P, Utter J, Ni Mhurchu C. Feasibility of a SenseCamassisted 24-h recall to reduce under-reporting of energy intake. *Eur J Clin Nutr.* 2013;67(10):1095-9.

214. Bai Y, Li C, Yue Y, Jia W, Li J, et al., editors. Designing a wearable computer for lifestyle evaluation. Bioengineering Conference (NEBEC), 2012 38th Annual Northeast; March 16-18, 2012. IEEE:93-4. http://ieeexplore.ieee.org/xpl/articleDetails.jsp?arnumber=6206978. Accessed July 8, 2014.

215. McCabe-Sellers B. Advancing the art and science of dietary assessment through technology. *J Am Diet Assoc.* 2010;110(1):52-4.

216. Buhl KM, Gallagher D, Hoy K, Matthews DE, Heymsfield SB. Unexplained disturbance in body weight regulation: diagnostic outcome assessed by doubly labeled water and body composition analyses in obese patients reporting low energy intakes. *J Am Diet Assoc.* 1995;95(12):1393-400.

217. Prentice AM, Black A, Coward W, Davies H, Goldberg G, et al. High levels of energy expenditure in obese women. *BMJ*. 1986;292(6526):983-7.

218. Goris AHC, Westerterp-Plantenga MS, Westerterp KR. Undereating and underrecording of habitual food intake in obese men: selective underreporting of fat intake. *Am J Clin Nutr*. 2000;71(1):130-4.

219. Black AE, Jebb SA, Bingham SA, Runswick SA, Poppitt SD. The validation of energy and protein intakes by doubly labelled water and 24-hour urinary nitrogen excretion in post-obese subjects. *J Hum Nutr Diet*. 1995;8(1):51-64.

220. Khanna N, Boushey CJ, Kerr D, Okos M, Ebert DS, et al., editors. An overview of the technology assisted dietary assessment project at purdue university. ISM : IEEE International Symposium on Multimedia : proceedings IEEE International Symposium on Multimedia; December 13-15, 2010. PMC3183748:290-5. http://ieeexplore.ieee.org/xpls/abs_all.jsp?arnumber=5693855. Accessed July 8, 2014.

221. Sun M, Fernstrom JD, Jia W, Hackworth SA, Yao N, et al. A wearable electronic system for objective dietary assessment. *J Am Diet Assoc*. 2010;110(1):45-7.

222. Martin CK, Han H, Coulon SM, Allen HR, Champagne CM, et al. A novel method to remotely measure food intake of free-living individuals in real time: the remote food photography method. *Br J Nutr*. 2009;101(3):446-56.

223. Wang DH, Kogashiwa M, Ohta S, Kira S. Validity and reliability of a dietary assessment method: The application of a digital camera with a mobile phone card attachment. *J Nutr Sci Vitaminol (Tokyo)*. 2002;48(6):498-504.

224. Rollo ME, Ash S, Lyons-Wall P, Russell A. Trial of a mobile phone method for recording dietary intake in adults with type 2 diabetes: evaluation and implications for future applications. *J Telemed Telecare*. 2011;17(6):318-23.

225. Illner AK, Freisling H, Boeing H, Huybrechts I, Crispim SP, et al. Review and evaluation of innovative technologies for measuring diet in nutritional epidemiology. *Int J Epidemiol.* 2012;41(4):1187-203.

226. Wang DH, Kogashiwa M, Kira S. Development of a new instrument for evaluating individuals' dietary intakes. *J Am Diet Assoc.* 2006;106(10):1588-93.

227. Lassen AD, Poulsen S, Ernst L, Andersen KK, Biltoft-Jensen A, et al. Evaluation of a digital method to assess evening meal intake in a free-living adult population. *Food Nutr Res.* 2010;54(5311):1-9.

228. Schap TE. Advancing dietary assessment through technology and biomarkers [PhD thesis]. Ann Arbor, Purdue University; 2012.

229. Lazarte CE, Encinas ME, Alegre C, Granfeldt Y. Validation of digital photographs, as a tool in 24-h recall, for the improvement of dietary assessment among rural populations in developing countries. *Nutrition Journal*. 2012;11(61):1-14.

230. O'Loughlin G, Cullen SJ, McGoldrick A, O'Connor S, Blain R, et al. Using a wearable camera to increase the accuracy of dietary analysis. *Am J Prev Med.* 2013;44(3):297-301.

Weiss R, Stumbo PJ, Divakaran A. Automatic food documentation and volume computation using digital imaging and electronic transmission. *J Am Diet Assoc*. 2010;110(1):42-4.

232. Shang J, Pepin E, Johnson E, Hazel D, Teredesai A, et al., editors. Dietary intake assessment using integrated sensors and software. IS&T/SPIE Electronic Imaging; 22-26 January, 2012. International Society for Optics and Photonics:830403--11. http://proceedings.spiedigitallibrary.org/proceeding.aspx?articleid=1284602. Accessed July 8, 2014.

233. Williamson DA, Allen HR, Martin PD, Alfonso AJ, Gerald B, et al. Comparison of digital photography to weighed and visual estimation of portion sizes. *J Am Diet Assoc*. 2003;103(9):1139-45.

234. Williamson D, Allen H, Martin P, Alfonso A, Gerald B, et al. Digital photography: a new method for estimating food intake in cafeteria settings. *Eat Weight Disord*. 2004;9(1):24.

235. Martin CK, Kaya S, Gunturk BK, editors. Quantification of food intake using food image analysis. Engineering in Medicine and Biology Society, 2009 EMBC 2009 Annual International Conference of the IEEE; September 3-6, 2009. IEEE:6869-72. http://ieeexplore.ieee.org/xpl/articleDetails.jsp?arnumber=5333123. Accessed July 8, 2014.

236. US Department of Agriculture. USDA Food and Nutrient Database for Dietary Studies 3.0. Beltsville, Maryland; 2008.http://www.ars.usda.gov/Services/docs.htm?docid=12089. Accessed Arpil 17, 2014.

237. Zhu F. *Multilevel image segmentation with application in dietary assessment and evaluation* [PhD thesis]. Ann Arbor, Purdue University; 2011.

238. Zhu F, Bosch M, Schap T, Khanna N, Ebert DS, et al., editors. Segmentation Assisted Food Classification for Dietary Assessment. Proceedings of SPIE; February 7, 2011. PMC3224860:78730B. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3224860/. Accessed July 8, 2014.

239. Zhu F, Bosch M, Woo I, Kim S, Boushey CJ, et al., editors. The use of mobile devices in aiding dietary assessment and evaluation. IEEE J Sel Top Signal Process; September 4-6, 2010. PMC2941896:756-66. http://ieeexplore.ieee.org/xpls/abs_all.jsp?arnumber=6046629. Accessed July 8, 2014.

240. Zhu F, Bosch M, Boushey CJ, Delp EJ. An image analysis system for dietary assessment and evaluation. *Proceedings / ICIP International Conference on Image Processing*. 2010:1853-6.

241. Six BL, Schap TE, Zhu FM, Mariappan A, Bosch M, et al. Evidence-based development of a mobile telephone food record. *J Am Diet Assoc.* 2010;110(1):74-9.

242. Daugherty BL, Schap TE, Ettienne-Gittens R, Zhu FM, Bosch M, et al. Novel technologies for assessing dietary intake: Evaluating the usability of a mobile telephone food record among adults and adolescents. *JMIR*. 2012;14(2):e58.

243. Hodges S, Williams L, Berry E, Izadi S, Srinivasan J, et al. SenseCam: A retrospective memory aid. In: Dourish P, Friday A, editors. UbiComp 2006: Ubiquitous Computing; Sep 17-21 2006; Orange County, California, USA. Springer-Verlag; 2006. p. 177-93.

244. Doherty AR, Hodges SE, King A, Smeaton AF, Berry E, et al. Wearable Cameras in Health: The State of the Art and Future Possibilities. *Am J Prev Med*. 2012;44(3):320-3.

245. Arab L, Winter A. Automated camera-phone experience with the frequency of imaging necessary to capture diet. *J Am Diet Assoc.* 2010;110(8):1238-41.

246. Hebert JR, Clemow L, Pbert L, Ockene IS, Ockene JK. Social desirability bias in dietary self-report may compromise the validity of dietary intake measures. *Int J Epidemiol.* 1995;24(2):389-98.

247. Hebert JR, Ma Y, Clemow L, Ockene IS, Saperia G, et al. Gender differences in social desirability and social approval bias in dietary self-report. *Am J Epidemiol*. 1997;146(12):1046-55.

248.Puri M, Zhu Z, Yu Q, Divakaran A, Sawhney H, editors. Recognition and volume
estimation of food intake using a mobile device. Applications of Computer Vision (WACV), 2009
Workshop on; December 7-8, 2009. IEEE:1-8.
http://ieeexplore.ieee.org/xpls/abs_all.jsp?arnumber=5403087&tag=1. Accessed July 9, 2014.

United List of research project topics and materials

249. Zhang Z, Yang Y, Yue Y, Fernstrom JD, Jia W, et al., editors. Food volume estimation from a single image using virtual reality technology. Bioengineering Conference (NEBEC), 2011 IEEE 37th Annual Northeast; April 1-3, 2011. IEEE:1-2. http://ieeexplore.ieee.org/xpls/abs_all.jsp?arnumber=5778625. Accessed July 8, 2014.

250. Barton J, Barlow J, Chabot C, Johnson Jr H, Jones W, et al. Congressional Bi-Partisan Privacy Caucus *Letter to Google CEO* 2013; http://www.joebarton.house.gov/press-releasescolumns/lawmakers-express-concern-that-google-glass-provides-a-window-into-people-s-privacy/. Accessed October 7, 2013.

251. Kelly P, Marshall SJ, Badland H, Kerr J, Oliver M, et al. An ethical framework for automated, wearable cameras in health behavior research. *Am J Prev Med*. 2013;44(3):314-9.

252. Amft O, Tröster G. Recognition of dietary activity events using on-body sensors. *Artif Intell Med.* 2008;42(2):121-36.

253. Amft O, Tröster G. On-body sensing solutions for automatic dietary monitoring. *IEEE Pervasive Comput.* 2009;8(2):62-70.

254. Higgins JA, LaSalle AL, Zhaoxing P, Kasten MY, Bing KN, et al. Validation of photographic food records in children: are pictures really worth a thousand words. *Eur J Clin Nutr*. 2009;63(8):1025-33.

255. Boushey CJ, Kerr DA, Wright J, Lutes KD, Ebert DS, et al. Use of technology in children's dietary assessment. *Eur J Clin Nutr*. 2009;63 Suppl 1:S50-7.

256. Serra-Majem L, Frost Andersen L, Henríque-Sánchez P, Doreste-Alonso J, Sánchez-Villegas A, et al. Evaluating the quality of dietary intake validation studies. *Br J Nutr*. 2009;102(S1):S3-S9.

257. Jonnalagadda SS, Mitchell DC, Smiciklas-Wright H, Meaker KB, Heel NV, et al. Accuracy of energy intake data estimated by a multiplepass, 24-hour dietary recall technique. *J Am Diet Assoc.* 2000;100(3):303-11.

258. Sawaya AL, Tucker K, Tsay R, Willett W, Saltzman E, et al. Evaluation of four methods for determining energy intake in young and older women: comparison with doubly labeled water measurements of total energy expenditure. *Am J Clin Nutr.* 1996;63(4):491-9.

259. Tran KM, Johnson RK, Soultanakis RP, Matthews DE. In-person vs telephoneadministered multiple-pass 24-hour recalls in women: validation with doubly labeled water. *J Am Diet Assoc.* 2000;100(7):777-83.

260. Bathalon GP, Tucker KL, Hays NP, Vinken AG, Greenberg AS, et al. Psychological measures of eating behavior and the accuracy of 3 common dietary assessment methods in healthy postmenopausal women. *Am J Clin Nutr*. 2000;71(3):739-45.

261. Kroke A, Klipstein-Grobusch K, Voss S, Möseneder J, Thielecke F, et al. Validation of a self-administered food-frequency questionnaire administered in the european prospective investigation into cancer and Nutrition (epic) study: comparison of energy, protein, and

macronutrient intakes estimated with the doubly labeled water, urinary nitrogen, and repeated 24h dietary recall methods. *Am J Clin Nutr*. 1999;70(4):439-47.

262. Blanton CA, Moshfegh AJ, Baer DJ, Kretsch MJ. The USDA Automated Multiple-Pass Method accurately estimates group total energy and nutrient intake. *J Nutr.* 2006;136(10):2594-9.

263. Google. Google glass; start; what it does 2013; http://www.google.com/glass/start/what-itdoes/. Accessed 16th May, 2013.

264. Gemmell J, Williams L, Wood K, Lueder R, G. B, editors. Passive capture and Ensuing issues for a personal lifetime store. roceedings of the the 1st ACM workshop on continuous archival and retrieval of personal experiences; 15th October, 2004; New York, USA.

265. Cherry S. Total recall [life recording software]. Spectrum, IEEE. 2005;42(11):24-30.

266. Hodges S, Berry E, Wood K. SenseCam: A wearable camera which stimulates and rehabilitates autobiographical memory. *Memory*. 2011;19(7):685-96.

267. Browne G, Berry E, Kapur N, Hodges S, Smyth G, et al. SenseCam improves memory for recent events and quality of life in a patient with memory retrieval difficulties. *Memory*. 2011;19(7):713-22.

268. Kelly P, Doherty AR, Hamilton A, Matthews A, Batterham AM, et al. Evaluating the feasibility of measuring travel to school using a wearable camera. *Am J Prev Med.* 2012;2012(45):5.

269. Kelly P, Doherty A, Berry E, Hodges S, Batterham AM, et al. Can we use digital life-log images to investigate active and sedentary travel behaviour? Results from a pilot study. *Int J Behav Nutr Phys Act.* 2011;8(1):44.

270. Hamilton A, Doherty A, Marshall S, Kelly P, Kerr J, et al. Measuring episodes of sedentary behaviour using SenseCam: a pilot study. 8th International Conference on Diet and Activity Methods; May 14 -17 2012; Rome, Italy. 2012.

271. Doherty AR, Kelly P, Kerr J., Marshall SJ, Oliver M, et al. Use of wearable cameras to assess population physical activity behaviours: an observational study. *Lancet Public Health Science Conference*. 2012;(in press)(380):S35.

272. O'Loughlin G, Cullen SJ, McGoldrick A, O'Connor S, Warrington GD. Using a wearable camera to increase the accuracy of dietary analysis. *Am J Prev Med.* 2013;44 (3):297-301.

273. Probst Y. Food portion images for dietary assessement: graduated food portion photos of *Australian foods*. Wollongong: The University of Wollongong; 2012.

274. Doherty AR, Moulin CJA, Smeaton AF. Automatically assisting human memory: a sensecam browser *Memory: Special Issue on SenseCam: The Future of Everyday Research?* 2010;19(7):785-95.

275. Higgins JA, LaSalle AL, Zhaoxing P, Kasten MY, Bing KN, et al. Validation of photographic food records in children: are pictures really worth a thousand words&quest. *Eur J Clin Nutr.* 2009;63(8):1025-33.

276. Hernández T, Wilder L, Kuehn D, Rubotzky K, Moser-Veillon P, et al. Portion size estimation and expectation of accuracy. *J Food Compost Anal*. 2006;19:S14-S21.

277. Ovaskainen M, Paturi M, Reinivuo H, Hannila M, Sinkko H, et al. Accuracy in the estimation of food servings against the portions in food photographs. *Eur J Clin Nutr*. 2007;62(5):674-81.

278. Tooze JA, Subar AF, Thompson FE, Troiano R, Schatzkin A, et al. Psychosocial predictors of energy underreporting in a large doubly labeled water study. *Am J Clin Nutr*. 2004;79(5):795-804.

279. Kretsch MJ, Fong AKH, Green MW. Behavioral and body size correlates of energy intake underreporting by obese and normal-weight women. *J Am Diet Assoc.* 1999;99(3):300-6.

280. Poppitt S, Swann D, Black A, Prentice A. Assessment of selective under-reporting of food intake by both obese and non-obese women in a metabolic facility. *Int J Obes*. 1998;22(4):303.

281. Taren DL, Tobar M, Hill A, Howell W, Shisslak C, et al. The association of energy intake bias with psychological scores of women. *Eur J Clin Nutr*. 1999;53(7):570-8.

282. Six BL, Schap TE, Kerr DA, Boushey CJ. Evaluation of the Food And Nutrient Database for Dietary Studies for use with a mobile telephone food record. *J Food Compost Anal.* 2011;24(8):1160-7.

283. Bland J, Altman D. Statistical methods for assessing agreement between two methods of clinical measurement. *The lancet*. 1986;327(8476):307-10.

284. Lenard NR, Berthoud HR. Central and peripheral regulation of food intake and physical activity: pathways and genes. *Obesity (Silver Spring)*. 2008;16(S3):S11-S22.

285. De Castro JM. Eating behavior: lessons from the real world of humans. *Nutrition*. 2000;16(10):800-13.

286. Meiselman HL. Methodology and theory in human eating research. *Appetite*. 1992;19(1):49-55.

287. Berry SL, Beatty WW, Klesges RC. Sensory and social influences on ice cream consumption by males and females in a laboratory setting. *Appetite*. 1985;6(1):41-5.

288. Polivy J, Herman CP, Younger JC, Erskine B. Effects of a model on eating behavior: The induction of a restrained eating style. *J Pers*. 1979;47(1):100-17.

289. Bellisle F, Dalix AM, Slama G. Non food-related environmental stimuli induce increased meal intake in healthy women: comparison of television viewing versus listening to a recorded story in laboratory settings. *Appetite*. 2004;43(2):175-80.

290. Rozin P, Tuorila H. Simultaneous and temporal contextual influences on food acceptance. *Food Quality and Preference*. 1993;4(1):11-20.

291. Marsh S, Ni Mhurchu C, Maddison R. The non-advertising effects of screen-based sedentary activities on acute eating behaviours in children, adolescents, and young adults. A systematic review. *Appetite*. 2013;71:259-73.

292. Meiselman HL, Shepherd R, Raas M. The role of context in food choice, food acceptance and food consumption. *Frontiers in nutritional science*. 2006;3:179.

293. Kanarek RB, Orthen-Gambill N. Complex interactions affect nutrition behaviour research. *Nutr Rev.* 1986;44:172-5.

294. Dhurandhar N, Schoeller D, Brown A, Heymsfield S, Thomas D, et al. Energy balance measurement: when something is not better than nothing. *Int J Obes*. 2014.

295. Schoeller DA, Thomas D, Archer E, Heymsfield SB, Blair SN, et al. Self-report–based estimates of energy intake offer an inadequate basis for scientific conclusions. *The American journal of clinical nutrition*. 2013;97(6):1413-5.

296. Gemming L, Rush E, Maddison R, Doherty A, Gant N, et al. Wearable cameras can reduce dietary under-reporting: doubly labelled water validation of a camera-assisted 24-hr recall *Br J Nutr.* 2015;113:284-91.

297. Chen J, Marshall SJ, Wang L, Godbole S, Legge A, et al., editors. Using the SenseCam as an objective tool for evaluating eating patterns. Proceedings of the 4th International SenseCam & Pervasive Imaging Conference; 2013. ACM:34-41.

298. Laska MN, Graham D, Moe SG, Lytle L, Fulkerson J. Situational characteristics of young adults' eating occasions: a real-time data collection using Personal Digital Assistants. *Public Health Nutr.* 2011;14(03):472-9.

299. Thomaz E, Parnami A, Essa I, Abowd GD, editors. Feasibility of identifying eating moments from first-person images leveraging human computation. Proceedings of the 4th International SenseCam & Pervasive Imaging Conference; 2013. ACM:26-33.

300. Wansink B. Environmental Factors That Increase the Food Intake and Consumption Volume of Unknowing Consumers*. *Annu Rev Nutr.* 2004;24:455-79.

301. Klesges RC, Bartsch D, Norwood JD, Kautzrnan D, Haugrud S. The effects of selected social and environmental variables on the eating behavior of adults in the natural environment. *Int J Eat Disord*. 1984;3(4):35-41.

302. Barr M, Signal L, Jenkin G, Smith M, editors. Using SenseCam to capture children's exposure to food marketing: a feasibility study. Proceedings of the 4th International SenseCam & Pervasive Imaging Conference; 2013. ACM:50-1.

303. Sommer R, Steele J. Social effects on duration in restaurants. *Appetite*. 1997;29(1):25-30.

304. Meyers AW, Stunkard AJ. Food accessibility and food choice: A test of Schachter's externality hypothesis. *Arch Gen Psychiatry*. 1980;37(10):1133.

305. Doherty AR, Kelly P, Kerr J, Marshall S, Oliver M, et al. Using wearable cameras to categorise type and context of accelerometer-identified episodes of physical activity. *Int J Behav Nutr Phys Act.* 2013;10(22.10):1186.

306. Jia W, Chen H-C, Yue Y, Li Z, Fernstrom J, et al. Accuracy of food portion size estimation from digital pictures acquired by a chest-worn camera. *Public Health Nutr.* 2013:1-11.

307. Doherty AR, Caprani N, Conaire CÓ, Kalnikaite V, Gurrin C, et al. Passively recognising human activities through lifelogging. *Comput Human Behav.* 2011;27(5):1948-58.

308. Landis JR, Koch GG. The measurement of observer agreement for categorical data. *Biometrics*. 1977:159-74.

309. Illner A, Freisling H, Slimani N. Feasibility, validity and potential of new dietary assessment technologies for large-scale epidemiological studies. *Ann Nutr Metab.* 2011;58:398-.

310. O'loughlin G, Cullen S, McGoldrick A, Warrington G. a comparison and validation of dietary analysis using a standard one day food diary and the sensecam in trainee jockeys. *Proceedings from SenseCam 2012, 3rd Symposium.* 2012.

311. Flegal KM, Carroll MD, Ogden CL, Johnson CL. Prevalence and trends in obesity among US adults, 1999-2000. *JAMA*. 2002;288(14):1723-7.

312. Flegal KM, Carroll MD, Kit BK, Ogden CL. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1999-2010. *JAMA: the journal of the American Medical Association*. 2012;307(5):491-7.

313. Willett W. An overview of issues related to the correction of non-differential exposure measurement error in epidemiologic studies. *Stat Med.* 1989;8(9):1031-40.

314. Kerr J, Marshall SJ, Godbole S, Chen J, Legge A, et al. Using the SenseCam to Improve Classifications of Sedentary Behavior in Free-Living Settings. *Am J Prev Med*. 2013;44(3):290-6.

315. Sheats JL, Winter SJ, Padilla-Romero P, Goldman-Rosas L, Grieco LA, et al., editors. Comparison of passive versus active photo capture of built environment features by technology naïve Latinos using the SenseCam and Stanford healthy neighborhood discovery tool. Proceedings of the 4th International SenseCam & Pervasive Imaging Conference; 2013. ACM:8-15.

316. Li Z, Wei Z, Sclabassi RJ, Jia W, Sun M, editors. Blur detection in image sequences recorded by a wearable camera. Bioengineering Conference (NEBEC), 2011 IEEE 37th Annual Northeast; 2011. IEEE:1-2.